Analyzing Sustainable Energy Opportunities for a Small Scale Off-Grid Facility: A Case Study at Experimental Lakes Area (ELA), Ontario

By

Bhanu Duggirala

A Thesis Submitted to the Faculty of Graduate Studies In Partial Fulfillment of the Requirements For the Degree of

> Master of Natural Resources Management

Clayton H. Riddell Faculty of Environment Earth and Resources Natural Resources Institute University of Manitoba Winnipeg, Manitoba R3T 2N2

Copyright (c) 2010 Bhanu Duggirala

# UNIVERSITY OF MANITOBA

# FACULTY OF GRADUATE STUDIES

# Analyzing Sustainable Energy Opportunities for a Small Scale Off-Grid Facility: A Case Study at Experimental Lakes Area (ELA), Ontario

By

#### Bhanu Duggirala

A Thesis submitted to the Faculty of Graduate Studies of The University of Manitoba in partial fulfillment of the requirement of the degree

of

Master of Natural Resources Management (M.N.R.M)

# (c) 2010

Permission has been granted to the Library of the University of Manitoba to lend or sell copies of this thesis/practicum, to the National Library of Canada to microfilm this thesis and to lend or sell copies of the film, and to University Microfilms Inc. to publish an abstract of this thesis/practicum.

This reproduction or copy of this thesis has been made available by authority of the copyright owner solely for the purpose of private study and research, and may only be reproduced and copied as permitted by copyright laws or with express written authorization from the copyright owner.

#### Abstract

This thesis explored the opportunities to reduce energy demand and renewable energy feasibility at an off-grid science "community" called the Experimental Lakes Area (ELA) in Ontario. Being off-grid, ELA is completely dependent on diesel and propane fuel supply for all its electrical and heating needs, which makes ELA vulnerable to fluctuating fuel prices. As a result ELA emits a large amount of greenhouse gases (GHG) for its size. Energy efficiency and renewable energy technologies can reduce energy consumption and consequently energy cost, as well as GHG.

Energy efficiency was very important to ELA due to the elevated fuel costs at this remote location. Minor upgrades to lighting, equipment and building envelope were able to reduce energy costs and reduce load. Efficient energy saving measures were recommended that save on operating and maintenance costs, namely, changing to LED lights, replacing old equipment like refrigerators and downsizing of ice makers. This resulted in a 4.8% load reduction and subsequently reduced the initial capital cost for biomass by \$27,000, by \$49,500 for wind power and by \$136,500 for solar power.

Many alternative energies show promise as potential energy sources to reduce the diesel and propane consumption at ELA including wind energy, solar heating and biomass. A biomass based CHP system using the existing diesel generators as back-up has the shortest pay back period of the technologies modeled. The biomass based CHP system has a pay back period of 4.1 years at \$0.80 per liter of diesel, as diesel price approaches \$ 2.00 per liter the pay back period reduces to 0.9 years, 50% the generation cost compared to present generation costs. Biomass has been successfully tried and tested in many off-grid communities particularly in a small-scale off-grid setting in North America and internationally. Also, the site specific solar and wind data show that ELA has potential to harvest renewable resources and produce heat and power at competitive rates compared to diesel and propane.

iii

# Acknowledgements

I would like thank Dr. Shirley Thompson for her guidance and support throughout this project. I would also like to thank my committee members Mr. Denis DePape and Dr. Eric Bibeau for their guidance and suggestions. This project would not have been possible without the financial support from the Department of Fisheries and Oceans (DFO). I would like to thank Paul Herage, Ray Pambrun, Ken Beaty, Duane Jordan for their support and in-kind contribution towards this project.

Abstract	iii
Acknowledgements	iv
List of Tables	vii
List of Figures	viii
Glossary of Terms	ix
CHAPTER 1: INTRODUCTION	1
1.1 Background	1
1.2 Study Area	5
1.3 Purpose and Objectives	6
1 4 Significance of Research	7
1.6 Research Approach	8
1.7 Assumptions	8
1.8 Thesis Organization	9
CHAPTER 2: LITERATURE REVIEW	10
2.1 Introduction	10
2.2 Climate Change	11
2.2 Childre Change	
2.2.1 Contributing Factors	11.
2.2.2 Northern Impacts of Climate Change	.12
2.2.3 Measures	.12
2.3 Energy Generation in Off-Grid Communities	.12
2.3.1 Reasons for Diesel Power Generation	.13
2.3.2 Disadvantages	
2.4 Energy Efficiency	.15
2.5 Small Scale Renewable Energy	.16
2.5.1 Small Scale Wind	.17
2.5.2 Solar Power	.18
2.5.3 Biomass	20
2.6 Chapter Summary	22
References	23
CHAPTER 3: RESEARCH METHODS	
3.1 Introduction	
3.2 Research Steps	27
3.2.1 Energy Efficiency Analysis	27
3.2.2 RETScreen 4.0 Analysis	
CHAPTER 4: DEMAND REDUCTION FINDINGS REGARDING LOW-COST	
AND NO-COST RETROFIT ALTERNATIVES	
4.1 Introduction	32
4.2 Energy Retrofits of Existing Buildings	33
4.2.1 Low-Cost and No-Cost Retrofits Vs Comprehensive Retrofits	33
4 3 Base Case Characteristics	35
1.5 Dase Case Characteristics	35
A A 1 Current Energy Sources and Distribution	.35
4.4.1 Current Energy Sources and Distribution	.30
4.4.2 Cost of Electrical energy derived from dieset generators	
4.5 Lignung System	.31
4.0 Laboratory Equipment	
4.0 building Envelope	.40
4.0.1 Air Leakage	.40
4.0.2 Insulation	.41

# **Table of Contents**

4.6.3 Infiltration Losses	42
4.7 Chapter Summary	43
Acknowledgements	46
References	47
CHAPTER 5: FEASIBILITY STUDY OF RENEWABLE ENERGY	
TECHNOLOGIES AT ELA	48
5.1 Introduction	48
5.2 Findings	50
5.2.1 Energy Efficiency	51
5.2.2 Impacts of RETs on GHG Emissions	55
5.3 Chapter Summary	56
References	60
CHAPTER 6: CONCLUSION AND RECOMMENDATIONS	62
6.1 Introduction	62
6.2 Summary	63
6.3 Conclusion	63
6.4 Recommendations for Diesel Off-Grid Community	65
6.4.1 Current System	65
6.4.2 Proposed Energy Management System	65
6.5 General Recommendations	68
Appendix – I	70
Appendix – II	79

# List of Tables

Table 2.1:	Characteristics of Diesel power generation	.13
Table 3.1:	Meteorological data of ELA field station for the year 2007	29
Table 4.1:	Comparison of Low-Cost & No-Cost Retrofits Vs Comprehensive	
	Retrofits	34
Table 4.2:	General Characteristics	35
Table 4.3:	Fuel Characteristics of existing energy system (2006)	.36
Table 4.4:	Base Case Characteristics	37
Table 4.5	Replacement costs for upgrading existing exit lamp with LEDs	38
Table 4.6:	Potential savings by switching to a smaller ice maker	38
Table 4.7:	Potential savings by turning one of the 600°C ovens OFF overnight	39
Table 4.8:	Energy cost saved by downsizing to a smaller more efficient oven	39
Table 4.9:	Summary of potential savings with Recommendations and Attractive Indicator.	45
Table 5.1:	Selection of Demand Reduction Recommendations with High Attractiveness Index	51
Table 5.2:	Economic and Financial parameters for RETs	53
Table 5.3:	GHG emissions from fuels combusted at ELA	56
Table 5.4:	RETScreen analysis of the three scenarios at different diesel prices	59
Table 6.1:	A Comparisons of six small-scale biomass CHP systems	64

# List of Figures

Figure 1.1:	Existing Energy Model at ELA4
Figure 1.2:	Parameters considered for Sustainable Energy Analysis at ELA4
Figure 1.3:	Location of ELA on provincial map
Figure 1.4:	Ariel view of field station surrounded by woods and lakes6
Figure 2.1:	Wind map for study area region at 50m hub height17
Figure 3.1:	Methodology to Determine Demand Reduction
Figure 3.2:	Study Methodology for Renewable Energy Technology Analysis26
Figure 4.1:	Fuel Consumption Breakdown at ELA
Figure 4.2:	Cracks found on the outer wall in the Laboratory Building40
Figure 4.3:	Air Conditioning cable holes can be better insulated in laboratory building
Figure 4.4:	Indicates that basements can still be improved for better sealing41
Figure 4.5:	Potential areas where basement insulation can be improved41
Figure 4.6:	Outside vent foundation with exposed wood foundation42
Figure 4.7:	Doors were found to have been installed with brush insulation which could be replaced with magnetic strips which provides better insulation
Figure 4.8:	Other doors where drafts were noticed
Figure 5.1:	Energy Map of ELA revealing the flow of energy through the facility for the 2006/2007 fiscal year
Figure 5.2:	Electrical energy breakdown indicates that HVAC and lighting combined consume 2/3 <sup>rds</sup> of the total power
Figure 6.1:	Comparison of Existing Energy Model and the Recommended Energy Model
Figure 6.2:	Layout of Sustainable Energy Plan for Small Off-Grid Diesel Communities

# **Glossary of Terms**

- **CHP** Combined Heat and power
- **ELA** Experimental Lakes Area
- GHG Green House Gases
- O&M Operations and Maintenance
- $\mathbf{PV}$  Photovoltaic
- **R -12** Thermal Resistance Factor
- **RET** Renewable Energy Technologies
- SPV Solar Photovoltaic

#### **CHAPTER 1: INTRODUCTION**

#### 1.1 Background

As the sixth largest energy consumer of energy in the world (Environment Canada, 2005), and with the second highest per capita energy consumption rate among the G-8 nations (International Energy Agency, 2004), Canada needs to explore sustainable energy opportunities for urban and rural developments and reduce energy consumption to make a steady shift towards demand reduction and renewable energy technologies. Industry and government interests have channeled most of the demand reduction and renewable energy research at highly populated, developed, and urbanized establishments like cities, business, sub-urban residences, industries and neglected remote communities. However, with over 300 off-grid communities in Canada and with a combined population of ~200,000 across the country operating on fossil fuels it is also important that these communities meet their energy requirements in a sustainable and cleaner manner (EIA, 2005a; Ah-You & Leng, 1999). This research explores demand reduction through energy efficiency, as well as renewable energy technologies, at one off-grid location.

Clean renewable energy sources have potential for implementation in small off-grid facilities because they presently require diesel generation to generate electricity, bulky to ship and costly to consume; however, with numerous renewable energy sources a complete economic and technological analysis is required to identify the most suitable and efficient resources (Ah-You & Leng, 1999). Suitable energy efficient measures and viable renewable energy technologies must be evaluated before being considered to realize the huge potential for application in off-grid facilities. As the quality of renewable energy resources vary with the location it is necessary to perform a site-specific analysis to compare renewable energy resource considered applicable to this specific site.

Energy is a driving force in any off-grid community. Existing heat and power generation technologies and energy usage patterns create negative environmental impacts such as green house gas production (Khan and Islam, 2003a, 2003b, 2004, 2005a), particularly at Experimental Lakes Area (ELA) where diesel is used. Therefore, an energy profile to determine sustainability should consider efficiency, cleanliness of energy, reliability and reliable and ready-to-use energy source for any off-grid<sup>1</sup> diesel establishment to satisfy the two most important amenities for operation, namely heat and electrical power that are necessary for steady and dependable operation of the facility (PWGSC, 2007). A sustainable energy plan for ELA will be a collective approach consisting of two key components, which are energy efficiency and renewable energy (REEEP, 2005). The research will consider: 1) the feasibility of efficiency measures by energy conservation; and 2) renewable energy technology feasibility.

Energy efficiency measures can yield significant savings for off-grid communities by reducing fuel use (PWGSC, 2007). The existing heating and electrical systems at ELA are inefficient, expensive to operate and maintain and completely dependent on diesel and propane fuels. The overall energy costs are dictated by diesel and propane prices, which in turn are dependent on fluctuating global oil prices. Based on the trend for diesel and propane prices, the energy cost at ELA are expected to increase thus making renewable energy more feasible in the future.

With increasing oil prices, Renewable Energy Technologies (RETs) are attractive options for off-grid communities and are becoming a reality in many off-grid communities across Europe and Asia. However, as every off-grid facility has unique resource availability and

<sup>&</sup>lt;sup>1</sup>Off-grid refers to a single or cluster of buildings or community that is not connected to the provincial electrical grid

varying base loads<sup>2</sup> there is no "one size fits all". Each site has to be individually assessed for energy resources, energy requirements needs and demand reduction by identifying retrofitting opportunities. By implementing demand reduction measures and adopting renewable energy technology ELA creates an opportunity not only to reduce its energy cost but also to mitigate its environmental footprint. The desirable features for a sustainable energy system for ELA were identified as:

- reduced demand for heat and power
- reduced GHG emissions
- lower and stable energy costs
- cleaner technology
- autonomous supply of heat and power minimum consumption of fossil fuels
- reduced risk from fuel spills and leaks during storage and transportation
- lower operating & maintenance (O&M) costs

Figure 1.1 represents the current energy model at ELA. A 100% diesel based power generation system with propane heating set-up has been chosen because of low capital cost and common availability of the technology. Limited random and incomplete energy efficiency measures have been implemented that lead to high energy cost and high GHG emissions. This model also requires large amounts of fuel to be stored on site that increases the risk of a fuel spill, which would be very damaging to the environment and expensive to clean up. Complete dependency on an inefficient system for diesel and propane is unsustainable in the long run as these fossil fuels are prone to fluctuation in fuel price and transportation costs. The aim of this thesis is to propose an improved energy system that incorporates a hybrid power generation using local renewable sources

<sup>&</sup>lt;sup>2</sup> Base load is the minimum level of demand on an electrical supply system over 24-hours, the load that exists 24 hours a day

and mitigating the dependency on fossil fuel while reducing the energy cost and GHG emissions.



Figure 1.1 Existing Energy Model at ELA

Figure 1.2 portrays various parameters of the analysis of sustainable energy development

at ELA considering energy efficiency and renewable energy.



Figure 1.2 Parameters Considered for Sustainable Energy Analysis at ELA

#### 1.2 Study Area

The study area is the experimental lakes area (ELA) field station located at 50 km south of Kenora, Ontario (See Figure 1.2) in Canada.



Figure 1.3 Location of ELA on provincial map of Ontario

Source: http://www.umanitoba.ca/institutes/fisheries/location.html

This northern location of ELA is latitude 49 degrees, 47 minutes and 15 seconds north. The weather at this latitude varies from +30 degrees Celsius in the summer to -30 degrees Celsius in the long winter. The field station includes 20 buildings, which are mainly clustered around the laboratory and kitchen as shown in Figure 1.3. The total laboratory space is about 6,900.00 square feet. Although some buildings date back to 1968, in 2001 three new buildings were added to the facility, namely, a new laboratory and two R-2000 energy efficient residences. These residences provide common areas and about ten single rooms.

#### **Study Location Information:**

Latitude: between 49° 34' and 49° 47' North Longitude: between 93° 36' and 93° 52' West Elevation: 411m Heating Design Temperature: - 29.8 °C Cooling Design Temperature: 27.3 °C



Figure 1.4 Ariel view of field station surrounded by woods and lakes

DFO (Department of Fisheries and Oceans) operates this unique facility in the remote region of northwestern Ontario year round. It has very limited occupancy (about 4 to 5 people) during the winter/fall and spring months from October to April but houses about 40 people from mid-May to September (Pambrun, Personal Communication). The primary purpose of the facility is to accommodate researchers and students to carry out research experiments in the lakes, which requires collecting samples and then analyzing them. The on-site meteorological station has a range of measuring instruments, which are monitored and recorded on a daily basis (Beaty, Personal Communication). Energy for the facility is currently provided by a diesel power generator.

#### **1.3 Purpose and Objectives**

The main purpose of this thesis is to explore improvements that can be made through energy efficiency measures and to evaluate renewable energy opportunities compared to the current diesel situation. This thesis will attempt to answer the following research questions:

1. What is the current form of energy supply and what are the resulting environmental and economic impacts?

- 2. Are there any opportunities to reduce energy consumption by improving energy efficiency that result in savings and a short payback period?
- 3. What type of renewable energy technologies would be most cost-effective at providing a consistently stable and reliable energy supply at the scale and geographical location of the facility?

The project's three objectives, derived based on the above mentioned research questions are as follows:

- To understand the existing energy set up at ELA by creating an energy profile of ELA and to develop a base case scenario for the year 2006/2007.
- 2. To identify and recommend energy saving opportunities using demand reduction approach and to estimate the resulting savings.
- 3. To perform a renewable energy analysis that compares different renewable energy resources to identify the most feasible renewable technology for the site and to integrate it with identified energy efficiency measures.

#### **1.4 Significance of Research**

This thesis is significant as it applies RETScreen to compare different renewable energies for the same site. It is also one of the few research analysis of renewable energy that considers energy efficiency prior to assessing renewable energy. This research is also applied and, as such, may have environmental and economic benefits to ELA. Being a long-term off-grid research establishment, which was built in wilderness, ELA has the responsibility to conduct research in an environmentally friendly manner to maintain the ecological and biological integrity of the surrounding ecosystem. Economically, it contributes to the potential reduction in heat and power generation cost with improved efficiency. Overall, this work will contribute by discovering the potential opportunities

for communities, particularly off-grid but with applications to on-grid communities, to make them more sustainable.

#### **1.5 Research Limitations**

There are certain limitations on this study due to the lack of baseline data and facility status,

- *Lack of baseline data:* Although some studies have carried out renewable energy analysis for wind, solar radiation and biomass there was no study that has carried energy analysis from a demand reduction point of view.
- *Energy Monitoring:* No energy monitoring systems exist at ELA to provide an indication of energy usage between different buildings or different equipment.

# **1.6 Research Approach**

The research objectives were met through collection of primary data using a walk-through survey, informal discussions with DFO people, literature review and analysis by RETScreen. Further explanation of these methods is presented in Chapter 3.

# **1.7 Assumptions**

The following assumptions were made:

- The demand for power and heat would remain constant through the study period independent of number of occupants (assuming that heating needs in winter for building maintenance would balance the increased electrical needs in summer).
- The efficiency of the diesel power generators is assumed to remain constant for the period of study.
- The fuel price for diesel and propane remain constant throughout the study period and therefore the cost per kWh remains constant.

- The mechanical efficiency of the generators, appliances and HVAC systems do not change over the study period.
- The ELA facility has four main areas where most of the daily activities are carried out and therefore are assumed to be the major consumers of heat and power, namely:
  - Kitchen and dining hall
  - o Chemistry lab
  - o General workshop
  - Residence buildings.

#### **1.8 Thesis Organization**

This thesis is organized into six chapters. Chapter 1 provides a general introduction. Chapter 2 is the literature review of energy management in small off-grid communities in Canada and of renewable energy technologies. Chapter 3 outlines the study methods. Chapter 4 discusses the results of the demand reduction analysis for ELA and provides recommendations. Chapter 5 consists of the renewable energy analysis. Chapter 6 summarizes and concludes considering the overall significance of this thesis work.

#### **CHAPTER 2: LITERATURE REVIEW**

#### **2.1 Introduction**

Energy management at a small off-grid community is an important issue driven mainly by economic factors like fuel cost, equipment efficiency, operation and maintenance costs, greenhouse gas (GHG) emissions and environmental risk (e.g. fuel spill). There are also efficiency issues of off-grid communities that contribute to higher demand like poorly insulated buildings in low quality housing, inefficient lighting and equipment. Historically, off-grid facilities in Canada have always had higher energy costs primarily due to expensive operations and maintenance (O&M) costs which are dictated by fuel prices, high cost for shipping and also due to diesel generation being an inefficient heat and power generation system. These communities consume large quantities of fossil fuels every year in order to meet their energy needs and as a result, emit large quantities of greenhouse gas (GHG) emissions contributing to the national emissions (affordable power in rural Alaska, 1996). As a result the energy use in these communities would be even higher than that of the average Canadian. Canadians use more energy per person than people in any other industrialized nation and on average, each citizen accounts for about 21 tonnes of greenhouse gases per year (Natural Resources Canada, 2008).

Regardless of the climate, higher energy costs and environmental impacts occur at offgrid facilities that use fossil fuels (diesel and propane) for heat and power generation (Isherwood et. al, 2000). Heat and power generation from diesel and propane is not only expensive in a remote setup for reasons like higher transportation cost for fuel, smaller population base, higher O&M cost and greater need for space heating but also carries a higher environmental risk of fuel spill during transportation and storage (Chapman,

1996). These challenges are exclusive to any off-grid facility and are to be overcome in order to make the desired shift towards sustainable energy.

#### 2.2 Climate Change

There is strong evidence that levels of greenhouse gases in the atmosphere are increasing and that the world is getting warmer (Ayalon et al., 2001; IPCC, 1996). Climate change has unpredictable and far reaching environmental, economic and social consequences. The changes in temperature impact climate patterns such as wind, snow and storm intensity. Impacts include flooding and erosion, increased risk of forest fires, water shortages and drought.

#### 2.2.1 Contributing Factors

Activities that contribute to greenhouse gas emissions are as follows.

- energy generation, particularly coal-based and fossil fuel based generation, such as the diesel at ELA;
- heating and cooling;
- transportation; and,
- high energy use

The six main greenhouse gases covered under the *Kyoto Protocol* are carbon dioxide, methane, nitrous oxide, hydroflurocarbons (HFC), perflourocarbons and sulphur hexafluoride. Of all the above mentioned gases, carbon dioxide is the main concern as it is closely associated with human activities and is thought to be the main contributor to climate change, especially through burning of fossil fuels like coal, gasoline, diesel, and propane. The concentrations of carbon dioxide in the atmosphere have increased by 30 percent and concentrations of methane and nitrous oxide have increased by 145 percent and 17 percent respectively (Goudie, 2001). In Canada GHG emissions in 2000 were 15

percent greater than they were in 1990. Canada is the third largest per capita emitter of these gases, after the United States and Australia (Turton, et.al, 2002).

#### 2.2.2 Northern Impacts of Climate Change

The impact of climate change is likely to be more severe in the polar regions than near the equator. Rising global temperatures are melting glaciers and decreasing ice cover, affecting the way of life for northern communities. Many northern communities rely on winter ice roads during a brief window of time during the winter season that allows for safe transport of people and material. Should climate change progress significantly the reliability of these routes are threatened. Due to this the energy costs are significantly higher due to the high cost of transporting fuel to sites, if they have to be flown in, and also due to accessibility issues like long, cold winters and short hours of daylight.

#### 2.2.3 Measures

The purpose of this chapter is to provide an overview of literature regarding the energy situation in small off-grid communities. This chapter is divided into three sections that look at the key areas of energy management that are relevant to this project. The three sections are: **Energy generation, Energy efficiency and Renewable energy**.

#### 2.3 Energy Generation in Off-Grid Communities

Many off-grid communities in Canada produce power from diesel and heat from propane fuels. Diesel is combusted to produce electricity and propane is combusted in furnaces to produce hot water and space heating. The fuel is shipped into the community through trucks and is stored in large capacity storage tanks. Table 2.1 gives the general characteristics of a diesel based power generation.

	Diesel Engine
Capacity Range	5 kW – 20 MW
Electrical Efficiency (%)	35 - 45
Overall Efficiency (%)	65 – 90
Power to heat ratio	0.8 - 2.4
Noise	Loud (Continuous)
CO <sub>2</sub> emissions (Kg/MWh)	650
NO <sub>x</sub> emissions (Kg/MWh)	10
Availability (%)	95
Part load performance	Good
Life cycle (yr)	20
Average cost investment (\$/kW)	340 - 1000
O & M cost (\$/kWh)	0.0075 - 0.015

**Table 2.1 Characteristics of Diesel Power Generation** 

Source: Adapted from Wu and Wang, 2006

#### 2.3.1 Reasons for Diesel Power Generation

Reciprocating engines are a proven technology with a range of sizes and the lowest initial capital costs. In addition to fast start-up capability and good operating reliability, high efficiency at partial load operation give a flexible power source, allowing for a range of different energy applications - especially for off-grid locations. Reciprocating engines are by far the most commonly used power generation equipment under 1 MW capacity (Wu and Wang, 2006). Hanley & Nevin (1999) identified the major characteristics that contribute for existing system of generation as:

- fuel has been cheap historically;
- readily available reliable reciprocating engine technology;

- physical access constraints to connect to grid;
- lack of infrastructure; and
- lack of knowledge and high price of renewable energy equipment.

These communities have long been neglected due to small populations, lack of political clout and poverty, which prevented a connection to the provincial grid. The power companies do not see a business case in connecting the small remote communities to their grid.

#### 2.3.2 Disadvantages

Although reciprocating engines are a mature technology, obvious drawbacks exist. Relatively high vibrations require shock absorption and shielding measures to reduce acoustic noise. A large number of moving parts and the requirement of frequent maintenance intervals increase maintenance costs and strongly offset any fuel efficiency advantages. Moreover, these systems produce toxic air emissions, particularly nitrogen oxides (Wu and Wang, 2006). There are various economic, as well as environmental disadvantages that result from this technology being used in heat and power generation. Some of the important disadvantages are listed below according to their category of economic, environmental and social:

#### **Economic:**

- Higher energy cost
- Energy cost totally and directly dependent on global oil price
- Low efficiency
- Poor energy security (No diversity in energy sources)

#### **Environmental:**

- Greenhouse gas emissions
- Particulate emissions
- Noise pollution
- Environmental risk associated with transportation and storage of fuels (Spillage and Cleanup)
- Larger environmental footprint

#### Social:

- provide limited local employment initiatives
- contamination of land and food supplies due to fuel spills

#### 2.4 Energy Efficiency

Many opportunities exist to improve energy efficiency using current off-the-shelf commercial technologies. Such technical and economic opportunities are considered "win-win" as their advantages include reducing energy consumption and by indicators of their cost effectiveness, such as cost of conserved energy, simple pay-back time, and economic rate of return (Martinot and McDoom, 2000). Large number of scientific studies combined with the extensive practical experience of the past 30 years point to many technology applications that meet cost-effectiveness criteria (such as 20% rate of return on investment or five-year simple payback time), and that offer large potential for CO<sub>2</sub> emissions reduction. There are many innovative technologies that enable us to meet the energy efficiency requirements. Some of these technologies are energy efficient Heating Ventilation and Air Conditioning, LED lighting, waste heat recovery and Energy Star rating standards. In many sectors, 10 to 30 percent (or more) of energy consumption can be saved using measures that have already been commercialized and that are costeffective to consumers and society (Martinot and McDoom, 2000).

#### 2.5 Small Scale Renewable Energy

Though many provinces in Canada have been producing hydroelectric power, the oil crises of the 1970s ignited a strong interest in some other forms of renewable energy. With its extensive geography, Canada has vast renewable energy resources (Islam, et. al., 2004). Renewable-energy technologies that are already or nearly commercialized include solar, small-scale biomass power generation and small scale off-grid wind power. Renewable energy potential depends on geographic resources such as wind speeds, solar radiation, and biomass residues from agriculture and other industries. If good geographic resources are present, several applications offer plentiful opportunities for cost-competitive commercial or near-commercial renewable energy (Martinot and McDoom, 2000).

In remote locations, renewable energy technologies (RETs), coupled with state-of-the-art energy storage methods (e.g. batteries), can economically compete favorably with conventional fossil fuel generation when the comparisons include environmental quantitative and qualitative parameters for the entire integrated energy system (i.e., heating along with electrical power). RETs apply particularly well where electric costs are high because of fuel transportation expense, there is a reasonable renewable resource available (e.g., wind, biomass, solar) and there is no inter-connection to a large-scale power grid (Isherwood, et. al, 2000). Renewable energy combined with energy storage also has the potential to provide the important benefit of increased system reliability; this has been recognized as one of the highest priorities in the design of remote power systems (Brown et. al, 1996).

#### 2.5.1 Small Scale Wind

Wind power is considered a clean renewable energy. The high cost of energy in diesel powered communities combined with a desire to become more self-sufficient has led to an interest in wind energy systems from communities, governments and utilities (Timothy & Adrian, 2008). However, wind power fails to be a stable power source as wind behavior is intermittent and undispatchable<sup>3</sup>. To compensate for this instability of wind, wind-diesel hybrid systems have been developed. This reduces diesel consumption, reducing at least 30% of the final cost of the electricity. Small wind turbine generators that are connected to batteries provide sufficient electricity for rural dwellings, remote communications and other isolated areas. Figure 1 shows the wind atlas for the study area between 49° 34' and 49° 47' north latitude, and between 93° 36' and 93° 52' west longitude. The wind map (Figure 2.1) shows that on an average 5 to 6 m/s wind speeds are available for power generation at ELA.



Figure 2.1 Wind map for ELA Study Area at 50m Hub Height

Source: Canadian Wind Energy Atlas, 2008

<sup>&</sup>lt;sup>3</sup> To produce electricity when needed

Below are listed the important advantages and disadvantages of wind energy technology with rest to remote off-gird locations:

#### Advantages:

- each megawatt-hour of electricity generated by wind energy helps to reduce 0.8 to
   0.9 tonnes of greenhouse gas emissions that are produced by coal or diesel fuel
   generation each year (NRC, 2006)
- wind energy does not release carbon dioxide, nitrogen oxide, sulphur dioxide or mercury into the atmosphere like many traditional forms of electrical generation does therefore contributes in decreasing GHG emissions

#### **Disadvantages**

- the wind speed is not constant, varying from zero meters/sec to storm force, which means that wind turbines do not produce the same amount of electricity all the time and there will be times when they produce no electricity at all.
- maintenance is a problem in remote regions without qualified personnel to perform annual or seasonal maintenance checks on the tower and turbine

#### 2.5.2 Solar Power

Sun's energy has long been used for common activities such as preserving food for longterm storage and for drying different materials. Today technology allows us to utilize the sun's energy for diverse applications like: *Photovoltaic systems, solar air and water heating systems and passive solar systems*. Solar radiation energy can be converted to electricity or heat. A photovoltaic (PV) cell made of semiconductor materials (e.g. silicon), can convert solar energy to electricity with 15-20% efficiency. Currently, Solar Photovoltaic (SPV) modules generate electricity for homes, cottages, and are effective in meeting power needs in remote locations and as an alternative to transmission lines or diesel generators. *Photovoltaic's (PV):* Photovoltaic cells convert suns energy into electricity for use in homes, buildings or remote applications (off-grid communities). The efficiency of PV modules increases in colder temperatures and is well suited for Canada climate. PV technology is still relatively expensive but the costs are predicted to come down with less expensive technologies being developed. PV systems are most cost-effective in small load applications in remote areas.

*Solar Air and Water Heating Systems:* The sun's energy can be used for space heat in buildings. The solar wall is one application. When sunlight hits a dark metal it is absorbed heating the air space and therefore preheating the air drawn into the building's main heating system. Use of the solar wall is most cost effective in northern locations where the sunlight reflects off the snow to improve the solar gain. Similarly, solar water heaters collect the suns energy to heat water for domestic uses like cooking, washing, etc as well as for space heating.

*Passive Solar:* Passive solar is a method of building construction that takes advantage of the solar radiation through placement of windows and use of materials that absorb, reflect and store solar radiation as needed to regulate the temperatures indoor. It is not necessary to live in a hot climate to take advantage of solar energy; in fact, some technologies operate more efficiently in cold climates. Important relevant factors in evaluating feasibility of solar renewable energy technology include number of hours of sunshine on a daily basis and the intensity of the solar radiation.

#### <u>Advantages</u>

solar energy systems are virtually maintenance free and will last for decades.
 Once installed, there are no recurring costs.

- solar energy systems operate silently, have no moving parts, do not release
  offensive smells and do not require you to add any fuel and more solar panels can
  easily be added in the future when the need arises.
- solar energy systems can operate independently without a connection to a power grid at all. Systems can therefore be installed in remote locations, making it more practical and cost-effective than the supply of utility electricity.

#### **Disadvantages**

- the initial cost of installing a solar energy system is high because of the expensive cost of the semi-conducting materials required for it
- the efficiency of the system also relies on the location of the sun, which is overcome by the installation of motors to change the direction of the solar panel
- the production of solar energy is influenced by the presence of clouds or pollution in the air

#### 2.5.3 Biomass

With over 2.4 million km<sup>2</sup> of forest area, Canada has the world's third largest forest area that supports a massive wood-based sector consisting of timber, pulp and paper and other associated products (World Energy Council, 2001). Approximately 6% of Canada's primary energy is from Biomass energy in the form of combustion of wood and wood derivatives for industrial process heat, generation of electricity, and space heating (Natural Resources Canada, 2002). Wood based energy generation units can use the surplus residue to produce heat and power simultaneously in a system called Combined Heat and Power (CHP). ELA location provides an abundant supply for bio-mass like waste lumber from the forest and waste wood from nearby urban areas. Biomass for decentralized power generation for off-grid communities has broad load range

application. The following are some of important factors that are to be considered for Biomass based CHP system.

#### Scaling

Some technologies are better suited for smaller scales of energy production. For example, a combustion furnace or wood stove is appropriate for space heating. Any proponent of a proposed biomass facility must undertake an analysis to determine the most appropriate technology to use. Combined heat and power (CHP) biomass facilities (typically using the Rankine Cycle) are able to achieve high levels of efficiency only by capturing low-quality heat for uses such as space and water heating (combined with using high quality heat to produce electricity). Biomass facilities that do not fully capture heat produced result in a waste of energy resources, and will put an unnecessary strain on a sustainable supply of fuel wood. Therefore, it is imperative that heat production from CHP biomass facilities be considered and paired with requirements for such heat during project design and development.

#### Transportation

A large portion of the cost of feedstock acquired by biomass based energy plants are transportation costs (Combs, 2008). A study in Wisconsin has shown that, the use of switch-grass as fuel in a CHP system contributes to 10% increase in production costs of energy for every 30 miles (48 km) increase in fuel transportation (Porter et al, 2008). Similar transportation distance limitations likely apply for wood. Therefore, any biomass facility must be located near sufficient fuel sources, such as wood processing facilities, or properly managed forests to yield a positive energy balance. The rise in the cost of fossilderived transportation fuels, and the high costs of highway maintenance, encourage the location of biomass facilities near biomass sources.

#### **Advantages**

- sources are commonly available, locally produced and variable including: wood, wood chips, switch grass, wheat straw, etc
- carbon neutral technology in the case that new plants are grown to replace the ones harvested for fuel

#### **Disadvantages**

 maintaining a steady supply of wood can be difficult, therefore, need masses of storage space and sheds for wood storage for continuous operation and also need to maintain a large inventory of biomass to avoid fuel supply irregularities

#### 2.6 Chapter Summary

Energy management is becoming critical in moving towards a more sustainable community. It permeates all aspects of a community from transportation to building design. Having a sustainable energy plan in place, one that includes energy efficiency objectives as well as renewable energy, will enable a community to better manage the impacts of rising fuel costs, greenhouse gases and energy cost while becoming less dependent on diesel and propane.

The literature review identifies how energy efficiency and renewable energy (though best suited for large towns and cities) could benefit small off-grid communities. Energy efficiency measures need not be prohibitively expensive as they can be customized based on budget and payback period constraints of the community. Sustainable energy evaluation for a small off-grid community provides both challenges and opportunities. Challenges in terms of lack of base line data, energy monitoring, remote location and opportunities in terms of demand reduction through energy efficiency measures, reduced GHG emissions. The literature points out that energy efficiency and renewable energy

analysis can be applied not only to large on-grid communities but also to small scale offgrid communities that form the central focus of this study. Unlike sustainable energy plans for large communities with a few thousand people (where utility companies design a program and implement on a large scale), for small communities a strategy with closer attention is required examining the small details that add up to important savings.

#### References

- Ah-You. K, Leng. G. (1999). Renewable energy in Canada's remote communities. CANMET Energy Diversification Research Lab Publication.
- Alaska Rural Electricity Cooperative Association. (1996). Affordable power in rural Alaska: ARECA rural issues forum report
- Ayalon O. et al., 2001. "Solid Waste Treatment as a High-Priority and Low-Cost Alternative for Greenhouse Gas Mitigation", Environmental Management Vol. 27, No. 5, pp. 697–704, DOI: 10.1007/s002670010180
- BCIT: Renewable energy: Overview (2007). Retrieved 6/3/2008, from http://www.bcit.ca/appliedresearch/re/
- Brown. C, Guichard. A, Lyons. D. (1996). Wind energy in Polar Regions: Casey Station Antarctica. In: Proceedings of the 12th Annual Conference of the Canadian Wind Energy Association, Kananaskis, Canada, 1–4.

Chapman, R. N. (1996). Hybrid power technology for remote military facilities.

- Combs, S. (2008). The Energy Report. Texas Comptroller of Public Accounts
- ecoEnergy for aboriginal and northern communities- overview- Indian and northern affairs Canada. (2008). Retrieved 5/10/2008, from <u>http://www.ainc-inac.gc.ca/clc/prg/eco/ovr\_e.html</u>
- Energy consumption issue context. (2008). Retrieved 5/9/2008, from <u>http://www.ec.gc.ca/soer-ree/English/Indicators/Issues/Energy/Bulletin/ec\_iss\_e.cfm</u>
- Energy consumption per capita in G-8 countries, (1990 and 2002). Retrieved 5/9/2008, 2008, from <a href="http://www.statcan.ca/english/research/11-621-MIE/2005023/tables/table1.htm">http://www.statcan.ca/english/research/11-621-MIE/2005023/tables/table1.htm</a>
- Energy information administration international total primary energy consumption and energy intensity data. (2008). Retrieved 5/9/2008, from <a href="http://www.eia.doe.gov/emeu/international/energyconsumption.html">http://www.eia.doe.gov/emeu/international/energyconsumption.html</a>

Goudie, A. (2001). The Human Impact on the Natural Environment.

Hanley, N., & Nevin, C. (1999). Appraising renewable energy developments in remote communities: The case of the north assent estate, Scotland. *Energy Policy*, 27(9), 527-547.

- Intergovernmental Panel on Climate Change (IPCC), 1996. "Good Practice Guidance and Uncertainty Management in National Greenhouse Gas Inventories." <<u>http://www.ipcc-nggip.iges.or.jp/public/gp/english/5\_Waste.pdf</u>>.
- Isherwood, W., Smith, J. R., Aceves, S. M., Berry, G., Clark, W., Johnson, R., et al. (2000). Remote power systems with advanced storage technologies for Alaskan villages. *Energy*, 25(10), 1005-1020.
- Islam, M., Fartaj, A., & Ting, D. S. (2004). Current utilization and future prospects of emerging renewable energy applications in Canada. *Renewable and Sustainable Energy Reviews*, 8(6), 493-519.
- Khan, M. I. Chhetri, A. B. Islam, M. R. (2007). Community-based energy model: A novel approach to developing sustainable energy. *Energy Sources Part B Economics Planning and Policy*, 2(4), 353-370.
- Khan, M. I., and Islam, M. R. (2003a). Wastes management in offshore oil and gas: A major challenge in integrated coastal zone management. In: CARICOSTA 2003—1st International Conference on Integrated Coastal Zone Management ICZM, L. G. Luna (ed.). University of Oriente, Santiago du Cuba, May 5–7.
- Khan, M. I., and Islam, M. R. (2003b). Ecosystem-based approaches to offshore oil and gas operation: An alternative environmental management technique. Annual Technical Conference and Exhibition, Society of Petroleum Engineers, Denver, CO, SPE Paper: 84058.
- Khan, M. I., and Islam, M.R. (2004). Assessing environmental fate and behavior of oil discharges in marine ecosystem: Using fugacity model. In: Proceedings of the Offshore Oil and Gas Environmental Effects Monitoring Workshop: Approaches and Technologies, Bedford Institute of Oceanography, (2003). S. L. Armsworthy, P. J. Cranford, and K. Lee (Eds.). Battelle Press, 145–165.
- Khan, M. I., and Islam, M. R. (2005a). Environmental modeling of oil discharges from produced water in the marine environment, Paper no.: GC-225. 33rd Annual General Conference of the Canadian Society for Civil Engineering, Toronto, Ontario, Canada.
- Martinot, E., and McDoom, O. (2000). *Promoting energy efficiency and renewable energy: GEF climate change projects and impacts*. Retrieved 6/2/2008, from <a href="http://66.102.1.104/scholar?hl=en&lr=&q=cache:7DAQ1yGYfeIJ:www.martinot.info/Martinot\_McDoom\_GEF.pdf+">http://66.102.1.104/scholar?hl=en&lr=&q=cache:7DAQ1yGYfeIJ:www.martinot.info/Martinot\_McDoom\_GEF.pdf+</a>
- Natural Resources Canada. (2008). Climate change plan for Canada. Retrieved 5/9/2008, from <a href="http://www.nrcan.gc.ca/es/etb/cetc/combustion/co2network/pdfs/climate change plan for c\_anada\_e.pdf">http://www.nrcan.gc.ca/es/etb/cetc/combustion/co2network/pdfs/climate change plan for c\_anada\_e.pdf</a>
- Natural Resources Canada. (2006). *About wind energy*. Retrieved 6/3/2008, from <u>http://www.canren.gc.ca/tech\_appl/index.asp?CaID=6&PgID=232</u>
- Pneumaticos, S. (2003). *Renewable energy in Canada: Status report 2002* Ottawa: Natural Resources Canada, c2003.

Porter, P., Barry, J., Samson, R., & Doudlah, M. (2008). *Growing Wisconsin energy; A native grass pellet bio-heat roadmap for Wisconsin*. Agrecol and DATCP Publication.

- Statistics Canada. (2006). Riding the wind. Retrieved 6/3/2008, from http://www41.statcan.ca.proxy1.lib.umanitoba.ca/2006/1741/ceb1741\_002\_e.htm
- Turton, H., Hamilton, C. (2002). Updating Per Capita Emissions for Industrialized Countries. The Australian Institute. <u>https://www.tai.org.au/documents/downloads/WP37.pdf</u>
- World Energy Council. (2001). *Survey of energy resources 2001*. United Kingdom: World Energy Council.
- Wu, D. W., and R. Z. Wang. (2006). Combined Cooling, Heating and Power: A Review. Progress in Energy and Combustion Science, 32.5-6: 459-95.

# **CHAPTER 3: RESEARCH METHODS**

### **3.1 Introduction**

The research method primarily consists of two parts: 1) demand reduction analysis and 2) alternative energy feasibility study. Figure 3.1 outlines the steps in the method used to determine the demand reduction, considering no cost and low cost options.



# Figure 3.1 Methodology to Determine Demand Reduction

RETScreen<sup>®</sup> International 4.0 was used to model renewable energy feasibility. Figure 4 outline the steps of the study method used for renewable energy analysis using RETScreen 4.0 energy modeling software.



# Figure 3.2 Study Methodology to Analyze Renewable Energy Technologies

Due to lack of energy meters at ELA, the researcher had no access to actual energy use at the study site. However, previous energy evaluation reports provided by Department of Fisheries and Oceans (DFO) had some information about the location and purpose of the facility.

#### **3.2 Research Steps**

#### 3.2.1 Energy Efficiency Analysis

In order to perform the energy efficiency analysis a walk through survey (See Appendix – I) was conducted in selected high energy consumption buildings suggested by the building manager. The survey (Refer to Appendix – I) is a modified version of Manitoba Hydro for office buildings was used to gather all the information about the energy generation and building and equipment usage. As part of this research a NRI class of eight students was enlisted to perform a detailed walk through survey at the field station by Dr. Thompson with my assistance. The walk through survey recorded all the various loads in the building and other important parameters required for the analysis, namely:

- Bulk fuel analysis
- Lighting
- Building Envelope
- HVAC
- Office and Lab Equipment
- Miscellaneous Equipment

Once the survey data was gathered it was compiled to identify opportunities that fit the research objectives outlined i.e. low-cost and no-cost retrofit opportunities for energy efficiency. Relevant opportunities were selected based on the following criteria that:

- initial investment for any single retrofit recommended should be less than
   \$ 2,000.00
- pay-back period for the recommended retrofit should be less than 2 years
- low-skill level required to carry out the retrofits
## 3.2.2 RETScreen 4.0 Analysis

RETScreen is a comprehensive product of its kind, allowing engineers, architects, and financial planners to model and analyze any clean energy project. Decision-makers can conduct a five step standard analysis, including energy analysis, cost analysis, emission analysis, financial analysis, and sensitivity/risk analysis. This standardized and integrated renewable energy project analysis software evaluates the energy production, life-cycle costs and GHG emission reductions for various types of RETs. RETScreen is used by more than 240,000 people in 222 countries and territories, thus proving to be a very accurate and efficient tool for RET analysis. Though there are other RET analysis software available, RETScreen's suitability to allow for off-grid power generation set-up makes it a preferred tool. The model's meteorological inputs are shown in Table 3.1.

The other suitable program available for RET analysis is HOMER. HOMER is a stand alone program, and as such it can handle a much denser simulation. While RETScreen splits the model into monthly chunks, HOMER can handle fluctuations on an hourly basis. This makes HOMER useful for modeling the intermittency of solar and wind power. HOMER is also capable of doing brute-force system optimization, given a number of variables. While HOMER is more powerful than RETScreen, it requires much more in the way of data inputs. Since, ELA was not metered and there was no utility power data available, RETScreen 4.0 provided the better choice for this simulation. Also, the economic modeling is the strength of RETScreen which is better than that of HOMER.

The RET Analysis is carried out in four steps as laid out in this section. RETScreen International  $4.0^{\text{®}}$  was used to compare the feasibility of three different RETs to diesel

generation. RETScreen is a renewable energy decision-support and capacity-building tool developed by Natural Resources Canada (NRC) with the contribution of 85 experts including from United Nations Environment Programme (UNEP) and the National Aeronautics & Space Administration (NASA<sup>®</sup>). The computer program, RETScreen<sup>®</sup> 4.0, provides a common platform ideal for educational purposes and industry/market analysis and development purposes and is free of charge (Ackermann et.al, 1999).

Month	Daily Solar Radiation (kWh/m²/d)	Mean Temp (°C)	Wind speed @ 10m (m/s)
Jan	1.48	-17.8	3.9
Feb	2.51	-14.1	3.9
Mar	4.12	-6.2	4.2
Apr	5.35	3.3	4.4
May	5.96	11.2	4.2
Jun	6.01	16.4	4.2
Jul	5.99	19.6	3.9
Aug	5.01	17.9	3.9
Sep	3.43	11.8	4.2
Oct	2.19	5.5	4.4
Nov	1.36	-4.5	4.2
Dec	1.13	-14.4	3.9
Annual Avg.	3.72	2.5	4.1

 Table 3.1 Meteorological data of ELA field station for the year 2007

Source: NASA<sup>®</sup> Global Climate 2007

The following four steps were applied in the RETScreen analysis:

**Step 1:** Evaluated the present energy, economic and environmental situation by referring to ELA fuel bills, manuals, and audit reports. Data on diesel, propane, and gasoline consumption were gathered from the facility log books. Preliminary data about installed electrical and mechanical equipment were gathered from

manufacturer's manuals, previous studies and interviews with the field manager and other key personnel at ELA. Data from different consultant's reports on alternative energies were gathered (Research Facilities, 2008; Planning Study for Experimental Lakes Area filed station for Public Works and Government Services Canada, 2003; Phase I/II Environmental Site Assessment, 2002) but none of these reports considered biomass or demand reduction.

**Step 2:** Performed a modified Manitoba Hydro energy efficiency audit of the field station that involved lighting efficiency and determining equipment loads for six buildings. This audit included interviews and a walk through tour noting characteristics, usage and amounts of: 1) bulk fuel use; 2) building envelope (quality of sealing of windows, doors); 3) lighting (load); 4) heating, ventilation and air conditioning (kW/hr); 5) office and lab equipment (usage and quantity); and 6) other machines and equipment (usage and quantity). The current energy consumption was estimated from the audit, as no metering was installed on site, to identify direct low cost energy conservation measures.

**Step 3:** Applied the present-day load of 115 kW minus the 4.8% energy efficiency determined in step 2 to RETScreen. RETScreen analysis was undertaken to ascertain the technological, cost, emissions and risk analysis on the three RETs namely wind, solar and biomass.

**Step 4:** The three different RET scenarios were calculated based on parameters in Table 2. As well the following were considered:

- local climatic data (solar radiation, wind speed, ambient air temperature, humidity)
- 2. the assumption that any new load will be balanced by increased energy efficiencies

- 3 diesel price of \$0.80/liter, propane price of \$0.45/liter
- 4 higher heating value (HHV<sup>4</sup>) setting is used as ELA is in a northern climatic zone with an average winter temperature of -17.4°C and reaches above 30°C in the summer for reference year 2006. The occupancy varies from full capacity of 40 to 45 people in the summer to about 3 to 5 people in winter.

<sup>&</sup>lt;sup>4</sup> *ELAs geological location requires the model to be simulated in a higher heating value setting.* 

# CHAPTER 4: DEMAND REDUCTION FINDINGS REGARDING LOW-COST AND NO-COST RETROFIT ALTERNATIVES

## 4.1 Introduction

Energy efficiency retrofits of existing buildings are the best way to reduce energy use and greenhouse gas emissions in off grid communities. Although some buildings may have had some efficiency upgrades in the past, there is almost always room for improvement. New technologies, increased awareness and changing energy prices often mean that measures rejected in the past as not being cost-effective are viable today. The rationale for an energy retrofit can go far beyond just energy savings. Energy retrofits often introduce new technologies or operating controls that can improve occupant comfort. New, energy efficient T8 lighting is a good example of this, providing reduced flicker and hum with improved light quality that can reduce occupant eyestrain. Energy retrofits also provide an opportunity to replace aging equipment, down size equipment and repair or upgrade old systems.

In this chapter we are concerned only about the *Operating Energy* of the ELA buildings. Building *Operating Energy* is the energy associated with the normal operation of the building for space heating, domestic water heating and operating lights and appliances. Operating energy is to be impacted by retrofit measures. The main factors that affect operational energy of buildings are location, occupant density, occupant behavior and building technology. Retrofits to improve the operational efficiency of buildings generally relate to: building envelope, glazing and door technologies, higher efficiency space and water heating system and appliance upgrade *or* downsize.

## 4.2 Energy Retrofits of Existing Buildings

Energy efficiency improvements are the most cost-effective way to reduce greenhouse gas emissions and help reduce energy demand. Many energy efficiency measures yield great returns on investment and can be repaid within one to three years. Unlike most capital projects, energy efficiency projects provide a monetary return through the energy savings they generate. This means they can be viewed as investments, rather than as simply expenditures (CAEE, 2007). As for any large investment, when investing in energy efficiency it is appropriate to perform a full lifecycle cost analysis (CAEE, 2007). A life cycle cost analysis takes into account the energy savings over the life of the project, deferred maintenance and equipment replacement costs. In spite of being large financial investments, energy projects are often assessed solely in terms of simple payback (cost divided by annual savings), with expectations that paybacks will be very short. This type of analysis does not reflect the true long-term value of a project. More sophisticated indicators such as net present value (NPV) and internal rate of return will more accurately reflect the benefit of the investment. When fully accounted for over their life cycle, projects with simple paybacks as long as 15 to 20 years may still show a positive net present value and be a good investment. However, the manager at ELA mentioned that they are interested in short-term and minimum investment energy efficiency improvements and that they require basic indicators like payback period and energy cost savings to move ahead. The following section deals with retrofits that are Low-Cost or No-Cost, discussing their suitability for ELA.

## 4.2.1 Low-Cost and No-Cost Retrofits Vs Comprehensive Retrofits

Small communities and companies undertaking energy retrofits tend to seek quick payback measures in order to keep project costs down. Comprehensive retrofits are another consideration with additional advantages beyond financial returns. As they

involve all the civic facilities, they result in a coordinated approach throughout. This means consistent lighting and control systems as well as consistent documentation. They may also result in reduced workload for staff, as the project is completed quickly rather than ongoing for years. And the scale of the project will result in lower costs, both in construction capital and engineering design. Although the initial payback may be longer, a comprehensive retrofit will usually have a better financial return when looked at over the life cycle cost. However, for ELA, given the nature of operation and occupancy levels a Low-Cost & No-Cost retrofit option approach makes more sense both financially and operationally. Table 4.1 compares both types of retrofits and their suability to ELA circumstances.

Low-Cost & No-Cost Retrofits	<b>Comprehensive Retrofits</b>	
Initial Cost: Zero or Very Low in most	Initial Cost: High Initial Cost. Most cases	
cases. Usually ranges between a few	require a significant initial amount in the	
hundreds of dollars to a few thousand	range of few thousands of dollars to a	
dollars.	hundred of thousands of dollars.	
Payback Period: Short to Medium	Payback Period: Medium to Long	
payback period. Ranges from few months	payback period. Can range from 4 to 8	
to up to 3 years	years based on the retrofit.	
Annual Savings: Low	Annual Savings: High	
Easy to implement and monitor.	Implementation and monitoring require a	
	significant time and resources.	
Skill Level Required: Low	Skill Level Required: Medium to High or	
	Professional	

Table 4.1 Comparison of Low-Cost & No-Cost Retrofits Vs Comprehensive Retrofits

Source: Community Action on Energy and Emissions (CAEE) manual, 2007

## 4.3 Base Case Characteristics

Diesel, Propane, and gasoline data were gathered from the facility for the past available three years and a base-case scenario was developed to evaluate against wind, solar and biomass analysis. Preliminary data has been gathered about site and installed electrical and mechanical appliances from manufactures manuals, previous studies and interview with field manger and other key personnel at ELA. A walk through survey (see Appendix-I) was conducted at the facility buildings to estimate the current energy consumption and identify. The survey also estimated the annual energy demand at site and annual base load profile. Table 4.2 identifies the general characteristics of ELA.

Summer	Characteristic	Winter
April-Oct	Duration	Nov-Mar
35 to 40	Occupancy (No of persons)	4 to 5
All	No. of buildings in use	1 to2
Maximum (~100 kW)	Power demand	Minimum (~25 kW)

Table 4.2 General Characteris	ICS

Summer to winter inhabitants' ratio: 8.75

Source: Personal Communication with Ray Pambrun ELA, 2007

# 4.4 Demand Reduction at ELA

Energy management at off grid locations is an important issue driven mainly by economical factors like fuel cost, equipment efficiency, operation and maintenance cost. These issues concern the supply side management. Apart from supply side management there are also issues like poorly insulated buildings, inefficient lighting and equipment which further constitute to the poor overall energy situation.

# 4.4.1 Current Energy Sources and Distribution

Overall the fuel profile consists of 61% diesel, 33% propane and 6% gasoline as shown in Figure 4.1.

Fuel used	Purpose	Fuel cost (\$/L)	Annual Consumption (L)
Diesel	Electricity	0.80	112,151.10
Propane	Heat (space and water	) 0.47	60,377.70
Gasoline	Transportation	0.88	11,531.20

Table 4.3 Fuel characteristics of Existing Energy System (2006)

Sources: Personal Communication with Ray Pambrun, ELA 2007



Figure 4.1 Fuel Consumption Breakdown at ELA

# 4.4.2 Cost of Electrical energy derived from diesel generators

ELA has an average electrical load of approx. 50 to 60 kW/hr; with peak supply close to

100 kW during maximum occupancy from April to October.

Average weekly consumption of diesel fuel: 2,600 liters

Cost of diesel (per liter): \$0.80

Based on the above information, I calculated the following:

- Daily diesel consumption: 2,600/7 = **371.43 lit/day** (or 11,142.66 lit/month)
- Cost of diesel per month: 11,142.66L X 0.80 = **\$8,914.29** (or \$106,971.48 per year)

- Total kWh supplied in a month: 55kw X 720hr = **39,600 kWh** 

- Cost per kWh = (\$8,914.29 / 39,600) = 0.225 \$/kWh

The annual cost of running the Laboratories for 2006/2007: 63% X \$65,937.60 (average annual diesel cost) = **\$41,540.68/yr** (varies with diesel price).

Grid Type & Technology	Off grid / Reciprocating engine	
Fuel Type & Cost (\$/L)	Diesel @ 0.80	
Capacity (kW)	115	
Heat Rate5	11,000 (or)	
(kJ/kWh)	~25% efficient	
Electricity rate		
(\$/kWh)	0.225	

 Table 4.4 Base-Case Characteristics

## 4.5 Lighting System

Lighting constitutes 22% of the total power consumed by the building and perhaps this is the simplest and easiest area to make necessary changes without affecting the operations in any way and without major renovation.

## Recommendations:

 Two potential areas have been identified where small modifications can result in reduction of energy cost for the laboratory with a reasonable payback period for lighting. Changing the existing exit lamps with LED backlights next time would result in savings of approx. \$289.43/yr. (Ref Table.4.5).

<sup>&</sup>lt;sup>5</sup> Heat Rate is the amount of energy input (in kJ or Btu) from the fuel required to produce 1kWh of electricity (RET Screen, 2008)

	Annual Expenditure (\$)
5 X 30 watt incandescent	434.15
Replace with 5 X 10 watt LED	144.72
Potential Savings	289.43
Approx. Capital cost of replacement	225.00
@ \$45/LED X 5	223.00
Pay Back	< 1 yr

# Table 4.5 Replacement costs for upgrading existing exit lamp with LED

- 2. Occupant behavioral changes will also contribute to the energy conservation. Fans and lights in the laboratory
- 3. Lights in laboratory were observed to be on without anybody using them. New people visit ELA every year and if they were educated about this facility as a high energy conservation facility this would mould their behavior with ease

# 4.6 Laboratory Equipment

Specialty equipment could provide energy conservation opportunities in three ways:

- 1. Replacement of existing equipment with more energy efficient equipment
- 2. Downsizing to suite the task 3. Efficient usage

## Recommendations:

 Down size the icemaker to a more appropriate scale to match lab needs and if possible decommission the unit during winter. The existing ice maker is of 1100 watts with an annual operating expenditure of about \$1215.00.

	Annual Expenditure (\$)	
Ice Maker (1100watts)	1,214.14	
Replace with	6 34 66	
Ice Maker (575 watts)	0,34.00	
Potential Savings	579.48	
Approx. Capital cost of replacement	2,500.00	
Pay Back	4.3 yrs	

 Table 4.6 Potential savings by switching to a smaller ice maker

- 2. When replacing or upgrading any existing laboratory equipment consider buying energy efficient equipment and make it a priority.
- 3. The two ovens are operated exclusively at 90°C while the units are clearly oversized as they can heat to over 600°C. Also it's known that ovens are run overnight irrespective of their need thus increasing the annual operating cost. A simple solution would be to just turn OFF one of the ovens overnight and leave only one of them ON so that it can be used. The following table explains the annual cost that can be saved. (Refer to Table 4.7)

Table 4.7 Potential savings by turning one of the 600° C ovens OFF overnight

	Annual Expenditure (\$)
2 X 1300 watt	1,749.20
1 X 1300 watt (one off overnight)	1,311.90
Potential Savings	437.30
Capital Cost	0.00

4. Down size the oven to a more appropriate scale to match lab needs and if possible decommission the unit during winter. The existing ovens are of 1300 watts each with an annual operating expenditure of about \$1,725.87. By downsizing to a new smaller oven of 350 watts the potential energy savings are \$1,250.93. The cost of new equipment is estimated to be \$1,425.71 with a pay-back period of 1.1 years. (Refer to Table 4.8)

	Annual Expenditure (\$)
2 X 1300 watt 90° C	1,721.87
1 X 350 watt	470.94
Potential Savings	1,250.93
Approx. Capital Cost of new equipment	1,425.71
Payback period	1.1 yrs

Table 4.8 Energy cost saved by	downsizing to a smaller	more efficient oven
--------------------------------	-------------------------	---------------------

# 4.6 Building Envelope

The chemistry laboratory building surveyed was built with many energy efficient features. However, the building envelope has the potential to be further improved in the areas of insulation and better door drafts insulation.

# 4.6.1 Air Leakage

Air leakage from the building due to improper sealing can increase the heating or cooling cost by up to 25%. Figures 4.2 and 4.3 expose the cracks and incomplete air sealing found in the building. Air leakage through these cracks and holes can increase the heating or cooling cost for the building (BC Hydro, 2004a).

# Recommendations:

- 1. Caulking any openings and cracks is easy and a low cost approach to prevent heat loss through them thus saving substantial amounts of heating or cooling bill.
- Other option is aerosol foam sealants; these can be used for sealing larger cracks and openings around pipe and wire penetrations and at the foundation sill joint. These foams provide a good tight seal by expanding in the cracks.



Photo: Daniel Gagne 2007

# Figure 4.2 Cracks found on the outer wall in the Laboratory building



Photo: Daniel Gagne 2007

# Figure 4.3 Air Conditioning cable holes can be better insulated in laboratory building

# 4.6.2 Insulation

Incomplete insulation work in the basement can cause heat loss through the walls.

Figures 4.4, 4.5 and 4.6 show where improvements can be made.



Photo: Daniel Gagne 2007

# Figure 4.4 Indicates that basements can still be improved for better sealing



Photo: Daniel Gagne 2007

Figure 4.5 Potential areas where basement insulation can be improved



Photo: Daniel Gagne 2007

# Figure 4.6 Outside vent foundation with exposed wood foundation

# <u>Recommendations:</u>

- The chemistry lab basement is not using rigid foam board insulation, extending it down below the frost line to about 0.6 meters (2 feet) will greatly reduce the basement losses (BC Hydro 2004b)
- 2. For insulating the exterior using several inches of foam board insulation (enough to achieve R-12) down to the footing should give adequate insulation.

# 4.6.3 Infiltration Losses

Sealing joints and surfaces that move, such as where doors or windows meet frames, is a huge concern for any building manager. As seen in Figure 4.7 and 4.8 the entrance and exit doors of the building had brush insulation and a draft was palpable, indicating the need for better insulation (BC Hydro, 2004b)



Photo: Daniel Gagne 2007

Figure 4.7 Doors were found to have been installed with brush insulation these could be replaced with magnetic strips that provides better insulation



Photo: Daniel Gagne 2007

## Figure 4.8 Other doors where drafts were noticed

# Recommendations:

- Using weather stripping techniques can mitigate drafts. Vinyl V-strip and spring metal weather stripping have a life span of at least five years. All exterior doors or doors between conditioned and unconditioned spaces such as basements, attics, or garages can benefit from the application of weather-stripping. The air-conditioned "clean room" door in the chemistry lab should also be considered.
- 2. Incorporating magnetic weather-stripping, similar to the seal on a refrigerator door can cut air leakage significantly.

## 4.7 Chapter Summary

In this chapter I have looked at achieving energy efficiency through low-cost and no-cost retrofits. A walk through survey has been conducted at the study site to identify all the potential opportunities for energy efficiency at four key high-occupancy and high-usage buildings. The data from the survey has been analyzed for potential energy savings, payback period, and d capital cost for all the recommendations. Table 4.9 summarizes the potential savings and associated with the potential retrofits.

Considering the situation and circumstances at ELA an attractive index based on payback period has been developed that helps decide on retrofits that can be done with minimal cost and low skill level. All the retrofits that have a payback period of less than or equal to two years and that require a capital cost of less than \$2,000.00 are considered to be high priority. For example, problem identified No. 2 in the Table 4.9 is all the seven exit lamps are incandescent lamps, the recommended retrofit is to replace all the seven exit lamps with LED exit lamps. LED exit lamps use considerably minimal power for operation compared to the existing incandescent lamps. The estimated energy savings from this retrofit are 613.2 kWh and the estimated cost savings are \$153.00. The payback for this retrofit is approximately two years. Because this retrofit complies with the preset attractiveness index rules, its attractiveness index is rated as **High**. Similarly, all the problems identified from the walk though survey are analyzed and are presented in the Table 4.9 with their respective attractiveness index.

Problem Identified		Potential Retrofits	Capital cost of recommen dation	Energy Savings in kWh	Estimated cost savings per year	Pay Back period	Attractiv eness Indicator
1. Lighting in the laboratory		Replace the 5 existing incandescent lights with CFL (Compact Fluorescent Lamps)	\$35	396	\$99.00	Under 4month s	High
2.	Exit lamps	Replace all 7 existing exit lamps with LED exit lamps	\$315	613.2	\$153.00	approx 2years	High
3.	Two, old 40 cubic foot refrigerator s	Replace both with Energy star units	2 X apprx. \$7500.00 = \$15000.00	$20\%^{6}$ saving on 9,066.6 (existing) = 7,253.28, A saving of 1813.32	\$518.00	>20yrs	Low
4.	Two small refrigeratio n units	Replace them with a single bigger energy star unit	approx \$7500.00	20% savings on 3679.2 (existing) = 2943.36, A saving of 735.84	\$762.00	10yrs	Low
5.	Oversized ice maker (1100 Watts)	Downsize to a smaller (575 watts) unit	approx \$2500.00	50% savings on 4876.7 = 2438.35, A saving of 2438.35	\$580.00	4.3yrs	Medium
6.	Two 90°C	a) Turn off one oven during nights	Nil	25% savings on 6832.8 = 5124.6, A saving of 1708.2	\$437.00	nil	High
	ovens run continuou sly			(OR)			
	overnight	b) Unplug one 1300 watts oven and downsize the other to a 350 watts unit	approx \$1425.00	75% savings on 5142.6 = 1285.6, A saving of 3856.95	\$1,250.00	1.1yrs	High
7.	Organic waste incineratio n generates smoke and	Setup bear proof garbage bins and ship waste to the	approx \$250-\$350 ranging from	n/a	Safer disposal of waste resulting in a cleaner	n/a	

Indicator

<sup>&</sup>lt;sup>6</sup> Energy star appliances save 20% of standard equipment

particulate matter affecting the air quality samples	nearest landfill	64 to 95 gallons		environment at ELA		High
8. Infiltration losses at doors (Improper door seals)	Use weather stripping techniques to mitigate this problem. Replace existing brush insulation strips with magnetic or metal strips	Less than \$25 per door	n/a	n/a	Less than a year	High
9. Basement Insulation	Use rigid foam board insulation and extending it down below the frost line to about 0.6 meters (2 feet) will greatly reduce the basement losses (BC Hydro 2004b)	Between \$1.50 and \$2.00 per Sq. foot	n/a	n/a	Less than a year	High
10. Heat and power generati on	Switch to a greener and more efficient generation technology	further study under review	n/a	n/a	n/a	Low
11.Space heating	Geothermal heating analyzed, may not be suitable for ELA	approx \$26,000.00	n/a	\$6,696.28	4.7yrs	Medium
Total		\$51,628.00 (excluding the basement insulation cost and the cost a new downsized refrigerator)	7704.91	\$9,245.28 (excludes the savings from installing a new downsized refrigerator)	Avg. 5.4 yrs	

By adopting the recommendations with **High** attractive index in Table 4.9, ELA can easily identify energy savings with minimum investment and relatively quick payback period without significant labor and equipment costs. These retrofit measures have the combined potential of reducing the demand at ELA by 4.8%.

## Acknowledgements

I would like to thank to James Kornelson, Kent Pearce, Jessica Saunders, Godwin Chang, Daniel Gagne, Tyler Tarnoczi and Jeff Valdivia for their help in conducting the walk through survey.

## References

- CAEE, 2007. Energy Efficiency and Buildings: A Resource for BC's Local Government. A Manual on Community Action on Energy and Emissions.
- Martinot, E., and McDoom, O. (2000). Promoting energy efficiency and renewable energy: GEF climate change projects and impacts. Retrieved 6/2/2008, from <u>http://66.102.1.104/scholar?hl=en&lr=&q=cache:7DAQ1yGYfeIJ:www.martinot.info/Martinot\_McDoom\_GEF.pdf+</u>
- Research facilities. Retrieved 5/8/2008, 2008, from <u>http://www.dfo-</u> <u>mpo.gc.ca/regions/central/science/research-recherche/index\_e.htm#Experimental</u>

## **Personal Communications**

Beaty, Ken. Hydrologist. Environmental Science Division. Fisheries and Oceans Canada. ELA Facility, Ontario. July, 2006

Jordan, Duane. Project Manager. Fisheries and Oceans Canada. Fresh Water Institute. July, 2006

Pambrun, Ray. Field Station Manager. Fisheries and Oceans Canada. ELA Facility, Ontario.

September 2006

Shearer, John. Fisheries and Oceans Canada. ELA Facility, Ontario. July, 2006

# CHAPTER 5: FEASIBILITY STUDY OF RENEWABLE ENERGY TECHNOLOGIES AT ELA

## **5.1 Introduction**

Since the early 1990's there have been significant developments in various RETs for commercial, industrial and residential sectors making them ever more competitive with fossil fuels. RETs have advantages over non-renewable energy technologies that include low energy-cost, oil independence and pollution free generation, but also have disadvantages, that include reduced reliability and high initial costs. Energy sources, such as wind and solar require back-up by a stable source, if there is no grid connection (due to their intermittent nature) to guarantee reliable power supply (Weis et.al, 1998). A reserve capacity is necessary to act as a back up to overcome fluctuations and reliability issues with wind and solar intermittent sources that don't generate energy when the wind is not blowing or the sun is not shining (Ah-You, 1999). Although RETs do not burn fossil fuels, they often require back-up systems that do. All renewable energies require that resource availability be compared to the loads to determine if the site specific production meets the local need. RETs combined with energy storage systems provide a reliable energy supply is the highest priority in the design of an isolated power system (Weis et.al, 1998). Natural energy flows vary and make the techno-economic performance of renewable energy conversion highly site specific. There are a host of renewable energies, including wind power, solar PV, biomass, etc, but are any feasible at the ELA location? The benefits and applications of these RETs in Canada will be profiled to consider their feasibility.

Wind power is a clean renewable energy but is intermittent requiring wind-diesel hybrid systems to provide a stable capacity. At ELA, wind power is considered feasible because

mean annual wind speeds are5 m/s to 6 m/s. With a battery storage unit the hybrid wind power system, with a back-up diesel system, could mitigate diesel consumption by about 30% to 40% annually. Currently, solar PV modules with battery back-up are effective in meeting power needs in remote locations for homes, cottages as an alternative to installing new transmission lines or diesel generators. The solar resource in Canada compares favorably with other regions of the world, due in part to its ''clear-sky'' climate. At ELA, solar radiation is approximately 3.72 kWh/m<sup>2</sup>/day (Huang et.al, 2001). Although there are many possible applications of solar energy including water heating, passive heating and space cells made of semiconductor materials like silicon, can convert solar energy to electricity with 10% to 20% efficiency (Bernotat et.al, 2004). Solar generation is a good match to energy demands at ELA as during summer; when ELA has the highest power demands, extended daylight hours of as much as 17 hours produce the maximum power; while in winter shorter daylight hours produce minimum power when power loads at ELA are small.

Wood based energy generation units can use the surplus bio-residue to produce heat and power simultaneously in a system called combined heat and power (CHP). In biomass based CHP, both heat and power are generated from biomass with a back-up system of diesel generators to handle peak load demands. Biomass resources are typically forestry products such as wood waste or wood pellets but can include agricultural residues, landfill gas, municipal solid wastes and energy crops. Small scale biomass CHP have been used extensively in space and water heating for housing, process heat for industry since the 1940's in Sweden, Finland and other Baltic states like Latvia, Estonia, and Lithuania (Sims et.al, 2003). Approximately 6% of Canada's primary energy is from biomass energy in the form of combustion of wood and wood derivatives for industrial

process heat, generation of electricity, and space heating. Canada, with over 2.4 million km<sup>2</sup> of forests, has many of its remote areas surrounded by forest, a renewable source of energy (Mustafa Omer, 2008). At ELA, dead wood from the nearby forest could provide sufficient biomass and their collection would reduce the risk of forest fire and reduce the cost of maintenance for fire suppression system and clearing cost.

## **5.2 Findings**



# Figure 5.1 Energy Map of ELA revealing the flow of energy through the facility for the 2006/2007 fiscal year

The demand for electricity could be reduced with energy an efficiency measure that decreases the cost of capital equipment. Figure 5.1 is an energy map summarizing the annual total input and output energy at ELA for the 2006/2007 fiscal year. The existing system has a total peak power generation of nearly 115 kW and a total operating cost of \$84,821/yr. Most of GHG emissions and energy costs at ELA are for electricity production from diesel fuel. Only one-fifth of the fuel costs are from heating with propane.

# 5.2.1 Energy Efficiency

With almost majority of fuel being consumed for electricity production, demand reduction can make a difference in energy requirements. Table 5.2 shows many opportunities to reduce energy, identifying key areas for energy measures including changing every exit light in the 20 ELA buildings to light emitting diode (LED) lights, changing incandescent lights to compact fluorescents, changing old refrigerators and ice maker to energy efficient types and turning off ovens, that run continuously, although almost empty. Demand reduction can reduce energy demand by up to 4.8% of the existing energy consumption, which amounts to 14,130.37 kWh or about \$2,567.32 in savings annually. Table 5.2 identifies all the recommendations that fit the selection criteria for energy efficiency retrofits.

Problem Identified		Recommendation	Capital cost of recommen dation	Energy Savings in kWh	Estimated savings per year	Pay Back period	Attracti veness Indicat or
1.	Lighting in the laboratory	Replace the 5 existing incandescent lights with CFL (Compact Fluorescent Lamps)	\$35	396	\$99.00	Under 4month s	High
2.	Exit lamps	Replace all 7 existing exit lamps with LED exit lamps	\$315	613.2	\$153.00	approx 2years	High
3.	Two 90°C ovens run continuous ly overnight	a) Turn off one oven during nights	Nil	25% savings on 6832.8 = 5124.6, A saving of 1708.2	\$437.00	nil	High
				(OR)			
		b) Unplug one 1300 watts oven and downsize the other to a 350 watts unit	approx \$1425.00	75% savings on 5142.6 = 1285.6, A saving of 3856.95	\$1,250.00	1.1yrs	High

 Table 5.1 Selection of Demand Reduction Recommendations with High

 Attractiveness Index

4.	Infiltration losses at doors (Improper door seals)	Use weather stripping techniques to mitigate this problem. Replace existing brush insulation strips	Less than \$25 per door	n/a	n/a	Less than a year	High
		with magnetic or metal strips					
5.	Basement Insulation	Use rigid foam board insulation and extending it down below the frost line to about 0.6 meters (2 feet) will greatly reduce the basement losses (BC Hydro 2004b)	Between \$1.50 and \$2.00 per Sq. foot	n/a	n/a	Less than a year	High

The demand reduction savings are expected to be higher as not all buildings were analyzed. Other areas such as the building envelope appeared adequate.



Figure 5.2 Electrical energy breakdown indicates that HVAC and lighting combined consume 2/3<sup>rds</sup> of the total power

RET analysis results for the three models are shown in Table 5.5 along with existing diesel generator and new replacement diesel generator. Renewable energy technologies were economically competitive with the diesel system, particularly the biomass CHP system. At \$0.80 per liter, biomass combined heat and power (CHP) payback period was 4.1 years with a capital cost of \$2,162.9/kW compared to wind's 6.1 years due to its higher initial cost of \$3,300/kW and solar energies' 13.5 years due to its high initial cost

of \$ 9,100/kW. The CHP had an initial cost for equipment (hopper, conveyor belt and gasifier but not including piping for district heating) at \$ 2,162.9/kW with an energy cost of 0.12 \$/kW. The payback for CHP is much less at higher diesel prices of \$1.20/liter and \$2.00/liter respectively at 1.8 years and 0.9 years. Table 5.1 shows the economic and financial parameters for the model. Please refer to Appendix – II for RETScreen Analysis.

Parameter	Wind-Diesel Hybrid	Solar PV-Diesel Hybrid	Biomass
Fuel cost-Proposed case (\$/yr	) 41 449	56 139	63 729*
Fuel cost-Base case (\$/yr)	98 067	97 071	151 881*
Debit Ratio (%)	75	75	75
Debt Interest rate (%)	7.00	7.00	7.00
Debt Term (Yrs)	10	10	10

**Table 5.2 Economic and Financial parameters for RETs** 

\* Includes propane heating cost

When a liter of diesel approaches \$1.20, power generation by diesel generation costs as much as \$0.70 per kW. When the price of diesel is at \$2.00/l the cost of electricity from diesel is approaching \$0.89/kWh. This is twice as expensive as wind generation, seven to eight times as expensive as biomass generation and about nineteen times as expensive as solar power per kWh. Other fuels become affordable and the payback periods are reduced at these higher diesel prices. At \$1.20/kWh to \$2.00/kWh for diesel, the payback periods (years) of different RETs are, respectively: 1.8 years to 0.9 years for CHP, 3.6 years to 1.8 years for wind, and 6.7 years to 3.2 years for solar.

RETs will reduce greenhouse gas (GHG) emissions considerably by an estimated 187 tons per year by a wind-diesel hybrid system, 134 tons per year by a solar PV-diesel

hybrid system and 497 tons per year by the biomass system. Biomass CHP is a reliable technology and burning wood is considered a sustainable cycle as the carbon burned will be used up when new trees replace them, as long as the forested area's land use is unchanged from forest. Concerning power generation, the existing diesel system is inefficient at ~25% and is expensive to generate at \$0.23/kW, which is much higher than wind power at \$0.14/kW, solar power at \$0.045/kW and biomass at \$0.12/kW. Wind and solar technologies are considered to have zero emission technologies during power generation while both diesel and biomass emit pollution including GHG at the site of generation. However, biomass based power generation is a carbon neutral technology, as trees that replace those burned take up the carbon emitted during combustion if the land remains forest. Based on the initial cost for equipment, solar PV at \$9,100/kW is the most expensive among the technologies, requiring a long payback period of 13.5 years. At \$3,300/kW, wind power is more expensive than installing a new diesel generator or implementing a biomass system. Heat is a byproduct of the biomass CHP system that could replace the propane expense of almost \$9,150 for heating the residences and laboratories. Geothermal reduces propane use by two-thirds typically by using the earth's heat, whereas CHP heat is waste heat, not requiring any additional fuel.

### 5.2.2 Impacts of RETs on GHG Emissions

The average annual gasoline expenses for the year 2006/2007 are approx. \$9,147.80 for transportation by automobiles (vehicles leaving to Fresh Water Institute (FWI) Winnipeg), out boats, ATVs (All Terrain Vehicles) and other equipment like portable generators. Switching from conventional automobiles to hybrid vehicles has the potential to reduce the gasoline cost and also would mitigate GHG emissions. Table 5.3 shows the GHG emissions from the three different types of fuels consumed at ELA.

Fuel	$CO_2$ (Carbon dioxide)	$CH_4(Methane)$	$N_2O$ (Nitrous Oxide)
Diesel	2.730 kg/l	0.00013 kg/l	0.0004 kg/l
Propane	1.500 kg/l	0.000024 kg/l	0.000108 kg/l
Gasoline	2.360 kg/l	0.00025 kg/l	0.00026 kg/l

Table 5.3 GHG emissions from fuels combusted at ELA

Source: Community Energy Planning, A resource guide for remote communities in Canada 2005

## GHG Emissions at ELA for year 2006/2007 based on Table 5.3

From Diesel: 82,422\*2.730= 225,012.06kg CO<sub>2</sub>

82,422\*0.00013= 10.71kg CH<sub>4</sub>

82,422\*0.0004= 32.96kg N<sub>2</sub>O

From Propane: 21,586\*1.500=32,379.00kg CO<sub>2</sub>

21,586\*0.000024= 0.518kg CH<sub>4</sub>

21,586\*0.000108= 0.233kg N<sub>4</sub>O

From Gasoline: 9527\*2.360=22,483.72kg CO<sub>2</sub>

9527\* 0.00025= 2.38kg CH<sub>4</sub>

Total CO<sub>2</sub> Emissions: **279,874.78 Kg** 

Total CH<sub>4</sub> Emissions: 13.608Kg ~ 285.76Kg of CO<sub>2</sub>

(Note: Methane is 21 times more powerful than  $CO_2$  in terms of greenhouse effect)

Total N<sub>4</sub>O Emissions: 35.663Kg

Therefore, total GHG emissions from ELA are Approx. **280,410.54 Kg of CO<sub>2</sub> Eq per year**.

## **5.3 Chapter Summary**

Three renewable energy technologies (RETs) were analyzed for their feasibility for a small off-grid research facility dependent on diesel for power and propane for heat.

Presently, the electrical load for this facility is 115 kW but a demand reduction achieved through energy audit which revealed that 4.8% reduction was possible. This reduces the peak load to 110 kW which subsequently reduced the capital costs by \$27,000 for biomass, \$49,500 for wind and \$136,500 for solar.

The RETScreen International 4.0<sup>®</sup> model compared the economic and environmental costs of generating 125 kW of electricity for three RETs compared to the current (0 cost) and a replacement (160/kW) diesel equipment. Biomass was the most feasible at all the different diesel prices analyzed at. At 80 cents per liter, biomass' payback period was 4.1 years with a capital cost of \$ 2,162.9/kW compared to wind power payback period of 6.1 years due to its higher initial cost of \$ 3,300/kW. Solar PV had a payback of 13.5 years due to its high initial cost of \$ 9,100/kW. A biomass system would reduce annual energy costs by \$ 63,729 per year, and mitigate GHG emissions by over 98% to 10 t CO<sub>2</sub> from 507 t CO<sub>2</sub>. Diesel price increases to \$1.20 or \$2.00/liter will decrease the payback period in years dramatically to 1.8 and 0.9 for CHP, 3.6 and 1.8 for wind, and 6.7 and 3.2 years for solar, respectively.

Some RETs, particularly CHP at ELA, are feasible in off-grid communities, according to this study, and may soon be feasible in grid communities if fossil fuel prices increase. The utility of applying demand reduction prior to sizing RETs was demonstrated at ELA by reducing capacity from 115 kW to 100 kW, that reduced initial costs by \$26,000 for CHP, \$49,500 for wind and \$136,500 for solar. This study shows that demand reduction and RET can be applied effectively to dramatically improve the energy situation at ELA resulting in lower energy cost and cleaner energy production. Demand reduction had the potential of shaving 4.8% from the existing energy consumption, amounting to

14,130.37 kWh or \$ 2,567.32 in annual savings. Table 5.4 summarizes the three renewable technologies that are analyzed.

The table consolidates the model results for: 1) solar-PV – diesel hybrid, 2) wind-diesel hybrid, and, 3)biomass CHP system. For all these fuel sources the pay back period, cost of energy, reliability of the system, availability of the resource, capital cost, GHG emissions and annual fuel cost are compared. The pay back period on Solar PV hybrid is 13.5 yrs at \$0.80 per liter of diesel due to the capacity factor of the hybrid system. The hybrid system has been modeled at 10 (PV): 90 (Diesel) capacity factor and also the load in summer is low and in winter the load at station is only a tenth of the peak load in summer. As the diesel price increases the system pay back decreases and at \$ 2.00 per liter of diesel the payback is as low as 3.2 years. The wind-diesel hybrid system has been modeled at 30 (wind): 70 (diesel) capacity factor. This system has a pay back of 6.6 years at the diesel price of \$ 0.8 per liter and as the diesel price increases to \$ 2.00 per liter the pay back period reduces to 1.8 years. The biomass based CHP system modeled, however, is a stand alone system with the existing diesel generators as back-up. It has the shortest pay back period of the technologies modeled. The biomass based CHP system has a pay back period of 4.1 years at \$0.80 per liter of diesel, as diesel price approaches \$ 2.00 per liter the pay back period reduces to 0.9 years. All the systems show as the price of fossil fuel increases the hybrid RET systems become more feasible.

	Electricity		Heat & Power	Heating			
RET Para- Meters	Solar Power 125kW	Wind Power 125kW	Diesel Power 115kW (existing)	New Diesel Generators 115kW	Biomass 125kW	Geothermal Heating	Propane Heating
Reliability	Low-Moderate	Low- Moderate	High	High	High	High	High
Availability	~25-30%	~25%-30%	< 95%	<95%	< 95%	-	< 95%
Avg. Initial Cost (\$/kW)	9100	3300	-	~160	2162.9	n/a	n/a
Cost of power (in \$/kW)	0.045	0.145	0.225	0.200	0.120	0.083	0.454
Efficiency	12.3%	~30%	~25%	~30%	~85%	> 85%	~85%
Equity payback Period (yrs) at different diesel prices	13.5 @ \$0.8/1 6.7 @ \$1.20/1 3.2 @ \$2.0/1	6.6 @ \$0.8/1 3.6 @ \$1.20/1 1.8 @ \$ 2.0/1	-	-	4.1@ \$0.8/1 1.8@ \$1.2/1 0.9@ \$2.0/1	4.7	-
Capital Cost (\$)	1,137,500	412,500	-	16 000	270362.50 without district heating network	26 000	-
Annual fuel cost (\$) at different diesel prices	Nil	Nil	\$65937.60 @ \$0.8/1 \$98906.40 @ \$1.2/1 \$164 844.00 @ \$2.0/1	\$65937.60 @ 0.8/1 \$98906.40 @ \$1.2/1 \$164 844.00 @ \$2.0/1	~6500.00	3039.00	\$9735.28 @ 0.8/1 \$25 903.20 @ 1.2/1 \$43 172.00 @ 2.0/1
GHG Emission reduction (t CO <sub>2</sub> )	187	134			497		
Carbon tax savings @ \$10/ton/year	\$1870	\$1340			\$4970		
Carbon tax savings @ \$50/ton/year	\$9350	\$6700			\$24 850		

Of the three RETs analyzed, biomass was found to be more economically and environmentally feasible than wind and solar for ELA. A biomass CHP system would reduce annual energy costs by \$63,729 per year. *This is direct cost saving from mitigated diesel and propane fuel consumption annually.* GHG emissions were mitigated by over 98% to 10 t  $CO_2$  from 507 t  $CO_2$ . Wind power generation is very competitive with biomass if not for its high initial cost and moderate reliability. Solar has the lowest feasibility due to the long payback period and high initial cost. Biomass based CHP can achieve savings of about 50% with 4.8% demand reduction. Also, with the existing diesel generators coming towards the end of their operating life, ELA is in an ideal situation to shift from fossil fuel towards a renewable fuel. As well, this approach would be relevant to other off-grid communities with good biomass resources.

### References

Ah-You. K, Leng. G. (1999). Renewable energy in Canada's remote communities. *CANMET Energy Diversification Research Lab Publication*.

Weis, T. M., Ilinca, A., & Pinard, J. (2008). Stakeholders' perspectives on barriers to remote wind–diesel power plants in Canada. *Energy Policy*, *36*(5), 1611-1621.

Huang, B. J., Lin, T. H., Hung, W. C., & Sun, F. S. (2001). Performance evaluation of solar photovoltaic/thermal systems. *Solar Energy*, *70*(5), 443-448.

Bernotat, K., & Sandberg, T. (2004). Biomass fired small-scale CHP in Sweden and the Baltic states: A case study on the potential of clustered dwellings. *Biomass and Bioenergy*, *27*(6), 521-530.

Sims, R. E. H., Rogner, H., & Gregory, K. (2003). Carbon emission and mitigation cost comparisons between fossil fuel, nuclear and renewable energy resources for electricity generation. *Energy Policy*, *31*(13), 1315-1326.

Mustafa Omer, A. (2008). Ground-source heat pumps systems and applications. *Renewable and Sustainable Energy Reviews*, *12*(2), 344-371

Beccali, M., Brunone, S., Cellura, M., & Franzitta, V. (2008). Energy, economic and environmental analysis on RET-hydrogen systems in residential buildings. *Renewable Energy*, *33*(3), 366-382.

Ackermann, T., Garner, K., & Gardiner, A. (1999). Embedded wind generation in weak grids - economic optimization and power quality simulation. *Renewable Energy*, *18*(2), 205-221.

*Research facilities.* Retrieved 5/9/2008, 2008, from <u>http://www.dfo-</u> mpo.gc.ca/regions/central/science/research-recherche/index\_e.htm#Experimental

Planning Study for Experimental Lakes Area filed station for Public Works and Government Services Canada (2003), Calnitsky Associates Architects, 2003.

Phase I/II Environmental Site Assessment, DST Consulting Engineers Inc., 2002

## **CHAPTER 6: CONCLUSION AND RECOMMENDATIONS**

### 6.1 Introduction

Growing concerns about climate change, peak oil and rising energy costs, require that oil dependent communities, particularly off-grid communities, explore renewable energies. The historic price trend of diesel and gasoline shows that increasing demand and political instability together push the oil prices ever higher. Power generation is one of the largest contributors to GHG emissions that fuels global climate change particularly for smallscale diesel-generators (50 kW to 100 kW) that are only 25% to 35% efficient. Since costs for fuel in the remote off-grid communities, with diesel generation and freight costs, are three times more expensive than fuel prices elsewhere in Canada, due to transportation costs, renewable energy technologies (RETs) may make more economic sense in remote off-grid communities. Currently at ELA, diesel generates electrical power at the high energy cost of \$0.230/kW, resulting in GHG emissions of ~280 tons CO<sub>2</sub> annually. In addition, propane heats most building at an additional cost. ELA is considering reducing its emissions and costs through demand reduction and using RETs. This thesis looked at the feasibility of sustainable, reliable energy supply in off-grid communities by conducting a life-cycle analysis of northern Ontario's Experimental Lakes Area (ELA). Demand reduction measures including energy saving measures, retrofitting, and downsizing of equipment have been analyzed. The thesis also iterates that due to the expensive nature of RETs they are always require that demand be reduced to be considered to reduce the peak load, therefore decreasing the initial costs, as well as on-going fuel costs.

### 6.2 Summary

A feasibility study of energy efficiency and renewable energy technology was conducted at a small off-grid community in Ontario. The objectives of the study were met, namely: 1) To understand the existing energy set up at ELA by creating an energy profile of ELA and to develop a base case scenario for the year 2006/2007; 2) To identify and recommend energy saving opportunities using demand reduction approach and to estimate the resulting annual savings and; 3) To perform a renewable energy analysis for a suitable site specific renewable energy resources and to identify the most feasible renewable technology and to integrate it with identified energy efficiency measures. This research provided an assessment of the existing energy management system at ELA.

## 6.3 Conclusion

This thesis shows that improvements in energy efficiency can be made through low cost and no cost retrofitting. Renewable energy for off-grid communities plays a vital role in future as it is not feasible to connect all the off-grid communities to the grid or to displace the communities. At off-grid facilities most of GHG emissions and energy cost are from electricity production followed by heating and transportation. Of the three technologies analyzed, biomass at almost 50% of existing generation cost and neutral GHG emissions is found to be more economically and environmentally feasible than wind and solar for ELA in the long term. Biomass CHP would bring significant benefits in term of emission and environmental risk reduction as well as mitigate fossil fuel consumption.

This thesis also shows that demand reduction and renewable energy technology can be used effectively to dramatically improve the energy situation at ELA resulting in lower energy cost and cleaner energy production. With the existing diesel generators coming towards the end of their operating life, ELA is in an ideal situation to take advantage of the continuously expanding biomass energy market to make the shift from fossil fuel

towards a renewable fuel and mitigate fossil fuel consumption. Based on both economics and environmental considerations, a biomass based energy system would be more optimal than wind and solar power systems for ELA. However, due to large variations in engineering, development and other miscellaneous costs there is a need for further cost analysis to consider on-site seasonal load variability, prospective biomass technologies and a reliable biomass fuel supplier or labor for producing biomass.

Table 6.1 lists a few examples of Biomass CHP system that have been operating successfully in North America and Europe are listed below.

	Output	Overall	Fuel Type	Fuel	Investment	Status
	( <b>kW</b> )	Efficiency		Used/Hr	CAD\$ /	
					kWh	
CPC	25	70 - 80%	Wood	22 dry	4700 -	Fully
Biomax			chips,	kg/hr	7300	commercial
25, US			pellets,			
			shells, etc			
Tervola,	470	81.5%	Sawmill	unknown	3480	Fully
Entimos			wood			operational
Oy,			residues			(2002)
Finland			(bark, saw			
			dust, etc)			
Xylowatt,	300	75%	Wood	300-600	unknown	Fully
xW 300,			residue,	kg/hr		commercial
Belgium			agricultural			
			waste			

 Table 6.1 Comparison of Small-Scale Biomass CHP Systems

Source: Potential for Small-Scale, Community Based Biomass Energy Projects in Nova Scotia Dale Prest, Jamie Simpson, October 2009.

The examples provided in Table 6.1 demonstrate that decentralized small scale biomass benefits are already being realized in parts of North America and internationally. While some of the technologies are just entering commercial production, they represent what is possible in the very near future.
#### 6.4 Recommendations for Diesel Off-Grid Community

This research provides an assessment of the existing energy management system at ELA a small scale off-grid community. Analysis based on key factors like energy consumption, fuel cost, retrofitting and payback period shows that the current load at ELA can be reduced by 4.8% by low-cost and no-cost retrofitting. The general conclusions based on the study objectives are detailed below:

#### 6.4.1 Current System

The energy management system at ELA as currently practiced is unsustainable in the long run. The current system lacks energy monitoring system and has no consistent energy efficiency method to achieve load reduction. The study shows that when it comes to energy efficiency small changes make a big difference in off-grid communities.

#### 6.4.2 Proposed Energy Management System

Figure 6.1 compares the existing energy model at ELA to the recommended model. The chief difference between the models is the way they approach about the energy management in a small off-grid community where small changes can add up to make a considerable saving in energy cost and maintenance.



Figure 6.1 Comparison of Existing Energy Model and the Recommended Energy Model

The Figure 6.2 shows that a little cost and energy management focus at the beginning of the energy plan (RET fuel sources and energy efficiency measures) ends up with low energy costs and environmental costs. The research shows that the right-side up pyramid to be much more stable and sustainable design that rests on a sustainable energy management foundation than unsustainable energy management. The recommended model is aptly suitable for small off-gird communities that are remote and difficult to access thus making this model scalable to other larger off-gird diesel communities in Canada. Figure 6.2 lays out a three step template resulting from this study that can be

replicated for other off-grid communities in Canada that depend exclusively on fossil fuels for power and heat.



#### Figure 6.2 Layout of Sustainable Energy Plan for Small Off-Grid Diesel Communities

#### **6.5 General Recommendations**

Based on the study the following recommendations are drawn keeping in mind the overall energy management for small off-grid communities:

**Energy Monitoring and Targeting:** *"You cannot manage what you can't measure"*. By employing energy monitoring and targeting technique communities can identify where energy savings are and see results of energy management projects. This energy efficiency technique helps in:

- identifying and explain increase or decrease in energy use
- obtain weekly or monthly energy consumption trends
- determine future energy use
- identify specific areas where energy is wasted
- develop performance targets for energy management programs

**Identify Small Changes:** *"Small changes add up"*. Having shown that small changes in energy efficiency add up, the research recommends identifying small and miscellaneous loads that are usually over looked and the focus moves on to a more efficient Heating Ventilation and Air Conditioning (HVAC) units, or other big equipment.

**Set Targets and Goals:** Establish an energy consumption base line and set targets to improve the base line. This is a continuous way to improve the base line by setting future targets for better overall energy performance. Try to keep the consumption below or equal to the historical average is good starting point.

**Community Involvement/Education:** Occupant education and involvement is an important factor in achieving the goals set in the energy program. Let the occupants know the energy savings resulted from their actions.

The recommendations made in this thesis are basic and easily applicable in the off-gird diesel community. Emphasis should be given to a detailed evaluation of the current energy management system and not to ignore small and continuous loads, which can make a considerable impact in the overall energy strategy.

#### <u>Appendix – I</u>

#### **Energy Efficiency Walk Through Survey**

Date: \_\_\_\_\_

The first step is to record all energy consumption from utility bills for the last 12-months. Then walk through the facility and identify all the equipment and processes that use or cause the use of energy. Note size of the equipment, operating hours and temperatures, condition of insulation and weather-stripping, gaps around doors and windows etc.

#### General Information:

(Please circle units used where the second s	nere applicable)
Facility Name:	
Mailing Address:	
Town:	Postal Code:
Name of Facility Operator:	
Title:	_
Phone Number:	_Fax Number:
Name of person completing this form:	
Title:	_
Phone Number:	
Brief Description of Function or Use of Facility:	
Ener Description of Function of Ose of Facility.	
Total Floor area of Facility (sq. m. /sq. ft.):	

#### **Bulk Fuel Worksheet**

Complete one form for each bulk fuel (propane, oil, coal, wood, etc.) used in your facility. The completed form is necessary, as part of the information needed to establish your energy usage and GHG baselines. This information will also provide you with a much better understanding of what your actual energy costs are.

#### Facility Name:

 Fuel Company Name:

Fuel Type: \_\_\_\_\_ Fuel Delivery Units (litres, tonnes cords etc.): \_\_\_\_\_

Account Number: \_\_\_\_\_ Fuel cost / Unit: \_\_\_\_\_

Fuel Use (Entire Facility, Area, Equipment, Etc.):

Year: \_\_\_\_\_ No. Of Months: \_\_\_\_\_ First Month: \_\_\_\_\_

Provincial Tax (%): \_\_\_\_\_\_GST (%): \_\_\_\_\_City Tax (%): \_\_\_\_\_

#### Fuel Type: \_\_\_\_\_

Month/Year	Monthly Fuel	Total
Fuel Delivered	Consumption	Cost
	Units	\$
Total		

#### **Lighting**

Facility:

Location of Lights:

\_

.

Existing lights and controls:

	Type I	Type 2	Type 3	Type4
Type of fixtures (see legend):				
Number of fixtures:				
Number of lamps per fixture:				
If fluorescent indicate length of lamps (2 ft, 3ft, 4ft, 8ft):				
Watts per fixture: (Include ballast wattage if known)				
Fixture height from work surface(ft/m)				
Foot-candle level (if known) – measured at work surface - foot candles				
Present operation of lights - hours/day				
Present operation of lights - days/week				
Present operation of lights – weeks/year				
Present operation of lights - hours/day				
Present operation of lights - days/week				
Present operation of lights - weeks/year				
Present light levels: Bright	Adequat	te	Dim	
Reflectance of walls and ceilings: Good	Averag	e	Poor	
Can lights be switched on and off as desired? Yes No Comment:				
Can lower wattage lamps be installed? Yes <u>No</u> Comment:				
Can existing lamps/fixtures be retrofitted? YesNo Comment:				
Is there an automatic timer? Yes No Is it set properly? Yes No				
Is there an occupancy sensor? YesNoNo	If <u>No</u> , can a	an occupancy s	sensor be installe	ed? Yes

#### Lighting Legend

A. – Incandescent; B. - Fluorescent T-12; C. - Fluorescent T-12 HO (High Output); D. - Compact
Fluorescent; E. - Mercury Vapour; F. - Fluorescent T-12 VHO (VH Output); G. - High Pressure Sodium; H.
- Low Pressure Sodium; I. - Metal Halide (White Light); J.- Fluorescent T-8; K.- Quartz Halogen; L.- Exit
lamp – incandescent; M. - Exit lamp - compact fluor.; N. - Exit lamp – LED; O. Other-specify

#### **Building Envelope**

Facility:

#### **Direction Wall Faces**

For each wall area of facility (front, sides and back of a building) please use one sheet. Windows (Please circle appropriate Yes or No)

Are stor windows used?	m s	Number of glazings	Description of window type (double hung, slider, casement, etc)	Do windows open?	Window fit (poor, fair, good)	Number of windows
Yes	No			Yes No		
Yes	No			Yes No		
Yes	No			Yes No		
Doors (P	lease cire	cle appropriate `	Yes or No) (Please circle unit	s used)		
Are stor	m	Is door	Description of door	Condition of	Door Fit	Number of

Are stor doors us	m sed?	Is do Insul	or ated?	Description of door type (overhead, insulated metal, wood, etc)	door (warped, cracked)	Door Fit (poor, good)	Number of doors
Yes	No	Yes	No				
Yes	No	Yes	No				
Yes	No	Yes	No				

Number/Location of broken or cracked windows:

Description of door or window repairs or replacements needed (including door closers):

Caulking: \_\_\_\_\_\_ft/metres required

Weather-stripping: \_\_\_\_\_\_ft/meters required

**Inside** (Please circle appropriate Yes or No)

Insulation	Insulated?	Present	Insulation Types
		Thickness	
Location			
Ceiling (Attic)	Yes No		
Walls	Yes No		
Basement/Crawlspace walls	Yes No		
Floor / slab	Yes No		

Location of drafts (use strip of tissue to locate):e.g. doors, windows, elec. outlets, attic hatches cracks etc.

Is attic ventilation installed? Yes \_\_\_\_\_ No \_\_\_\_\_ Comments\_\_\_\_\_\_

#### Heating Ventilating and Air Conditioning (HVAC)

Facility Name:	
Please use another sheet id required	
Controls/Use (Please circle units used)	
Location(s) and description of thermostats:	
Location of setback clock/setback thermostat:	
Cold weather thermostat setting:ºC/ºF. Is temperature setback at night and	l on weekends?
If <u>Yes</u> what are setback times and temperatures for: nighttime	weekend
Is temperature setback automaticor manual?	
Hot weather thermostat setting:°C/°F. Is temperature setup at night and o	on weekends?
If <u>Yes</u> what are setup times and temperatures for: nighttime	weekend
Is temperature setup automaticor manual?	
How many hours a week and weeks per year is the system used?	
Hours & weeks in hot weatherHours & weeks in cold weather	
When is system turned on/off in relation to daily occupancy (i.e., before, after, by	how long)?

#### Fans (Supply, Return, Exhaust, Circulating etc.) (Please circle appropriate Yes or No)

× 11 0 /	, , , 8	/ (	11	1	,	
Function: (supply, return etc)	Area served:	Fai	n operating	hours	Can fans b to reduce of times?	be cycled
		hours / day	days / week	weeks/ year		
					Yes	No
					Yes	No
					Yes	No
					Yes	No
					Yes	No
					Yes	No
					Yes	No

### **Office and Lab Equipment**

(Computers, printers, photocopiers, etc)	
Facility Name:	
Please use more sheets if required	
Office machine:	
Machine type, location	
Wattage (nameplate watts or amps x volts):	
Is it left on overnight?	Over weekends?
Daily hours of operation: Office machine:	_Hours per day it could be turned off:
Machine type, location	
Wattage (nameplate watts or amps x volts):	
Is it left on overnight?	Over weekends?
Daily hours of operation: Office machine:	_Hours per day it could be turned off:
Machine type, location	
Wattage (nameplate watts or amps x volts):	
Is it left on overnight?	Over weekends?
Daily hours of operation: Office machine:	_Hours per day it could be turned off:
Machine type, location	
Wattage (nameplate watts or amps x volts):	
Is it left on overnight?	Over weekends?
Daily hours of operation: Office machine:	_Hours per day it could be turned off:
Machine type, location	
Wattage (nameplate watts or amps x volts):	
Is it left on overnight?	Over weekends?
Daily hours of operation: Office machine:	_Hours per day it could be turned off:
Machine type, location	
Wattage (nameplate watts or amps x volts):	
Is it left on overnight?	Over weekends?
Daily hours of operation:	Hours per day it could be turned off:

# **Machines and Equipment**

Facility Name:			
Please use another sheet if r Refrigeration and Freezing	equired (Please circle un	nits used	d)
Type, age, energy used:			
Compressor rating:	hp; age:	_years	Present temperature: °C/°F
Hours per day of use:	Weeks per yea	ar equip	ment is used
Do doors close completely, by	themselves?		_Condition of door seals:
Refrigeration and Freezing	(Please circle un	its use	d)
Type, age, energy used:			
Compressor rating:	hp; age:	_years	Present temperature: °C/°F
Hours per day of use:	Weeks per yea	ar equip	ment is used
Do doors close completely, by	themselves?		_Condition of door seals:
Refrigeration and Freezing	(Please circle un	its use	d)
Type, age, energy used:			
Compressor rating:	hp; age:	_years	Present temperature: °C/°F
Hours per day of use:	Weeks per yea	ar equip	ment is used
Do doors close completely, by	themselves?		_Condition of door seals:
Cooking (Range, oven, grill	, etc) (Please cir	rcle uni	ts used)
Type, age, energy used:			Temperature now used: °C/°F
Is this the lowest possible tem	perature? Yes	_No	_Is equipment turned off when possible?_
Are exhaust hoods installed ov	ver all cooking eq	uipment	t? YesNo
Cooking (Range, oven, grill	, etc) (Please cir	rcle uni	ts used)
Type, age, energy used:			Temperature now used: °C/°F
Is this the lowest possible tem _Are exhaust hoods installed	perature? Yes over all cooking e	No equipme	Is equipment turned off when possible? ent? YesNo
Cooking (Range, oven, grill	, etc) (Please cir	rcle uni	ts used)
Type, age, energy used:			Temperature now used: °C/°F
Is this the lowest possible tem	perature? Yes	_No	_Is equipment turned off when possible?_
Are exhaust hoods installed ov	ver all cooking eq	uipmen	t? YesNo

# **Miscellaneous Equipment**

Facility Name:			
Please use another sheet if requ Washer Dyer (If applicable)	ired		
Type, age, energy used:			
Temperature now used: Hot	Warm	Cold	
Are machines fully and properly le	oaded? Yes	No	
Can lower washing/rinse water	temperatures be	used? Yes	No
Dish Washing (If applicable)			
Type, age, energy used:			
Temperature now used: Hot	Warm	Cold	
Are machines fully and properly le	oaded? Yes	No	
Can lower washing/rinse water	temperatures be	used? Yes	No
Dish Washing (If applicable)			
Type, age, energy used:			
Temperature now used: Hot	Warm	Cold	
Are machines fully and properly le	oaded? Yes	No	
Can lower washing/rinse water	temperatures be	used? Yes	No
	N	otos	

Notes



# <u>Appendix – II</u>

# **<u>RETScreen 4.0: Wind Analysis</u>**

### **Project Information:**

3 9 · · · ) ·	Wind Analysis- ELA. [Compatibility Mode] - Microsoft Excel	(chiế)
Home Insert Page Layout Formulas	Data Review View Add-Ins	9 -
D14 🔹 🌾 125kW		
	Clean Energy Project Analysis Software	
Project information	See project database	
Deviant same	12540	
Project location	E.A.	
Prepared for	OFO	
Prepared by Bha	u and Dr. Stirley Thompson	
Project type	Ruer	
Technology	Wind turbine	
Grid type	Off-grid	
Analysis type	Method 2	
Heating value reference	gher heating value (HHV)	
Show settings 🛛 🖗		
Language - Langue	English - Anglais	
User marual	chipish - Anges	
Currency	5	
Units	Metric ands	
Site reference conditions	lect clinate data location	
2014 State 201		
Climate data location	Kenora Airport	
Show data 🛛		
Climate data	Project	
tude 71 49.8	49.8	
gtude E -34.4	-944	
Catality chergy model of the malyss of	Disperir miarpa // Citatival miarpa // Non miarpa // 1006 / C	

# Site Conditions:

	a) 📙 🌒 -	(□ - ) ₹	i han	und they	-	and they	Wind Analy	/sis- ELA [Com	patibility Mode]	- Microsoft Ex	cel			_ 0	X	
U	Home	Insert P	age Layout	Formulas	Data Revi	iew View	Add-Ins							0 -	. 🔊 )	c
	D14	<b>•</b> ()	fx	125kW											;	*
			<b>9</b>												E	-
	Site r	<b>reference c</b> Climate c	conditions	<u>. Se</u>	<i>lect climate data lo</i> Kenora Airpor	o <u>cation</u> t	]	-							Í	
			Show data	¥												
	Latitude Longitude Elevation Heating design Cooling design Earth temperatu	temperature temperature ure amplitude	Unit °N °E ©C °C °C	Climate data location 49.8 -94.4 411 -29.8 27.3 25.3	Project location <u>49.8</u> -94.4 411											
	Month January February March April May June Juny August September October November December Annual Measured at		m	Air temperature *C -17.8 -14.1 -6.2 3.3 11.2 16.4 19.6 17.9 11.8 5.5 -4.5 -14.4 2.5	Relative humidity % 72.5% 70.0% 60.5% 60.0% 66.0% 66.0% 67.5% 73.0% 73.0% 73.0% 78.0% 69.5%	Daily solar radiation - horizontal KWh/m?/d 1.48 2.51 4.12 5.35 5.96 6.01 5.99 5.01 3.43 2.19 1.36 1.13 3.72	Atmospheric pressure 96.5 96.7 96.6 96.5 96.5 96.5 96.5 96.5 96.5 96.5	Wind speed           m/s           3.9           4.2           4.4           4.2           3.9           4.2           4.2           3.9           4.2           4.2           3.9           4.2           3.9           4.2           3.9           4.2           3.9           4.2           3.9           4.1           10.0	Earth temperature °C -17.2 -14.0 -7.9 2.3 11.4 16.9 19.2 17.7 11.4 3.4 -6.7 -14.6 1.9 0.0	Heating degree-days °C-d 1,110 899 750 441 211 48 0 0 3 186 388 675 1,004 5,715	Cooling           degree-days           °C-d           0           0           0           0           0           0           0           0           0           0           0           0           0           0           0           298           245           54           0           0           0           0           826					
н	Start	Energy Mo	reeep del Cos	<u>Com</u> t Analysis	<i>aplete Energy Mod</i> Emission Analysis	l <u>el sheet</u> Financial A	Analysis / Risk	: Analysis / Ti	pols / 🛱			4			•	
Re	ady												₩ 🛛 🛄 80% 🤆	)	-(	)

# **Power Project:**

💽 🚽 🕈 C 🗸 C	Wind Analysis- ELA [Compatibility Mode] - Microsoft Excel	- 0
Home Insert Page Layout	Formulas Data Review View Add-Ins	🗋 _ 5 X
Link_M_Start • (* fx		\$
RETScreen Energy Model - Power project		
Power project		1
Base case power system		
Grid type	Off-grid	
Technology	Redprocating engine	
Fueltype	Diesel (#2 oi) - L	
Fueirate	SL 0.000	
Lapacity Heat rate	112.00	
Annual D&V cost	S 500	
Electricity rate - base case	SN/h 0.230	
Total electricity cost	\$ 97,567	
Load characteristics		_
	R liethod 1 <sup>C</sup> liethod 2	
	Init Data sate Departed sate	
Electricity - daily - DC	Wh base case Proposed case	
Electricity - daily - AC	Wh 1.159.820 800.000	
Intermittent resource-load correlation	Negative	
Percent of month used		
	Base case Proposed case Energy saved	
Electricity - annual - DC	W/h 0.000 0.000	
Electricity - annual - AC	IWh 423.334 292.000 31%	
Peak load - annual	kW 100.00	Ĩ
Proposed case power system		
Inverter		
Capacity	kiv 100.0 Peak load - annual - AC	
Efficiency	7 <u>307</u>	
Miscelateous (cases	78	
Battery		
Days of autonomy	d 5.0 See mans	
Votage	V 460 A Bankelon / Dented Analysis / Det Bankelon / Tech (00)	1
Start Energy Model / Cos	st Analysis 🦯 Emission Analysis 🦯 Hitanciai Analysis 🦯 Kisk Analysis 🖉 1006 / 🎾	
Ready		

## **Proposed Case:**

	Wind Analysis- ELA [Compatibility Mode] - Microsoft Excel	
Home Insert Page Layout Formulas	Data Review View Add-Ins	0 _ o x
Link M Start 🔹 🎜		*
Proposed case power system		
		1
Inverter Capacity KW Efficiency % Niscelaneous losses %	100.0         Peak load - annual - AC           90%         0%	
Battery Days of autonomy d Votage V Efficiency % Naximum depth of discharge % Charge controller efficiency % Temperature control method Battery temperature   Conarity Average battery temperature derating % Charge Ab	5.0         Set maps           48.0         85%           60%         95%           55%         Constant           24.0         0.4%           12.000         154.906	
Capacity All Battery KWh Technology	576 Wind turbine	
Resource assessment	IF ,Show data See maps Electricity	Ē
Resource method January February March April May June June June July August September October November December December	Wind speed         Kenora Airport         load           mis         m/s         MWh           3.9         3.9         8.13           3.9         3.9         7.34           4.2         4.2         10.03           4.4         4.4         10.93           4.2         4.2         9.70           3.9         3.9         8.13           3.9         3.9         8.13           3.9         3.9         8.13           3.9         3.9         8.13           4.2         4.2         9.70           4.4         4.4         11.29           4.2         4.2         9.70           3.9         3.9         8.13           4.1         4.1         111.23	
H + + H Start Energy Model Cost Analysis	Emission Analysis / Financial Analysis / Risk Analysis / Tools / 🕲	
Ready		

#### Wind Data:



### **Cost Analysis:**

A 9 + (N + ) +     A  A     A	and the second	1000	Wind An	alysis- ELA [Compat	ibility Mode] - Microsoft I	Excel			x
Home Insert Page Layout	Formulas Data	Review V	iew Add-Ins					0 -	• )
Link_CA_Start 🝷 💿 🎜									3
RETScreen Cost Analysis - Power project									
Settings									
Method 1 Method 2	<ul> <li>Notes/Range</li> <li>Second current</li> </ul>	nev.							
	Cost allocation	1							
Initial costs (credits)	Unit	Quantity	Unit cost	Amount F	elative costs	% Am	ount		
Feasibility study Feasibility study	cost	1 5	10,000 S	10.000		\$			
Sub-total:			\$	10,000	2.3%	0% \$	•		
Development	cost	1 \$	- 5			5			
Sub-Istat Engineering			\$		0.0%	0% \$	•		
Engineering	cost	1 \$	5,000 S	5,000		\$	<u> </u>		
Sub-total: Power system			s	5,000	1.2%	0% \$			
Base load - Wind turbine	KW NW	125.00 \$	3,300 \$	412,500		\$			
Road construction	km	5	- 5			ŝ			
Transmission line Substation	project	\$	- 5	-		s			
Energy efficiency measures	project	5	- 5			5	•		
User-defined	COSI	3	5			5	:		
Sub-total: Balance of system & miscellaneous			\$	412,500	95.5%	0% \$	•		
Spare parts	%	10.0% \$	20,000 \$	2,000		s	•		
Training & commissioning	project p-d	3	5			5			
User-defined Contingencies	cost %	5	429.500 5	-		5			
Interest during construction	7.00%	2 month(s) S	429,500 5	2,505	1.44	S			
Total initial costs			5	4,505	100.0%	0% \$			
Annual costs (credits)	Unit	Quantity	Unitcost	Amount		% Am	ount		
O&M	amiant								
User-defined	cost		s			s			-
I FILL STATE FRAME MODEL OF A	and the second sec	and take / Plan	a shi kashala 🗸 🕷	fals Americals					1. C
Ready	nalysis / Emission A	knalysis 🏑 Fina	ncial Analysis 🏑 🖡	isk Analysis 📝 Tool	·/Q/				► ] 
Ready	nalysis / Emission A	inalysis / Fina	nciel Analysis 🏑 🖡	isk Analysis 🟑 Tool	i/Q/				)    -  €
Ready	nallysis <u>/</u> Emission A	inalysis 🧹 Fina	nclal Analysis 🧹 🖡 Wind A	isk Analysis / Tool	tibility Mode) - Microsoft	Excel			> 1 
Ready Home Inset Page Layout	Formulas Deta	inalysis / Fina Review	ncial Analysis / F Wind A View Add-Ins	isk Analysis _/ Tool Inalysis- ELA [Comp	tibility Mode) - Microsoft	Excel			> 1 
Ready Home Inset Page Layout Link_CA_Start •	nalysis, — Emission A Formulas Data	inalysis / Fina	ncial Analysis 🦯 8 Wind A View Add-Ins	isk Analysis —/ Tool	s 🤇 😒	Excel			) () () () () () () () () () () () () ()
Ready Home Insert Page Layout Link_CA_Start • A Read construction Transmission line	Formulas Data	nalysis / Fina	Ncial Analysis / 8 Wind A View Add-Ins S - S -	isk Analysis // Tool Inalysis: ELA (Comp S - S -	tibility Mode] - Microsoft	Excel	:		
Ready Home Insert Page Layout Link_CA_Start • fe Read construction Transmission line Substation	Formulas Data	Review	Wind A Wind A View Add-Ins S - S -	isk Analysis - / Tool Inalysis= ELA [Comp. S - S - S - S -	nibility Mode) - Microsoft	Excel S S S	:		
Ready Home Insert Page Layout Link_CA_Start • free Read construction Transmission line Substation Energy efficiency measures User-defined	Formulas Data	Review	Wind Analysis / 5 Wind A View Add-Ins S - S - S - S - S -	isk Analysis - / Tool Inalysis- ELA (Comp S - S - S - S - S - S - S -	nibility Mode] - Microsoft	Eccel S S S S S S S			
Ready Home Inset Page Layout Link_CA_Start Read construction Transmission line Substaten Energy efficiency measures User-defined Sub-total	Formulas Data	Review	Wind Analysis / 5 Wind A View Add-Ins S - S - S - S - S - S -	sk Analysis / Tool enalysis- ELA [Comp s - s - s - s - s - s - s - s - s - s -	tibility Mode) - Microsoft	Excel 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5			
Ready  Ready  Home Inset Page Layout  Link_CA_Start  Read construction  Transmission line  Substation  Energy efficiency measures  User-defined  Sub-totat  Balance of system & miscellaneous  Scener parts	Formulas Data	Review	Wind Analysis         / 5           Wind A         /           View         Add-Ins           S         -           S         -           S         -           S         -           S         -           S         -           S         -           S         -           S         -           S         -           S         -	Isk Analysis / Tool unalysis: ELA [Comp. s - s - s - s - s - s - s - s - s - s -	ntibility Mode] - Microsoft	Excel			F ] () 5 X 8
Ready  Ready  Home Insert Page Layout  Link_CA_Start  Read construction  Transmission line Substation Energy efficiency measures User-defined  Sub-total Balance of system & miscellaneous Spare parts Transportation Transportation	Formulas Data	Review	Wind Analysis / 5 Wind A View Add-Ins 5 - 5 - 5 - 5 - 5 - 5 - 5 - 5 - 5 - 5 -	Isk Analysis / Tool unalysis: ELA [Comp. s - s - s - s - s - s - s - s - s - s -	s / S	Excel			
Ready  Ready  Home Insert Page Layout  Link_CA_Start  Read construction  Transmission line Substation Energy efficiency measures  User-defined  Sub-totat Balance of system & miscellaneous Spare parts Transportation Training & commissioning  User-defined	Formulas Data	Review	Wind J           Wind J           S <td< td=""><td>Isk Analysis / Tool unalysis ELA [Comp. S - S - S - S - S - S - S - S - S - S -</td><td>nibility Mode) = Microsoft</td><td>Excel</td><td></td><td></td><td>× 1 • • • • • • • • • • • • • • • • • • •</td></td<>	Isk Analysis / Tool unalysis ELA [Comp. S - S - S - S - S - S - S - S - S - S -	nibility Mode) = Microsoft	Excel			× 1 • • • • • • • • • • • • • • • • • • •
Ready  Ready  Home Insert Page Layout  Link_CA_Start  Read construction  Transmission line  Substation  Energy efficiency measures  User-defined  Sub-total:  Balance of system & miscellaneous  Spare parts  Transing & commission ling  User-defined  Contingencies  Ware-defined  Contingencies  Ware-defined	Formulas Data	Review Review 10.0% 1 3 2 month(s)	Wind A           Wind A           View         Add-Ins           \$         -           \$         -           \$         -           \$         -           \$         -           \$         20,000           \$         429,500           \$         429,500	Isk Analysis - Tool unalysis- ELA [Comp. S - S - S - S - S - S - S - S - S - S -	nibility Mode] - Microsoft	Eccel			
Ready  Ready  Home Insert Page Layout  Link_CA_Start  Read construction  Transmission line  Substation  Energy efficiency measures User-defined  Sub-totat  Balance of system & miscellaneous  Spare parts  Training & commissioning  User-defined  Contingencies  Interest during construction  Sub-totat  Total initial costs	Formulas Data	Review Review 10.0% 1 3 2 month(s)	Wind A           Wind A           View         Add-Ins           \$         -           \$         -           \$         -           \$         -           \$         -           \$         -           \$         -           \$         -           \$         -           \$         -           \$         -           \$         -           \$         -           \$         -           \$         -           \$         -           \$         -           \$         -	sk Analysis _/ Tool knalysis ELA [Comp. s - s - s - s - s - s - s - s -	nibility Mode] - Microsoft	Eccel			
Ready  Ready  Home Inset Page Layout  Link_CA_Start  Read construction  Transmission line  Sub-totat  Balance of system & miscellaneous  Spare parts  Transportation  Transing & commissioning  User-defined  Contingencies  Interest during construction  Sub-totat  Totat initial costs  Construction	Formulas Data	Review Review 10.0% 1 3 2 month(s)	Wind A           Wind A           View         Add-Ins           S         -           S         -           S         -           S         -           S         -           S         -           S         -           S         -           S         -           S         -           S         -           S         -           S         -           S         429,500           S         429,500	sk Analysis , / Tool enalysis - ELA [Comp. 5 - 5 5 - 5	s, Calify Mode] - Microsoft	Excel			× I • • • • •
Ready  Ready  Home Insert Page Layout  Link_CA_Start  Read construction  Transmission line  Sub-totat  Balance of system & miscellaneous  Spare parts  Transportation  Training & Cormissioning  User-defined  Contingencies  kiferest during construction  Sub-totat  Total initial costs  Annual Costs (creates)  O&M	Formulas Data	Review Review 10.0% 1 3 2 month(s) Quantity	Wind J           Wind J           View         Add-Ins           S         -           S         -           S         -           S         -           S         -           S         -           S         -           S         -           S         -           S         -           S         -           S         -           S         -           S         -           S         -           S         -           Built cost         -	Isk Analysis - Tool Inalysis ELA [Comp. S - S - S - S - S - S - S - S -	1.0%	Excel	- - - - - - - - - - - - - - - - - - -		
Ready  Ready  Nome Insert Page Layout  Unk_CA_Start  Read construction  Transmission line Substation Energy efficiency measures User-defined  Sub-total: Balance of system & miscellaneous Spare parts Transportation Training & Commissioning User-defined  Contingencies Interest during construction Sub-total: Total initial costs  Annual costs (credits)  O&M Parts & labour User-defined	Formulas Data	Review Review 10.0% 1 3 2 month(s) Quantity	Wind J           Wind J           Wind J           S	Isk Analysis _/ Tool Inalysis ELA [Comp. S - S - S - S - S - S - S - S -	ntibility Mode] = Microsoft	Excel	- - - - - - - - - - - - - - - - - - -		
Ready  Ready  Ready  Home Insert Page Layout  Link_CA_Start  Read construction  Transmission line  Substation  Energy efficiency measures  User-defined  Sub-totat  Balance of system & miscellaneous  Spare parts  Training & commissioning  User-defined  Contingencies  sub-totat  Total initial costs  Annual costs  Annual costs  Annual costs  Certingencies  Sub-totat	Formulas Data	Review Review 10.0% 1 2 month(s) Quantity	Wind A           Wind A           View         Add-Ins           \$         -           \$         -           \$         -           \$         -           \$         20,000           \$         429,500           \$         429,500           \$         429,500           \$         429,500           \$         429,500	Isk Analysis - / Tool Isk Analysis - ELA [Comp. S - S - S - S - S - S - S - S -	nibility Mode] - Microsoft	Excel			
Ready  Ready  Ready  Home Insert Pape Layout  Link_CA_Start  Read construction  Transmission line  Substation  Energy efficiency measures User-defined  Sub-totat  Balance of system & miscellaneous  Spare parts  Training & commissioning User-defined  Contingencies Interest during construction Sub-totat  Total initial costs  Annuel costs  Annuel costs  Annuel costs  Contingencies Sub-totat  Fuel cost - proposed case Duser-defined	Formulas Data	Review Review 10.0% 1 2 month(s) Quantity 58.877	Wind A           Wind A           View         Add-Ins           \$         -           \$         -           \$         -           \$         -           \$         20,000           \$         429,500           \$         429,500           \$         429,500           \$         429,500           \$         429,500	Isk Analysis _/ Tool Isalysis ELA [Comp. S - S - S - S - S - S - S - S -	nibility Mode] - Microsoft	Eccel	- - - - - - - - - - - - - - - - - - -		
Ready  Ready  Ready  Home Inset Page Layout  Link_CA_Start  Read construction  Transmission line  Substaten  Energy efficiency measures  User-defined  Sub-totat  Balance of system & miscellaneous  Spare parts  Transportation  Training & commissioning  User-defined  Contingencies  Sub-totat  Total initial costs  Annual costs (credits)  O&M  Parts & labour  User-defined  Contingencies  Sub-totat  Fuel cost - proposed case  Dised (62 or)  Sub-totat	Formulas Data	Review	Wind A           Wind A           View         Add-Ins           \$         -           \$         -           \$         -           \$         -           \$         20,000           \$         429,500           \$         429,500           \$         429,500           \$         -           \$         -           \$         0,800	Isk Analysis _/ Tool knalysis - ELA [Comp. \$ - \$ - \$ - \$ - \$ - \$ - \$ - \$ -	s / 2	Eccel	- - - - - - - - - - - - - - - - - - -		
Ready  Ready  Home Inset Page Layout  Link_CA_Start  Read construction  Transmission line  Sub-totat  Balance of system & miscellaneous  Spare parts  Transportation  Transing & commissioning  User-defined  Contingencies  Interest during construction  Sub-totat  Totat initial costs  Annual costs (credits)  O&M  Parts & isboar  User-defined  Contingencies  Sub-totat  Fuel cost - proposed case  Dised (IC on)  Sub-totat  Annual (2 surges	Formulas Data Interproject State Sta	Review Review 10.0% 1 3 2 month(s) Quantity 58,877 Quantity	Wind A           Wind A           S           O           B000	Isk Analysis , / Tool knalysis - ELA [Comp. 5 - 5 5 - 5 5 - 5 5 - 5 5 - 5 5 - 7 5 - 7	s, 2 tibility Mode] - Microsoft	Excel			
Ready  Ready  Ready  Home Inset Page Layout  Link_CA_Start  Read construction  Transmission line  Sub-totat  Balance of system & miscellaneous  Spare parts  Transportation  Training & Cormissioning  User-defined  Contingencies  kiterest during construction  Sub-totat  Total initial costs  Annual costs  Annual costs  Annual cost proposed case Diesel (K2 oil)  Sub-totat  Fuel cost - proposed case Diesel (K2 oil)  Sub-totat	Formulas Data km km project project project p-d coat % 7.00% Unit project coat % 1 km km km km km km km km km km km km km	Review  Review  Review  10.0%  1  2.month(s)  Quantity  58,877  Quantity  121.334	Wind J           Wind J           View         Add-Ins           \$         -           \$         -           \$         -           \$         -           \$         -           \$         -           \$         -           \$         -           \$         -           \$         -           \$         -           \$         -           \$         -           \$         -           \$         -           \$         0.000           Unit cost         -           \$         0.000	Isk Analysis _< Tool unalysis ELA [Comp. \$ - \$ - \$ - \$ - \$ - \$ - \$ - \$ -	ntibility Mode] = Microsoft	Excel			
Ready  Ready  Nome Inset Page Layout  Unk_CA_Start  Read construction  Transmission line Substation Energy efficiency measures User-defined  Sub-total  Balance of system & miscellaneous Spare parts Transportation Training & Commissioning User-defined  Contingencies Interest during construction Sub-total  Total initial costs  Annual costs (credics)  O&M  Parts & labour User-defined  Contingencies Sub-total  Fuel cost - proposed case Dised (#2 oi) Sub-total  Fuel cost - base case Dised (#2 oi) Sub-total	Formulas Data	Review Review Review 10.0% 1 2.month(s) Cleandby 58,877 Quandby 121,334	Wind /           Wind /           View         Add-Ins           \$         -           \$         -           \$         -           \$         -           \$         -           \$         -           \$         -           \$         -           \$         -           \$         -           \$         -           \$         -           \$         -           \$         0.800           Bait cost         -	Isk Analysis _/ Tool Isalysis ELA [Comp. S - S - S - S - S - S - S - S -	nibility Mode] - Microsoft	Excel			
Ready  Ready  Ready  Ready  Ready  Ready  Ready  Ready  Read construction  Transmission line  Substation  Energy efficiency measures  User-defined  Sub-totat  Balance of system & miscellaneous  Spare parts  Training & commissioning  User-defined  Contingencies  Interest during construction  Sub-totat  Read Construction  Sub-totat  Read Construction  Sub-totat  Read Construction  Sub-totat  Read Contingencies  Interest during construction  Sub-totat  Read Contingencies  Interest during construction  Sub-totat  Read Contingencies  Interest during construction  Sub-totat  Read Contingencies  Read Contingencies  Sub-totat  Read Contingencies  Read Contingencies  Sub-totat  Read Contingencies  Sub-totat  Read Contingencies  Read Cont	Formules Data Inner Formules Data Inner Formules Data Inner Formules Data Inner Formules Form	Review	Wind A           Wind A           View         Add-Ins           \$         -           \$         -           \$         -           \$         -           \$         20,000           \$         429,500           \$         429,500           \$         429,500           \$         0,000           \$         0,000           \$         0,000           Unit cost         5           \$         0,804           Unit cost         1	Isk Analysis _< Tool traalysis ELA [Comp. \$ - \$ - \$ - \$ - \$ - \$ - \$ - \$ -	nibility Mode] - Microsoft	Eccel			
Ready  Ready  Ready  Ready  Ready  Ready  Ready  Ready  Read construction  Transmission line  Sub-totat  Balance of system & miscellaneous  Spare parts  Transportation  Training & commissioning  User-defined  Contingencies  Sub-totat  Read cost-sub-totat  Annual costs (credits)  O&M  Parts & labour  User-defined  Contingencies  Sub-totat  Fuel cost - proposed case  Dised (#2 oi)  Sub-totat  Periodic costs (credits)  User-defined  Contingencies  Sub-totat  Fuel cost - base case  Dised (#2 oi)  Sub-totat  Periodic costs (credits)  User-defined  Contingencies  Sub-totat  Fuel cost - base case  Dised (#2 oi)  Sub-totat  Periodic costs (credits)  User-defined  Contingencies  Contingencies  Sub-totat  Periodic costs (credits)  Contingencies  Periodic costs (credits)  Contingencies  Contingencies  Contingencies  Contingencies  Contingencies  Sub-totat  Periodic costs (credits)  Contingencies  Cont	Formulas Data Inner Formulas Inner	Review Review Review 10.0% 1 3 2 month(s) Quantity 58,877 Quantity 121.334 Year	Nime         Wind /           Wind /         Add-Ins           \$         -           \$         -           \$         -           \$         -           \$         -           \$         -           \$         20,000           \$         429,500           \$         429,500           \$         0,8000           \$         0,8000           Unit cost         \$           \$         0,800	Isk Analysis _/ Tool knalysis = ELA [Comp. \$ - \$ - \$ - \$ - \$ - \$ - \$ - \$ -	s, 2 nibility Mode] - Microsoft	Eccel			
Ready  Ready  Ready  Home Inset Page Layout  Link_CA_Start  Read construction  Transmission line  Sub-totat  Balance of system & miscellaneous  Spare parts  Transportation  Training & commissioning  User-defined  Contingencies  Veread Aring construction  Sub-totat  Totat initial costs  Annual costs (credits)  O&M  Parts & labour  User-defined  Contingencies  Sub-totat  Fuel cost - proposed case Diesel (#2 oi)  Sub-totat  Periodic costs (credits)  User-defined  Liser-defined  Contingencies  Sub-totat  Periodic costs (credits)  User-defined  End of project life	Formulas Data Interproject Formulas Data Interproject Formulas For	Review Review Review 10.0% 1 3 2 month(s) Quantity 58,877 Quantity 121,334 Year	Nind Analysis         /         5           Wind A         Wind A         1           S         -         5         -           S         -         5         -           S         -         -         -         -           S         -         -         -         -           S         -         -         -         -           S         -         -         -         -           S         429,500         -         -         -           S         429,500         -         -         -           S         0.800         -         -         -           S         0.800         -         -         -           S         0.800         -         -         -	Isk Analysis _< Tool knalysis = ELA [Comp. \$ - \$ - \$ - \$ - \$ - \$ - \$ - \$ -	s, 2 nibility Mode] - Microsoft	Excel			
Ready  Ready  Ready  Ready  Ready  Ready  Ready  Ready  Read construction  Transmission line  Sub-totat  Balance of system & miscellaneous  Spare parts  Transportation  Training & Cormissioning User-defined  Contingencies  kiterest during construction  Sub-totat:  Total initial costs  Annual costs  Annual costs  Annual cost proposed case Diesel (#2 oi)  Sub-totat  Periodic costs (credits)  User-defined  Contingencies  Sub-totat  Read (#2 oi)  Sub-totat  Periodic costs (credits)  User-defined  Contingencies  Sub-totat  Read (#2 oi)  Sub-totat  Periodic costs (credits)  User-defined  Contingencies  Ready  R	Formulas Data km km project project p-d coat % project p-d coat % 7.00% Unit project coat % Unit Unit Unit Unit Unit Unit Unit Unit	Review Review Review 10.0% 1 2 month(s) Quantity S8,877 Quantity 121,334 Year to Emission Analys	Wind /           Wind /           S           S           S           S           S           S           S           S           S           S           S           S           S           S           Unit cost           S           Unit cost           S           Unit cost           S           Unit cost           S <td>Isk Analysis _&lt; Tool coalysis ELA [Comp. 5 - 5 - 5 - 5 - 5 - 5 - 5 - 5 -</td> <td>s. 5%</td> <td>Excel</td> <td></td> <td></td> <td></td>	Isk Analysis _< Tool coalysis ELA [Comp. 5 - 5 - 5 - 5 - 5 - 5 - 5 - 5 -	s. 5%	Excel			
Ready  Ready  Ready  Nome Inset Page Layout  Unk_CA_Start  Road construction  Transmission line Substation Energy efficiency measures User-defined  Sub-total: Balance of system & miscellaneous Spare parts Transportation Training & Cormissionling User-defined Contingencies Interest during construction Sub-total: Total initial costs  Annual costs  Annual cost proposed case Diesel (#2 oi) Sub-total: Periodic costs (credits) User-defined Contingencies Cost proposed cose Disad-total: Periodic costs (credits) Cost propo	Formulas Data km project project coat s% 7.00% Unit project coat S% C.00% Unit Unit project coat C. Unit coat C.	Review  Review  Review  10.0%  1  2.month(s)  Quantity  Guantity  121.334  Year  to Emission Anelys	Niew         Add-Ins           S         -           S         -           S         -           S         -           S         -           S         -           S         -           S         -           S         -           S         429,500           S         429,500           S         0,800           Unit cost         -           S         0,800           Unit cost         -           S         0,800           Unit cost         -	Isk Analysis - Tool Isalysis ELA [Comp. S - S - S - S - S - S - S - S -	nibility Mode] - Microsoft	Eccel			

H 4 + H Start / Energy Model Cost Analysis / Emission Analysis / Financial Analysis / Risk Analysis / Tools / 27 Ready

# **GHG Emission Analysis:**

A . (* *)     F     F     F     F     F     F     F     F	1	- mark			Wind Analysi	s- ELA [Compatibili	ty Mode] - Micros	aoft Excel		- 0 -	X
Home Insert P	age Layout	Formulas	Data Review	View	Add-Ins					<u>0</u> - 0	X
Link_GHG2_Start •	f <sub>x</sub>										×9
RETScreen Emission Reduction	Analysis - Po	wer project									•
Emission Analysis											n
S Halland 1											
C Method 2											
C Method 2											
· WEDROG 5											
Base case system GHG summar	r (Baseline)										
						Fuel	GHG emission	GHG			
	Fuel mix					consumption	factor	emission			
Fuel type	5					MWh	tC02/MWh	1002			
Diesel (#2 oil)	100.0%					1,294	0.252	326			
1914	100.078					1,234	0.432	320			
Proposed case system GHG surr	mary (Power	r project)									
	Fual mix					Fuel	GHG emission factor	GHG			=
Fuel type	%					MWh	1CO2/MWh	1002			П
Diesel (#2 oil)	100.0%					628	0.252	158			
Wind	0.0%					1	0.000	0			
Total	100.0%					628	0.252	158			
CMC aminging radiation summa	P.										
ono emissión reductión summa	iy										
		Base case				Gross annual		Net annual			
		GHG	Proposed case			GHG emission	GHG credits	GHG			
		1007	ticop			tC02	s sacion lee	tC02			
Power project		326	158			168		168			
Net annual GHG emission reduc	tion	168	1002	is equivale	ntto 34.2	Cars & light truck	s not used				
		Cons	vete Financial Analy	sis steel							
											ľ
H 4 F H Start / Energy Mo	del 🖉 Cost	Analysis 📜 Em	ission Analysis /	Financial A	nalysis / Risk A	inalysis / Tools /	<b>9</b> /			•	1
N.CODY									6		111

## Financial Analysis:

<b>(</b> ) <b>( (</b> ) <b>( ( ( ) )</b>	These for	of they in	W	ind Analysi	is- ELA [Compatib	lity Mode	e] - Microsoft Exe	cel				_ 0 _>	-
Home Insert P	age Layout	Formulas Da	ata Review View Add	-Ins								0 _ =	χ
Link_FS_Start 🔻 🕤	$f_x$												¥
RETScreen Financial Analysis - F	Power project												
Financial parameters			Project costs and savings/incom	ne summa	rv	Year	v cash flows		ż				Π
General			Initial costs		.,	Year	Pre-tax	After-tax	Cumulative				
Fuel cost escalation rate	%	2.0%	Feasibility study	2.3%	\$ 10,0	00 #	\$	\$	\$				
Inflation rate	%	3.0%				0	-216,003	-216,003	-216,003				
Discount rate	%	7.0%	Engineering	1.2%	\$ 5,0	00 1	11,395	11,395	-204,608				
Project life	yr	20	Power system	95.5%	\$ 412,5	00 2	12,425	12,425	-192,183				
						3	13,475	13,475	-178,708				
Finance						4	14,546	14,546	-164,163				
Incentives and grants	S					5	15,638	15,638	-148,524				
Debt ratio	%	50.0%				6	16,753	16,753	-131,772				1
Debt	ş	216,003	Balance of system & misc.	1.0%	\$ 4,5	05 7	17,889	17,889	-113,883				1
Equity	ş	216,003	l otal initial costs 1	00.0%	\$ 432,0	05 8	59,129	59,129	-54,/54				1
Debt interest rate	%	7.00%				9	60,311	60,311	5,55/				
Debt term Debt permante	yr Shr	40.020				10	61,517	61,517	120 922				
Debt payments	əryi	40,000	Annual costs and debt payments	•		12	64,003	02,740	129,022				
			ORM	3	¢	0 13	65 283	65 283	259 108				
Income tax analysis		Π	Fuel cost - proposed case		s 471	02 14	66 588	66,588	325,697				
niconic tax analysis			Debt payments - 7 yrs		\$ 40.0	80 15	67,920	67,920	393,617				
			Total annual costs		\$ 87.1	82 16	69.279	69.279	462,895				
					•	17	70.664	70.664	533,560				=
			Periodic costs (credits)			18	72,077	72,077	605,637				
						19	73,519	73,519	679,156				
						20	74,989	74,989	754,146				
			Annual savings and income										
			Fuel cost - base case		\$ 97,5	67							
Annual income													
Electricity export income													
			Tetel		A 07.0	07							
			rotal annual savings and incom	ie	<b>)</b> 91,5	07							
GHG reduction income													
Net GHG reduction	tCO2/yr	168	Financial viability										
Net GHG reduction - 20 yrs	tCO2	3,354	Pre-tax IRR - equity		% 13.7	%							
			Pre-tax IRR - assets		% 6.7	%							
													V
K ← → M Start / Energy Mo	del 🖌 Cost A	Analysis / Emiss	sion Analysis Financial Analysi	is / Risk /	Analysis / Tools	1				[ (		<b>)</b>	
Ready											₩□ ₽ 80% (	) - 0	(
												2	$\sim$

# **Financial Analysis:**

	Wind Analysis- ELA (Compatibility Mode) - Microsoft Excel	- D - X-
Home Insert Page Layout Formulas	Data Review View Add-Ins	<b>9</b> _ 5 x
Link_FS_Start • 🔿 🏂		¥
Customer premium income (rebate)	Simple payback         yr         8.6           Equity payback         yr         8.9           Net Present Value (NPV)         \$         202,178           Annual Ife cycle savings         \$iyr         19,003           Benefit-Cost (B-C) ratio         1.94           Debt service coverage         1.28	
Other income (cost)	Cumulative cash flows graph  1,000,000	
Clean Energy (CE) production income		
H + + H Start Energy Wodel Cost Analyss	Errisson Analysis 🖉 Financial Analysis 🦯 Risk Analysis 🖉 Tools , 🧐	1
Ready		

# **RETScreen 4.0: Solar Analysis**

#### **Project Information:**

A 9:	Sciar Analysis-ELA (Compatibility	Model - Microsoft Exce	- 0
Home Insert Page Layout Form	ulas Data Review View		<u>v</u> -
ecurity Warning Some active content has been	disabled. Options		
D14 • 125 KV	i .		
	Clean Energy Project Analysis Software		
Project information	See project detablese		
Project name Project location	125 Kw ELA		
Prepared for	DF0 Branu and Dr. Strifey Thomason		
Project type	Power		
Technology	Photovotaic	20	
Grd type	Ott-grid		
Analysis type	Method 2		
fleating value reference	Higher heating value (HHV)		
Show settings			
Site reference conditions	Select citrate date location		
· · · · · · · · · · · · · · · · · · ·			
Climate data location	Keoora Airport		
Show data 🗔			
🕅 🙆 🚱 🧰	Consider Energy Model sheet		
icreen4 2008-01-25	© likelster of Natural Resources Canada 1997-2008	NRCanCETC - Varences	
Start / Energy Nodel / Cost Araly	as , Emession Analysis , Financial Analysis , Risk Analysis , Tools 🦉	J M.	

#### **Site Conditions:**



# **Power Project:**

A + 0 + 1     A + 0     A		Solar Analysis- ELA [Compatibility Mode] - Microsoft Escel	- 0 <b>- X</b> -
Home Insert Page Layout	Formulas Data Review	View Add-Ins	🗑 _ a X
H224 • 🔊 🍂			\$
RETScreen Energy Model - Power project			
Power project			1
Base case nower system			
Grid type	Off-orid		
Technology	Reciprocating eng	te	
Fueltype	Diesel (#2 oil) - I		
Fueirate	SIL 0.800		
Capacity	KW 125.00		
Heat rate	kJRWh 11,000		
Annual O&M cost	\$ 500		
Electricity rate - base case	\$8Wh 0.231	-	
Total electricity cost	\$ 92,394		
Load characteristics			
	Hethod 1		=
	O Method 2		
	Unit Base case	Proposed case	
Electricity - daily - DC	KWh		
Electricity - deity - AC	kWh 1,096.000	800.000	
Intermittent resource-load correlation		Negative	
Percent of month used			
	Base case	Proposed case Energy saved	
Electricity - annual - DC	MWh 0.000	0.000	
Electricity - annual - AC	Wh 400.770	232,000 27%	
Peak load - annual	WW .	115.00	
Proposed case power system			
Inverter			
Capacity	KW 100.0	Peak load - annual - AC	
Efficiency	% 90%		
Miscellaneous losses	% 0%	]	
Battery			
H + + H Start Energy Model Cost	t Analysis 🖉 Emission Analysis 🏑	Financial Analysis 🖉 Risk Analysis 🦯 Tools 🏸	1
Ready			

### **Proposed Case:**

C 20-1-	10. CAN			Solar Analysis- ELA [Compatibility Mode] - Micros	soft Excel	- 0 -
Home Insert Page Layou	d Formulias	Data Review	View	Add-Ins		<b>y</b> _ s
H224 • 🔿 🕺	£					
Show data	Month Jenusty Februsty March April Nay June Juny August September October November December Annual	Daily solar radiation - horizontal KWhim'id 1.48 2.51 4.12 5.35 5.96 6.01 5.99 5.01 3.43 2.19 1.36 1.13 3.72	Duily solar radiation - titled kWthimPild 3.28 4.45 5.58 5.59 5.31 5.59 5.31 5.31 5.31 5.31 5.31 5.31 5.31 5.31	Electricity delivered to Note 175 620 6.06 5.49 5.71 5.26 4.08 3.42 2.84 3.20 57.81		
Annual solar radiation - horizontal Annual solar radiation - titled	Within? Within?	1.36 1.61				
Photovoltaic Type Power capacity Illanufacturer Illodel Efficiency Krominal operating cell temperature Temperature coefficient Salar collector area Control method Illiscellaneous Iosaes	BW S V S/V AT	mono-S 50.04 08 mono-St - AP-120 12.3% 45 0.40% 406 um power point tri 5.0%	43.5%	417 unit(s)	See product detabase	
Summary Capacity factor Electricity delivered to lead	% With	15.7% 57.81	19.8%			
reak ked power system Technology Fuel type Fuel nate Charger efficiency Seconded capacity 4 4 b M Start Energy Model	SIL N KW Cost Analyss En	ciprocating engine Diesel (#2 ol) - L 0.800 85% 115.0 Historn Anglysis	Financial A	nakyss Risk Analysis Taols 🏷		
Ready	erer in miles 1 mil	and to apply		inter Contractor Cross C. C.		
Home Intert Page Layou	a Formulas	Data Review	View	Solar Analysis-ELA [Comparibility Mode] = Micros Add Inc	STREET,	
H224 • •	Cotober Nevember December Annual NWhiter NWhiter	2.15 1.35 1.13 3.72 1.36 1.61	3.05 2.52 2.45 4.40	3.42 2.24 3.30 87.45		
Professional Type Power Capacity International Environment Environment Temperature coefficient Centrol International Centrol International Intercellanceaus International	NAN Na Na Na Na Na Na Na Na Na Na Na Na Na	riene-Si 55:04 58: 12:314 0:40% 45: 0:40% 405 405 405 405 405 405 405 405 405 405	43.5%	3417 wagos	See unotest antabase	
Summary Caesely fector Electricity delement to tead Peek load sower system Technology Fuel tops Fuel tops Charger afficiency Sagpated capacity Sagpated capacity Electrosty delement to bed Marchachmer Heat tops	St. Br	15.7% 57.81 Constitute engine 0.000 55% 115.0 11	18.0%	3	äre avoikst sekkase	
		Conciente Cost	talitais aherr			
• • • • Start _ Energy Hodel _= 0	Cost Analyse En	ssion Analyse	Financial A	nalyse 🛫 Rok Analyse 🚬 Took , 🧐 🖉		In many second second

#### **Cost Analysis:**

(□→) =			Solar	Analysis- ELA [Com	patibility Mode] - Microsoft Excel		
Home Insert Page Layout	Formulas Data	a Review	View Add-Ins				0 - ° X
Link_CA_Start $\bullet$ $f_x$							¥
RETScreen Cost Analysis - Power project							-
Settings						i	
Method 1	Notes/Ran	ae					
Method 2	C Second cu	irrency	Notes/Range	None			
	Cost alloca	ation					
Initial costs (credits)	Unit	Quantity	Unit cost	Amount	Relative costs		
Feasibility study							
Feasibility study	cost	1	\$ 10,000	\$ 10,000			
Sub-total:				\$ 10,000	1.6%		
Development			a 45.000	1.0 45.000			
Development	COST	1	\$ 15,000	\$ 15,000	0.10/		
SUD-total:				\$ 15,000	2.4%		
Engineering	cost	1	\$ 55,000	s 55.000			
Sub-total:	GUGI		\$ 33,000	\$ 55,000	8.9%		=
Power system				a 221000	0.070		=
Base load - Photovoltaic	kW	50.04	\$ 9,100	\$ 455.364			
Peak load - Reciprocating engine	kW	115.00	• •,•••	s -			
Road construction	km			s -			
Transmission line	km			s -			
Substation	project			S -			
Energy efficiency measures	project			s -			
User-defined	cost	1	\$ 50,000	\$ 50,000			
				s -			
Sub-total:				\$ 505,364	81.7%		
Balance of system & miscellaneous				•			
Spare parts	%			s -			
Transportation	project			s -			
Training & commissioning	p-d	6	\$ 65	\$ 390			
User-defined	cost			S -			
Contingencies	%	5.0%	\$ 585,754	\$ 29,288			
Interest during construction	7.00%	2 month(s)	\$ 615,042	\$ 3,588	E 10/		
Sub-total:				\$ 33,265	5.4%		
				\$ 010,029	100.0%		
Annual costs (credits)	Unit	Quantity	Unit cost	Amount			
O&M	onn	quantity	oneooot	rinount			
Parts & labour	project	16	\$ 55	\$ 880			
User-defined	cost			s -			v
H + + H Start / Energy Model / Cost A	nalysis Emissio	n Analysis 🏑	Financial Analysis	Risk Analysis 🖉 T	ools 🖓		
Ready							<b>■ □ □</b> 86% (−) − <b>□</b> (+)
() () () () () () () () () () () () () (			Solar	Analysis-ELA [Com	patibility Mode] - Microsoft Excel		
Home Insert Page Layout	Pormulas Data	a Review	View Add-Ins				9 - ° X
Link_CA_Start • (* 5/2)	**			1		1.0	2
Transportation Training & commissioning	project	4	3 46	5			-
User-defined Contingencies	cest	5.0%	\$ 505.754	\$ 29,200			
Interest during construction Sub-total	7.86%	2 month(a)	\$ 615,642	5 3,500 8 33,265	5.4%		
Total initial costs				\$ 411,629	100.0%		
Annual coats (credits) O&M	Unit	Quantity	Unit cost	Amount			
Parts & labour User-defined	project (0.51	16	\$ 55	8 000			
Contingencies Sub-tata:	76		8 880	5 5 880			
Puel cost - proposed case Diesel (#2 oil)	L	73,225	\$ 0.800	5 55,550			
546-1916	11.12	Puscella	Incit a cast	\$ 56,550			
Fuel cost - base case	Unit	Guantity	Unit cost	Amount			
Sub-total:		114,007	3 0.004	5 92,394			
Periodic costs (credits)	Unit	Year	Unit cost	Amount			-
Red of review bis	CONT	12	3 50,000	8 -			
action of a programmer		to Emission Ann	should always				
H + + H Start / Energy Hodel Cost A	nalysis / Emissio	n Analysis 🏒	Financial Analysis	Risk Analysis / T	ook 🖉 🖘		14
Ready							

## **Emission Analysis:**

			Solar Anal	ysis- ELA [Cor	npatibility Mode] - Mi	icrosoft Excel		100	1	_ 0 X
Home Insert Page Layou	t Formulas Da	ta Review Vie	w Add-Ins							0 _ 🕫 X
Link_GHG2_Start 🔻 🕥 🎜	c .									×
RETScreen Emission Reduction Anal	ysis - Power project	t								-
Emission Analysis										
ି Method 1 ି Method 2 ି Method 3										
Base case system GHG summary (Baseli	ne)									
Fuel Fuel type 9 Diesel (#2 oil) 100.0 Total 100.0	mix % %				Fuel consumption <u>MWh</u> 1,225 1,225	GHG emission factor tCO2/MWh 0.252 0.252	GHG emission tCO2 308 308			
Pronosed case system GHG summary (D	ower project)									
Fuel type         Fuel           Diesel (#2 oil)         100.0           Solar         0.0           Total         100.0	mix % % %				Fuel consumption MWh 781 0 781	GHG emission factor tCO2/MWh 0.252 0.000 0.252	GHG emission tCO2 197 0 197			Ē
CHC omission reduction summary										
Power project	Base case GHG emission tCO2 308	Proposed case on GHG emission tCO2 197			Gross annual GHG emission reduction tCO2 112	GHG credits transaction fee % 0%	Net annual GHG emission reduction tCO2 112			
Net annual GHG emission reduction	112	tC02	is equivalent to	22.8	Cars & light truck:	s not used				Ţ
If     I     I     Start     ✓ Energy Model     ✓ C	Cost Analysis 🚶 Emissi	on Analysis Finan	cial Analysis 🔏 Ris	k Analysis 📈	Tools 🖉			I		
Ready									₩ 🛛 💾 88% (	9

# **Financial Analysis:**

Image: A = A = A = A = A = A = A = A = A = A				Solar Ana	Ilysis- ELA	[Compatibility	Mode]	- Microsoft Exc	el					X
Home Insert	Page Layout	Formulas Da	ata Review View	Add-Ins									0 -	= x
Link_FS_Start 👻	9 fx													¥
RETScreen Financial Analysis	s - Power project													-
Financial parameters			Project costs and savings/in	ncome sum	mary		Yearly	cash flows						
General			Initial costs				Year	Pre-tax	After-tax	Cumulative				
Fuel cost escalation rate	%	5.0%	Feasibility study	1.6%	ş	10,000	#	\$	\$	\$				
Inflation rate	% 9/	2.5%	Development	2.4%	ې د	15,000	0	-247,452	-247,452	-247,452				
Project life	/0 Vr	25	Power system	81.7%	ŝ	505,000	2	-21,500	-21,500	-205,415				
	<i>P</i>				•	000,001	3	-18,374	-18,374	-308,009				
Finance							4	-16,441	-16,441	-324,449				
Incentives and grants	\$						5	-14,410	-14,410	-338,859				
Debt ratio	%	60.0%					6	-12,277	-12,277	-351,136				
Debt	\$	371,178	Balance of system & misc.	5.4%	\$	33,265	7	-10,037	-10,037	-361,173				
Equity	S	247,452	Total initial costs	100.0%	\$	618,629	8	-7,684	-7,684	-368,857				
Debt interest rate	%	8.50%					9	-5,213	-5,213	-374,069				
Debt term	yr Shar	10					10	-2,617	-2,617	-3/6,68/				
Debt payments	əvyi	30,370	Annual coete and debt nave	onte			12	-7 703	-7 703	-320,000				
			O&M	icina	s	880	13	62 548	62 548	-265 163				
Income tax analysis			Fuel cost - proposed case		š	58,580	14	65,706	65,706	-199,457				
			Debt payments - 10 yrs		\$	56,570	15	69,022	69,022	-130,435				
			Total annual costs		\$	116,030	16	72,505	72,505	-57,929				=
							17	76,163	76,163	18,234				
			Periodic costs (credits)				18	80,005	80,005	98,239				
			User-defined - 12 yrs		\$	50,000	19	84,039	84,039	182,278				
							20	88,277	88,277	270,554				
							21	92,726	92,726	363,281				
			Annual savings and income				22	97,400	97,400	400,000				
			Fuel cost - base case		s	92 394	24	17.025	17.025	580 013				
Annual income					•	02,001	25	112,875	112.875	692,888				
Electricity export income										,				
			Total annual savings and ir	icome	\$	92,394								
CHC reduction income		-												
ono reduction income		U.	L											
Net GHG reduction	tCO2/vr	112	Financial viability											
Net GHG reduction - 25 vrs	tC02	2,795	Pre-tax IRR - equity		%	6.3%								
		_,	Pre-tax IRR - assets		%	2.0%								
														v
K + H Start / Eporov	Model Cost A	nalvsis / Emice	ion Analysis Financial An	alvsis 🖉 Pi	sk Analycia	Tools	ή				ī	(		
Death	HOUGE COSL P		ion Androis 2 manual An		ak milaiyaa		J							
кеаду												E U U 809		-U

	Solar Analysis- ELA [Compatibility Mode] - Microsoft Excel	- C - X-
Home Inset Page Layout Formulas D	ata Review Wew Addins	ý. 5 X
Link_FS_Start + (> fx		¥
GHG reduction income	Financial Viability	
Net GHG reduction - 25 yrs 1002 2,785	Pre-tax RR - equity         %         6.3%           Pre-tax RR - equity         %         2.0%           After-tax RR - equity         %         6.3%           After-tax RR - equity         %         6.3%           Simple payback         yr         18.8	
Customer premium income (rebate)	Equity payback yr 16.8 Net Present Value (NPV) \$ -117,691 Annual He cycle savings Slyr -11,962 Benefit-Cost (8-C) rate 0.52 Debt service coverage 0.61 GRG reduction cost SHC02 107	
Other income (cost)	Cumulative cash flows graph 800,000	
Clean Energy (CE) production income	600,000 400,000 0 1 2 3 4 6 5 7 8 9 10 11 12 13 14 15 16 17 18 19 20 21 22 23 24 25 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	
H + + H Start Energy Model Cost Analysis Emis	sion Analysis Financial Analysis Rek Analysis Tools 🖉	

### **<u>RETScreen 4.0: CHP Analysis</u>**

#### **Project Information:**

	RETSOR	end-1 - Microsoft Excel	a - m Da <mark>- Ka</mark>
Home Insert Page Layout	Formulas Data Review View RETScreen		🦉 - 5 X
• (* fr 0	HP		¥
Natural Resources Ressources naturelles Canada Canada		Canadă	
	RETScreen <sup>*</sup> International www.retscreen.net Clean Energy Project Analysis Software		
12			
Project information	See project delabase		
Project name Project location	CHP ELA		
Prepared for Prepared by	DFO Bhanu and Dr. Shirley Thompson		
Project type	Combined healing & power		
Grid type	Off-grid		1
Analysis type	Method 2		
Heating value reference	Higher heating value (HHV)		
Show settings	•		
Language - Langue User manual	English - Anglais English - Anglais		
Currency	\$		
Unds [	Netric units		
Site reference conditions	Select clinale data location		
Cimate data location	Kenora Airport		
Show data			
N + + H Start / Load & Network / Ene	rgy Nodel 🦿 Cost Analysis 🦯 Emission Analysis 🦯 Financial Analysis	Risk analysis _ Tools _ 10	PI I
Ready			

# Load and Network Design:

				RETSO	reen4-1 - Micro	osoft Excel							- 0	X
Home Insert Page Layout	Formulas I	Data Review V	iew RETSc	reen									() _ (	J X
B70 <b>▼</b> ∫ <sub>x</sub>														¥
RETScreen Load & Network Design - Combin	ed heating & nov	ver project												F
Heating project	linit	i ci project												Ē
noung project	onit			_										
Base case heating system	Mu	iltiple buildings - space he	ating											
See technical note on heating network design				Building cluste	ers	•						•		_
	42	0.000		1	2	3	4	5	6	1	8	9	1	<u>.0</u>
Number of buildings in building cluster	huilding	0,000		1,000	1,500	1,500	1,500							-
Fuel type	building	4		Dronane I	Dronane I	Dronana I	Dronana I							
Seasonal efficiency	96			80%	80%	80%	80%							=
Heating load calculation	/0	-		0070	0076	0076	00 /0							
Heating load for building cluster	W/m <sup>2</sup>	٦.		80	80	80	80							
Domestic hot water heating base demand	%	15%	1	00	00	00	00							
Total heating	MWh	114	1	29	29	29	29							
Total peak heating load	kW	45		11	11	11	11							
Fuel consumption - unit				L	Ĺ	Ĺ	Ĺ							
Fuel consumption - annual				4,845	4.845	4,845	4,845							
Fuel rate - unit				S/L	\$/L	\$/L	\$/L							
Fuel rate				0.450	0.450	0.450	0.450							
Fuel cost		\$ 8,722		\$ 2,180	\$ 2,180	\$ 2,180	\$ 2,180							
Proposed case energy efficiency measures				,										
End-use energy efficiency measures	%	20%		20%	20%	20%	20%							
Net peak heating load	kW	36		9	9	9	9							
Net heating	MWh	92		23	23	23	23							
														_
Proposed case district heating network		Estimate/Total												
Heating pipe design criteria			1											
Design supply temperature	<u>°C</u>	120	-											
Design return temperature	<u>°C</u>	60												
Differential temperature	-C	60												
Main neating distribution line			1											
Nam pipe network oversizing	70 Lood	Longth	Dino aiza	la tha huilding	aluator aunali	nd hu thia nina .	nantion? lunala							
Pipe sections	LOad	Lengui	Pipe size		ciuster suppli	ed by this pipe	section: (yes/iid	<u> </u>	6	7	0	0		-
Section 1	KIT			1	2	J	4	9	0	1	0	3		<u></u>
Section 2			-											
Section 3			-											
Section 4			1											
Section 5			1											
Section 6	-1		1											
Section 7			1											
Section 8			1											
Section 9			1											T
II	ergy Model 📈 C	Cost Analysis 🖉 Emiss	on Analysis	Financial Anal	rsis 🖉 Risk an	alysis / Tools	/2/		1					- [
Ready											80	% 🕣	<b>V</b> +	-ŧ

#### **Power Project:**



# **Energy Model:**

				RETScreen4-1	Microsoft Exce	ł			
Home Insert Page Layout	Formulas Di	ata Review View	RETScreen						🗑 _ 🔿 X
Link_ES_Start 🔹 🎓 🍂									\$
RETScreen Energy Model - Combined heating	& power project						0	Show alternative units	
Proposed case power system							ł		
System selection		Base load system							
Base load power system									
Technology		Other							
Availability	%		100.0%	8,760 h					
		Alasha kud							
Fuelbase		Single tuel Wood , callate							
Fuel rate	57	238.000							
1.001	-	1.00.000	1						
Other									
Description									
Power capacity	kW	100.00	114.4%					See product database	
Minimum capacity	%	25.0%							
Electricity delivered to load	WWh	496	100.0%						
Electricity exported to grid	IWh	0							
Manufacturer Madel									
Heat rate	kUKWh	11.000							
Heat recovery efficiency	%	55.0%	1						
Fuel required	GJ/h	1.1	-						
Heating capacity	KW	113.1	316.9%						
Operating strategy - base load power system									
Fuel rate - base case heating system	SAWh	76.21							
Electricity rate - base case	Sawn	230.91							
Flectricity export rate	SMM	42.00	1						
Electricity rate - proposed case	SMWh		1						
	-		,						
				Remaining		Remaining			
		Electricity delivered to	Electricity	electricity	Heat	heat	Power	Operating	
Country shall be		load	exported to grid	required	recovered	required	system fuel	profit (loss) Efficiency	
Euloperating strategy		496	344	A A	60	0	2.677	8 050 38 14	
Power load following		496		ő	92	ŏ	1.516	57,776 38,8%	
Heating load following		5		491	5	86	14	114,337 69.7%	
		Provide the second							
Select operating strategy		Power load following							
									×
H 4 F M Start / Load & Network Ene			1						
	argy Model Co	ist Analysis 🖉 Emission Ar	nalysia 🖉 Financi	al Analysis 🏑 I	čisk analysis 🏒	Tools / 🞾 🦯		14	
Ready	ergy Model / Co	ist Analysis / Emission Ar	nalysis 🏑 Financi	al Analysis 🏑 I	iisk analysis 🏒	Tools 🖉			
Ready	ergy Model <u>/ Cr</u>	est Analysis / Emission Ar	nalysis 🦯 Financi	RETScreend-1	isk analysis 📝	Tools / 🗐 🦯	_	R.	
Ready	Formulas	ast Analysis / Emission An	BETScreen	al Analysis / I	tisk analysis 🧹 - Microsoft Exce	Tools / 🞾 🦯	-	X	
Ready	Formulas D	ast Analysis / Emission Ar Data Review View	RETScreen	al Analysis / I RETScreen4-1	ösk analysis 🧹 - Microsoft Exce	Tools / 🖸 🦯			
Ready Home Insert Page Layout Link_ES_Start • 1	Formulas D	s <del>it Analysis <u>F</u>mission Ar</del> Data Review View	nalysis / Financi RETScreen	al Analysis 🦯 I	tisk analysis 🧹	Tools 🦯 🔁 🦯	-		x * *
Ready Home Insert Page Layout Link_ES_Start • 1 fe Proposed ease system characteristics Research	ergy Model <u>A Co</u> Formulas D Unit	ust Analysis <u>Emission Ar</u> Data Review View Estimate	nalysis / Financia RETScreen	al Analysis 🦯 I	čisk analysis 🧹	Tools 🦯 🔁 🦯	System des	i 4	
Ready Home Insert Page Layout Link_ES_Start • for Proposed rese system characteristics Power Base load power system	Formulas C	ost Analysis / Emission Ar Data Review View Estimate	nalysis / Financis RETScreen	al Analysis / I	ösk analysis 🧹	Tools / 🞾 🦯	System des	ilyn graph sfyn graph	
Ready Home Insert Page Layout Link_ES_Start • @ fe Proposed esse system characteristics Power Base load power system Technabay	Formulas ( Formulas (	ot Analysis / Emission An Data Review View Estimate Other	RETScreen	al Analysis / I	šisk analysis 🧹	Tools / 🖓 🦯	System des #Base	lign graph gPeak	
Ready Home Insert Page Layout Link_ES_start • fre Proposed tasks system characteristics Power Base load power system Technology Operating strategy Canadia	Formulas ( Unit	ot Analysis / Emission Ar Data Review View Estimate Other Power toot following	RETScreen	Al Analysis / I	šisk analysis 🧹	Tools / 12	System des #Base	ilign graph gPeak	
Ready Home Insert Page Layout Link_ES_start • file Proposed case system characteristics Power Base load power system Technology Operating strategy Capacity Electricity delivered to load	Formules ( Unit WW	st Analysis / Emission Ar Data Review View Estimate Power load following 100 496	RETScreen	Al Analysis / I	šsk analysis 🧹	Tools / 🔁 🦯	System des #Base	lign graph gPeak	
Ready Home Insert Page Layout Link_ES_Start  Forposed case system characteristics Power Base load power system Technology Operating strategy Capacity Electricity delivered to load Electricity exported to load Electricity exported to load	Formulas ( Romulas ) Unit WV MWh	Ist Analysis / Emission Ar Data Review View Estimate Other Power teat following 100 496 0	RETScreen	RETScreeni-1	- Microsoft Exce	Tools / 12 /	System des Base	ilgn graph gPeak	
Ready Hone Insert Page Layout Link_ES_Start • fa Proposed ease system Characteristics Power Base load power system Technology Operating strategy Capacity Electricity delivered to load Electricity delivered to load Electricity depower system Technology Departure power system Technology	Formulas ( Formulas ) Unit WV MVh	Data Review View  Cother  Power load following  100  496  0	RETScreen	RETScreen4-1	- Microsoft Exce	Tools / 2]	System des •Base	ign graph gPeak	
Ready Home Insert Page Layout Link_ES_Start  Foroposed ease system characteristics Proposed ease system characteristics Power Base load power system Technology Electricity delivered to load Electricity for the second to load Electricity delivered to load Electricity delivered to load	Formulas ( Romulas ( Rott WY MWh MWh	Data Review View  Etatimote  Other  Power load following  100  496  0  Reciprocating engine  Descript 20 - L	RETScreen	RETScreeni-1	isk analysis 🧹	Tools / 92 /	System des •Base	lign graph gPeak	
Ready Home Insert Page Layout Link_ES_Start  Forecased ease system Base load power system Technology Opensing strategy Capacity Electricity delivered to load Electricity delivered system Technology Failtype Failtype Failtype	Formulas ( Konstanting) KW MWh MWh S4,	Int Analysis / Emission Ar Data Review View Estimate Power load following 100 496 0 Reciprocating engine Direct (#2.86) - L 0: 800	RETScreen	RETScreeni-1	- Microsoft Exco	Tools / 92 /	System der #Bise	Besk	
Ready Hone Inset Page Layout Link_ES_Start  Forposed case system Technology Opersing systemy Capacity Electricity delivered to load Electricity exported to grid Peak load power system Technology Electricity exported to grid Peak load power system Technology Fairings Suggested capacity Fairings	Formulas ( Konk WW WW WW WW WW WW	Int Analysis / Emission Ar Data Review View Estimate Other Power load following 406 0 Reciprocating engine Desati (42.61) - L 0.800 0.0	RETScreen	Al Analysis / I	čisk analysis 🦯	Tools / ?] el 300% 250% 150%	System der #3ase	Bign graph gPeak	
Ready Home Insert Page Layout Link_ES_Start • fa Proposed lase system Characteristics Power Base bad power system Technology Densing strategy Capacity Electricity delivered to lead Electricity delivered to lead Fair table Suggested capacity Capacity Electricity delivered to lead	Formulas ( Formulas ) Unit WV WW WW WW WW WW WW	Int Analysis / Emission Ar Data Review View Estimate Other Power teat following 100 408 0 Reciprocating engine Deset (#2 el) - 1. 0.000 0.0 114 0	RETScreen  RETScreen  114.4% 100.0%  150.4% 0.0%	al Analysis / I	čisk analysis 🧹	Tools / ?] el 300% 250% 200% 150%	System dez 2000	sign graph gPeak	
Ready Hone Insert Page Layout Link_ES_Start • fa Proposed ease system characteristics Power Base load power system Technology Operating strategy Electicity delivered to load Electicity delivered to load Electicity exported to grid Peak load power system Technology Fuel type Fuel roles Suggested capacity Capacity Electicity delivered to load Manufacturer	Formulas ( Formulas ( Uotit WW WW WW WW WW WW WW WW WW	Int Analysis / Emission Ar Data Review View Estimate Other Power load following 100 496 0 Reciprocating angline Cleared (42 et) - L 0.00 114 0 Mino lighters	RETScreen  RETScreen  114.4%  100.0%  130.4%  0.0%  See FD8	al Analysis / I	čisk analysis 🧹	Tools / 2	System der •2ase	ilgin graph gPeak	
Ready Hone Insert Page Layout Link_ES_Start  Proposed esse system characteristics Power Base load power system Technology Operating strategy Capacity Electricity delivered to load Electricity delivered to load Massific turer Model Indel Indel Indel	Formulas C Formulas C WW WW WW SA, WW WW WW	Int Analysis / Emission Ar Data Review View Estimate Power load following 100 496 0 Reciprocating engine 0.000 0.0 114 0 Wing Motors Wojkorm	RETScreen  RETScreen  114.4%  100.0%  130.4%  0.0%  100.6%  100.4%  100.1%  10	RETScreeni-1	išk analysis 🧹	Tools / ?] el 300% 250% 200% 150% 50% 50%	System der •Base	lign grzęb gPesk	
Ready Home Insert Page Layout Link_ES_Start  Forecased ease system Base load power system Technology Opensing strategy Capacity Electricity delivered to load Electricity delivered to load Manufacturer Model Hoet role Back-up power system (optional)	Formulas ( KW KW KW KW KW KW KW KW KW KW	St. Analysis         Emission Ar           Data         Review         View           Estimate         Other         Other           Power load following 100 496 0         100 496 0         Other           Reciprocating engine         Disast (#2.et) - L 0.00 0.0         0.0           114 0         0         Mine Motors Weight (#1.000	No         No           RETScreen         %           114.4%         100.0%           100.0%         %           100.0%         %           100.0%         %           100.0%         %           100.0%         %	Al Analysis / I	itisk analysis 🧹	Tools / ?] el 300% 250% 200% 150% 50% 0%	System de: #Base	Ign graph BPeak Fearroy delivered	
Ready Home Insert Page Layout Link_ES_Start • Je Proposed case system characteristics Power Base bad power system Technology Operating strategy Capacity Electricity delivered to load Electricity devorted to grid Peak bad power system Technology Failings Failings Failings Suggested capacity Capacity Electricity delivered to load Manifecturer Model Model Mach-up power system (optional) Technology	Formulas ( Formulas ) WW WW WW WW WW WW WW WW WW KWW	Int Analysis / Emission Ar Data Review View Estimate Other Power teas following 100 498 0 Reciprocating engine Desail (42.et) - L 0.800 6.0 114 0 Hino Bioters W0407rth 11,000 Reciprocating engine	No         Financial           RETScreen         %           114.4%         100.0%           130.4%         0.0%           See FDB         1 un8(s)	Al Analysis / I	čisk analysis 🧹	Tools / 2	System des ©Esse	Bign graph gPeak Energy delivered	
Ready Hone Insert Page Layout Link_ES_Start • fa Proposed ease system Characteristics Power Base bad power system Technology Departing strategy Capacity Electricity delivered to load Electricity delivered to load Manufacturer Suggested capacity Capacity Electricity delivered to load Manufacturer Model Heat rate Back-up power system (optional) Technology Capacity Capacity Capacity Capacity Electricity delivered to load Manufacturer Model Heat rate Back-up power system (optional) Technology	Formulas ( Formulas ( Uotit WW WW WW WW WW WW WW WW WW WW WW WW	bit Analysis / Emission Ar Data Review View Essimate Other Power bad following 100 496 0 Reciprocating engine 0.0 114 0 Mine Noters V04QT/T1 11,000 Reciprocating engine 114	RETScreen 8 114.4% 100.6% 130.4% 0.0% <u>See POI</u> 1 unt(s)	Al Analysis / I	šisk analysis 🧹	Tools / 2	System det Base Capacity	Energy delivered	
Ready Hone Insert Page Layout Link_ES_Start • fa Proposed esse system characteristics Power Base bad power system Technology Opensing strategy Capacity Electricity delivered to lead Electricity delivered to lead Electricity delivered to lead Electricity delivered to lead Manifecturer Model Heat role Back-up power system (optional) Technology Capacity Electricity delivered to lead Manifecturer Model Heat role Back-up power system (optional) Technology Capacity Electricity delivered to lead Manifecturer Model Heat role Back-up power system (optional) Technology Capacity Electricity delivered to lead Manifecturer Model Heat role Back-up power system (optional) Technology	Formulas t Formulas t WY WY WY WY WY WY WY WY WY WY WY WY	Int Analysis / Emission Ar Data Review View Estimate Power load following 100 496 0 Reciprocating engine 0.00 114 0 Mino Motors V0940/7/1 11,000 Reciprocating engine 114	RETScreen	Al Analysis / I	šisk analysis 🧹	Tools / 22	System der #2ase Capacity	Energy delvered	
Ready         Hone       Insert       Page Layout         Link_ES_Start       Image: Compare the second secon	Formulas G Formulas G WW WW WW WW WW WW WW WW WW WW WW	Int Analysis / Emission Ar Data Review View Construction Power load following 100 406 0 Reciprocating angline Cireati (42.80 - L 0.00 104 0 Mice Bioters Wee Bioters Wee Bioters 114 0 Hice Bioters 114 0 Deter 114 0 Deter	No.         Pranci           RETScreen         N           114.4%         100.0%           130.4%         0.0%           See POI         1           1 met(s)         1	RETScreenI-L	itsk analysis 🧹	Tools / 2	System det Biss Capacity Biss Capacity Biss Signal Sign	Intercord delivered	
Ready           Home         Insert         Page Layout           Link_ES_Start         •         Ja           Proposed case system         Technology         Technology           Deschoor system         Technology         Deschoor system           Technology         Deschoor system         Technology           Deschoor system         Technology         Deschoor system           Technology         Deschoor system         Technology           Deschoor system         Technology         Deschoor system           Technology         Deschoor system         Technology           Pask bad power system         Technology         Deschoor system           Technology         Deschoor system         Technology           Base bad power system (optional)         Technology         Deschoor system           Heating         Base bad heating system         Technology         Deschoor system           Heating         Base bad heating system         Technology         Deschoor system	Formulas L KW KW KW KW KW KW KW KW KW KW KW KW KW	Int Analysis / Emission Ar Data Review View Estimate Power load following 100 406 0 Reciprocating engine Desat (42.e6) - L 0.000 0.0 114 0 Mine Noters Weak Store Wine Noters 11,000 Reciprocating engine 114 0 Cher 113.1 0	No.         Pranci           RETScreen         %           114.4%         100.0%           130.4%         0.0%           130.4%         0.0%           1 unit(s)         1 unit(s)	Al Analysis / I	itsk analysis 🖉	Tools / 92	System der #Base Capacity	Bign graph gPask Energy delvered	
Ready Hone Insert Page Layout Link_ES_Start • fa Proposed lase system Characteristics Power Base bad power system Technology Capach Electricity delivered to load Electricity delivered to load Electricity delivered to load Electricity delivered to load Electricity delivered to load Manifecturer Model Hoot role Base load houser system (optional) Technology Capachy Heating Base load heating system Technology Capachy Heating Base load heating system Technology Capachy Heating delivered Heating delivered	Formulas ( Formulas ) WW WW WW WW WW WW WW WW WW WW WW WW WW	Int Analysis / Emission Ar Data Review View Estimate Other Power teas following 100 498 0 Reciprocating engine Desail (42.81) - L 0.800 6.0 114 0 Hino Bioters W04007rth 11.000 Reciprocating engine 114 Other 113.1 92	No         Financial           RETScreen         %           114.4%         100.0%           130.4%         0.0%           See FOOI         1 und(s)           316.9%         100.0%	Al Analysis / I	čisk analysis 🧹 - Microsoft Exc	Tools / ?] el 300% 250% 50% 0%	System det @Esse Capacity at	Bign graph gPeak Energy delivered	
Ready Hone Insert Page Layout Link_ES_Start • fa Proposed ease system Characteristics Power Base bad power system Technology Departing strategy Capacity Electricity delivered to load Electricity delivered to load Electricity exported to grid Peak bad power system Technology Failings Failings Failings Electricity delivered to load Manute churer Model Heat rate Back-up power system (optional) Technology Capacity Electricity delivered to load Manute churer Model Heat rate Back-up power system (optional) Technology Capacity Heating delivered Intermediate load heating system Technology	Formulas ( Formulas ( Unit WY MWh SR, WY WW WW MWh WW MWh	International and a series of the series of	RETScreen % 114.4% 100.6% 130.4% 0.0% See POI 1 unt(s) 316.9% 100.9%	al Analysis / 1	šisk analysis 🧹	Tools / 2	System des Base Capacity	Energy delivered	
Ready         Hone       Insert       Page Layout         Link_ES_Start       Image: Comparison of Amage Comparison         Proposed esse system characteristics         Power       Base bad power system         Technology       Operating strategy         Capacity       Electricity delivered to load         Electricity delivered to load       Electricity delivered to load         Fuel role       Suggested capacity         Capacity       Electricity delivered to load         Fuel role       Back-cuper         Suggested capacity       Capacity         Electricity delivered to load       Manta fucturer         Model       Heating         Heating       Back-up power system (optional)         Technology       Capacity         Heating       Base load heating system         Technology       Capacity         Heating delivered       Intermediate load heating system         Technology       Paak load heating system	Formulas ( Formulas ( WW WW WW WW WW WW WW WW WW WW WW WW WW	International and a second sec	No         Financial           RETScreen         N           114.4%         100.0%           130.4%         0.0%           See POII         N           110.0%         316.9%	RETScreenI-L	itsk analysis 🖉	Tools / 22	System det Biss Capacity at Capacity	Intercontended and a second and	
Ready         Home       Insert       Page Layout         Link_ES_Start       •       Ja         Proposed case system       Technology         Descloy strategy       Capacity         Capacity       Exclosity delivered to load         Exclosity delivered to load       Exclosity woonted to grid         Pask bad power system       Technology         Technology       Exclosity woonted to grid         Pask bad power system       Technology         Fasi rate       Suggested capacity         Capacity       Exclosity delivered to load         Model       Heat rate         Base bad heating system       Technology         Capacity       Capacity         Capacity       Execting delivered         Heat rate       Base bad heating system         Technology       Capacity         Capacity       Capacity         Pask bad heating system       Technology         Capacity       Heating delivered         Mating delivered       Intermodiate bad heating system         Technology       Pask bad heating system         Technology       Pask bad heating bastem	Formulas I Formulas I WW WW WW WW WW WW WW WW WW WW	International sectors of the sector of the s	Allysis / Financia RETScreen 5 114.4% 100.6% 100.6% See FDB 1 unt(s) 316.9% 100.8%	Al Analysis / I	itisk analysis 🖉	Tools / 2	System de: #Base Capacity	Energy delivered	
Ready           Hone         Insert         Page Layout           Link_ES_Start         •         Ja           Proposed (ase system that selectistics         Power         Base bad power system           Base bad power system         Technology         Description           Description strategy         Capacity         Capacity           Capacity         Description         Description           Description         Description         Description           Description         Description         Description           Description         Description         Description           Pask bad power system         Technology         Description           Pask bad power system (optional)         Technology         Description           Description         Description         Description           Technology         Description         Description           Technology         Description         Description           Technology         Description         Description           Description         Description         Description           Technology         Description         Description           Description         Description         Description           Technology         Descripti	Formulas I Formulas I WW WW WW WW WW WW WW WW WW WW WW WW WW	International and a second sec	Allysis / Financia RETScreen 5 114.4% 100.0% 100.0% 5 5 5 6 7 100.0% 100.0%	al Analysis / 1	čisk analysis 🖉	Tools / 2	System des gSase Capacity Sacado	Bign graph gPeak Energy delivered	
Ready         Hone       Insert       Page Layout         Link_ES_Start       •       #         Proposed ease system Characteristics         Power       Base bad power system         Base bad power system         Technology         Deschool (ase system characteristics)         Power         Base bad power system         Technology         Deschool (ase system characteristics)         Power         Base bad power system         Technology         Fail type         Suggested capacity         Capacity         Electicity delivered to load         Model         Heat rate         Back-up power system (optional)         Technology         Capacity         Heating         Base bad heating system         Technology         Paak bad heating system         Technology         Paak bad heating system         Technology         Paak bad heating system         Technology         Back-up heating system         Technology         Back-up heating system         Technology         Back-up heating system (optiona	Formulas I Formulas I WY MWH MWH MWH MWH MWH MWH MWH MWH MWH	bit Analysis / Emission Ar Data Review View Estimate Other Power teat following 100 409 0 Reciprocating engine Deser (#2.80) - 1. 0.00 114 0. Hine Noters W040/T/1 11.1 92 Not required 0.0	No         No           REFScreen         %           114.4%         100.6%           130.4%         0.0%           See ADD         1           100.6%         1           316.9%         100.6%	al Analysis / I	šisk analysis 🧹	Tools / 2	System des 2255 2255 Capacity Capacity Capacity	Energy delivered	
Ready         Hone       Insert       Page Layout         Link_ES_Start       Image: Constraint of the second	Formulas I Formulas I WV WV WV WV WV WV WV WV WV WV WV WV WV	International and a series of the series of	No         Financial           RETScreen         N           114.4%         100.0%           130.4%         0.0%           See POII         N           110.0%         See POII	RETScreen-L	itsk analysis /	Tools / 2	System der Biss Biss Capacity Dapacity Eparere	Energy delivered	
Ready           Home         Insert         Page Layout           Link_ES_Start         Image: Comparison of the sector of t	Formulas I Formulas I WW WW WW WW WW WW WW WW WW WW WW	International and a second sec	Allysis / Financia RETScreen 114.4% 100.6% 130.4% 0.6% See FDB 1 unt(s) 318.9% 100.6%	Fuel consumption	- Microsoft Exc	Tools / 22	System der Base Capacity Capacity Energy delivered	Energy delivered	
Ready         Home       Inset       Page Layout         Link_ES_Start       •       Image: Compare the sector sec	Formulas I Formulas I WW WW SI WW WW WW WW WW WW WW WW	Int Analysis / Emission Ar Data Review View Estimate Power load following 100 406 0 Reciprocating engine Deset (42.et) - L 0.000 0.0 114 0 Hine liders Weat/orm 113.0 92 Not required 0.0 Fuel type	No         Si           114.4%         100.0%           130.4%         0.0%           100.0%         100.0%           100.0%         100.0%           100.0%         100.0%	Fuel consumption	- Microsoft Exc	Tools / 2	System des #Sase Capacity Energy deliverey	Energy delivered	
Ready         Home       Insert       Page Layout         Link_ES_Start       •       Ja         Proposed lase system Characteristics       Power       Base bad power system         Base bad power system       Technology       Description of the system         Technology       Description of the system       Technology         Description of the system       Technology       Description of the system         Technology       Fail rate       Suggested capacity       Capacity         Pask bad power system       Open of the system       Technology         Restricture       Model       Model         Host rate       Base bad heating system       Technology         Capacity       Capacity       Description of the system       Technology         Description       Model       Heating delivered       Model         Heating delivered       Mating delivered       Model       Model         Heating delivered       Mating delivered       Model       Model         Heating delivered       Mating delivered       Model       Model         Mating delivered       Model heating system       Technology       Descriptional)         Technology       Deschop the store       Model       Model     <	Formulas I Formulas I WV WV WV WV WV WV WV WV WV WV WV WV WV	International and a second sec	No         Financial           REFScreen         %           114.4%         100.0%           130.4%         0.0%           See FD0         1           100.0%         1           316.9%         100.0%	RETScreent-1	Fuel     consumption     277	Tools / 2	System des © Sase © Sase Capacity Capacity Energy delivered (MYM) 0 A <sup>MB</sup>	Energy delivered	
Ready         Hone       Insert       Page Layout         Link_ES_Start       Image: Construction of the second seco	Formulas I Formulas I WY MWH MWH MWH MWH MWH MWH MWH MWH MWH MWH	International and a series of the series of	No         Financial           RETScreen         %           114.4%         100.0%           130.4%         0.0%           .0%	ETScreen-L RETScreen-L consumption- unit L	Fuel     consumption     27	Tools / 2	System det 22556 2556 2556 2556 2556 2556 2556 25	Energy delivered	
Ready         Hone       Insert       Page Layout         Link_ES_Start       Image: Construction of the sector of t	Formulas I Formulas I WV WV WV WV WV WV WV WV WV WV WV WV WV	Inst Analysis / Emission Ar Data Review View Estimate Power load following 100 406 0 Reciprocating angline Denail (k2 al) - L 0.00 0.0 114 0 Hite Bibters Welloftriti 11.000 Reciprocating engine 114 0 Hite Bibters Welloftriti 11.000 Reciprocating engine 114 0 Hite Bibters Welloftriti 11.000 Reciprocating engine 114 0 Hite Bibters Welloftriti 11.000 Reciprocating engine 114 0 Hite Bibters Welloftriti 0.0 Field type Wood - pellets Diesel (k2 ol)	Allysis / Financia RETScreen 114.4% 100.0% 130.4% 0.0% See POI 1 unt(s) 316.9% 100.0%	ETScreen-L RETScreen-L consumption unit L	Fuel     consumption     277     Total	Tools / ?] el 300% 250% 200% 150% 150% 0% 0% 0% 0% 0% 0% 0% 0% 0% 0% 0% 0% 0	System det Bisse Bisse Capacity Energy delivered (MWh) 0 406 4 456	Energy delivered	
Ready         Home       Insert       Page Layout         Link_ES_Start       •       Ja         Proposed case system       Technology       Deschool (Sase System)         Technology       Operation (Sase System)       Technology         Deschool (Sase System)       Technology       Deschool (Sase System)         Technology       Operation (Sase System)       Technology         Deschool (Sase System)       Technology       Deschool (Sase System)         Technology       Deschool (Sase System)       Technology         Fail rate       Suggested capacity       Capacity         Capacity       Electricity delivered to load       Maan techner         Model       Heat rate       Base bad heating system       Technology         Capacity       Capacity       Capacity       Capacity         Heating       Base bad heating system       Technology       Capacity         Heating delivered       Intermodiate bad heating system       Technology         Technology       Capacity       Eactricity delivered       Intermodiate bad heating system         Technology       Capacity       Eactricity delivered       Intermodiate bad heating system         Technology       Capacity       Eactricity delivered       Inter	Formulas I Formulas I WW WW WW WW WW WW WW WW WW	bit Analysis / Emission Ar Data Review View Estimate Power load following 100 406 0 Reciprocating angine Diesel (42 et) - L 0, 800 0 114 0 Mine Noters 114 0 Mine Noters 113,1 92 Not required Not required Not required 0,0 Fuel type Wood - pellets Diesel (42 et) Becoverset hast	No         No           RETScreen         No           114.4%         100.0%           100.0%         0.0%           100.0%         No           376.9%         100.0%	ETScreen-L consumption unit L	Fuel     consumption     277     0     Total	Tools / 2	System de: Base Capacity Energy Capacity Energy Capacity 0 496 4 2 6 4 45 6 5 3 50	Energy delivered	
Ready         Home       Insert       Page Layout         Link_ES_Start       •       Ja         Proposed (ase system characleristics       Power       Base boad power system         Base boad power system       Technology       Description         Description       Base boad power system       Technology         Deach       Base boad power system       Technology         Pask baad power system       Technology       Deach         Pask baad power system       Technology       Deach         Pask baad power system (optional)       Technology       Deach         Base boad heating system       Technology       Deach         Base boad heating system       Technology       Deach         Deach       Deach ap heating system       Technology         Dapachy       Heating delivered       Hating delivered         Heating betweed beating system       Technology       Deach         Pask bad heating system       Technology       Deach         Deach-up heating system (optional)       Technology       Deach         Deach-up heating system       Technology       Deach         Deach-up heating system       Technology       Deach         Deach       Deach       Deach <td>Formulas I Formulas I W/ W/ W/ W/ W/ W/ W/ W/ W/ W/ W/ W/ W/</td> <td>Inst Analysis / Emission Ar Data Review View Estimate Power load following 100 406 0 Reciprocating engine Desat (42.et) - L 0.00 0.0 114 0 Hine Noters Wood - pailets Desat (42.et) Not required Not required Vioud - pailets Desat (42.et) Recovered heat</td> <td>Allysis / Financia RETScreen 3 114.4% 100.8% 130.4% 0.6% See FDB 1 unit(s) 318.9% 100.6%</td> <td>Fuel consumption unit t</td> <td>Fuel     consumption     277     0     Total</td> <td>Tools / 2</td> <td>System des gSase Capacity Capacity Energy delivered (MWh) 0 466 4 05 3 52 3 52 1 52</td> <td>Energy delivered</td> <td></td>	Formulas I Formulas I W/ W/ W/ W/ W/ W/ W/ W/ W/ W/ W/ W/ W/	Inst Analysis / Emission Ar Data Review View Estimate Power load following 100 406 0 Reciprocating engine Desat (42.et) - L 0.00 0.0 114 0 Hine Noters Wood - pailets Desat (42.et) Not required Not required Vioud - pailets Desat (42.et) Recovered heat	Allysis / Financia RETScreen 3 114.4% 100.8% 130.4% 0.6% See FDB 1 unit(s) 318.9% 100.6%	Fuel consumption unit t	Fuel     consumption     277     0     Total	Tools / 2	System des gSase Capacity Capacity Energy delivered (MWh) 0 466 4 05 3 52 3 52 1 52	Energy delivered	
Ready         Home       Insert       Page Layout         Link_ES_Start       •       Ja         Proposed lase system that selecistics       Power       Base bad power system         Technology       Operating strategy       Operating strategy         Operating strategy       Operating strategy       Operating strategy         Departing strategy       Operating strategy       Operating strategy         Departing strategy       Operating strategy       Operating strategy         Departing strategy       Departing strategy       Operating strategy         Departing strategy       Departing strategy       Departing strategy <tr< td=""><td>Formulas I Formulas I Unit Unit UNIT UNIT UNIT UNIT UNIT UNIT UNIT UNIT</td><td>tot Analysis / Emission Ar  Data Review View  Estimate  Other Power load following 100 496 0  Reciprocating engine Desat (42.61) - L 0.00 0  Hins Bioters Wood-paleta Desat (42.62) 0  Fuel type Vood - paleta Desat (42.60) Recovered heat Desat Analysis / Emission A</td><td>No.         Financial           RETScreen         %           114.4%         100.0%           130.4%         0.0%           100.0%         1           1100.0%         1           1100.0%         1           1100.0%         1           1100.0%         1           1100.0%         1</td><td>ETScreent-1</td><td>Fuel     consumption     277     0     Total     Fisk analysis</td><td>Tools / 2</td><td>System des gEsse gEsse Capacity Energy delivered (MWR) 0 409 4 0 4 499 3 52 3 52</td><td>Energy delivered</td><td></td></tr<>	Formulas I Formulas I Unit Unit UNIT UNIT UNIT UNIT UNIT UNIT UNIT UNIT	tot Analysis / Emission Ar  Data Review View  Estimate  Other Power load following 100 496 0  Reciprocating engine Desat (42.61) - L 0.00 0  Hins Bioters Wood-paleta Desat (42.62) 0  Fuel type Vood - paleta Desat (42.60) Recovered heat Desat Analysis / Emission A	No.         Financial           RETScreen         %           114.4%         100.0%           130.4%         0.0%           100.0%         1           1100.0%         1           1100.0%         1           1100.0%         1           1100.0%         1           1100.0%         1	ETScreent-1	Fuel     consumption     277     0     Total     Fisk analysis	Tools / 2	System des gEsse gEsse Capacity Energy delivered (MWR) 0 409 4 0 4 499 3 52 3 52	Energy delivered	

#### **Cost Analysis:**

C de la					RETScreen4-1	- Microsoft Excel	- 7 - <del>2</del>
Horse Insert Page Layout	Formulas Dat	Review	View	RETScreen	-		ψ_ π x
Link_CA_Start + 🗧 🕼							3
RETScreen Cost Analysis - Combined h	eating & power pr	oject					
Settings							
<sup>6</sup> lifethod 1	Notes/Ran	çe .					
C lietod 2	Cost aloca	imency stion	Notes/R	ange	None		
Initial costs (credits)	Unit	Quantity	Uni	it cost	Amoent	Relative costs	
Feasibility study							
Feasibility study	cost	1.	\$	5,000 \$	5,000		
Sub-total:	And States		6.10Y	\$	5,000	2.0%	
Development	1		2	14			
Development	cest	1	\$	5,000 \$	5,000		
Sub-totat				\$	5,000	2.0%	
Engineering	100 million 100		11/200				
Engineering	cost	1	\$	15,000 \$	15,000		-
Sub-Intel:				\$	15,000	6.1%	
Power system							
Base load - Other	KSV	100.00	\$	1,800 \$	180,000		
Peak load - Reciprocating engine	kiW	114.00		5	-		
Back-up - Reciprocating engine	KW .	114.00		5			
Road construction	kim			5			
Transmission line	km		-	\$			
Substation	project	3		\$			
Energy efficiency measures	project	S	-	5	+		
User-defined	cost			\$	-		
hanner	2	2		5	+		
Sub-total:				\$	180,000	73.6%	
Heating system	-	1	·	-			
Base load - Other	KW	113.1		5	+		
Energy transfer station(s)	building	4	100	s			
Main heating distribution line pipe	n	0		\$			
Secondary heating distribution line pipe		0	-	\$			
Energy efficiency measures	project			S	-		
Building & yard construction	cest		_	\$	-		
Average - Institutional	cost			5			
Sub-tolal				\$	10	0.0%	
Balance of system & miscellaneous			-	1.			
Spare parts	%	1.000	-	5			
Transportation	project	1	2	5,000 \$	5,000		
Tranno & commissioning	p-d	Anabarta A	12	60   \$	2,400	Rebarden Task (	
to a scart Load & network E	Cost	Analysis / t	chillission Ar	141/56 HI	ancal Analyse	TUM 4114/55 1005	
Ready							

					RETScreen4-	1 - Microsoft Excel	
Home Insert Page Layout	Formulas Data	Review	View	RETScreen	_		🗑 _ 8 X
Link CA Start 🔻 💿 🎜							*
Energy transfer station(s)	building	4		5			
Usin basing distribution ins nine	contrary .	-	-		-		
Sanndary basing distribution ins nine				-			
Secondary meaning data loaden whe pipe	and the set		-	:			
Building 5 und genelauties	proproc						
Average , lostformal	cost		+		-		
Fick latel	0091			-		A.0K	
Sub-total				*	-	0.0%	
East on a miscelaneous							
Spare parts	78			6 440			
Transportation	project	1		5,000 5	5,000		
Training a commasioning	p-a	40		00 3	2,400		
User-defined	cost	200	5	50 5	10,000		
Contrigencies	76	9.9%	- 2	222,400 \$	22,024		
interest during construction	0.9976	e month(s)		244,424		10.10	
Sub-total:				-	38,424	19.1%	
Total initial costs				5	244,424	100.0%	
Annual costs (credits)	Unit	Quantity	Unit	cost	Amount		
08M							
Parts & labour	orgiect	1 1	8	2,000 8	2,000		
Parts & labour User-defined	project	1	\$	2,000 \$	2,000		
Parts & abour User-defined Contingencies	cest	9.5%	5	2,000 \$	2,000		
Parts & labour User-defined Contingencies Sub-tatai	cost %	9.5%	\$	2,000 S S 2,000 S	2,000		
Parts & labour User-defined Contingencies Sub-total: Fuel cost _ proposed case	cost %	9.5%	5	2,000 S S 2,000 S S	2,000		
Parts & labour User-defined Configencies Sub-total: Fuel cost - proposed case Wood - celefax	project cost %	277	5	2,000 S 2,000 S 3 230.000 S	2,000		
Parts & labour User-defined Contingencies Sub-total: Fuel cost - proposed case Wood - pellets Sub-total	project cost %	277	\$	2,000 S S 2,000 S S 230.000 S S	2,000 190 2,190 63,729 63,729		
Parts & labour User-defined Contingencies Sub-latati Fuel cost - proposed case Wood - pellets Sub-latati	project cost %	277	\$	2,000 S S 7,000 S S 230.000 S S	2,000 190 2,190 63,729 63,729		
Patra & labour User-defined Cantingencies Sub-tate: Fuel cost - proposed case Wood - peleis Sub-tate: Sub-tate: Annual savings	t	1 9.5% 277 Quantity	\$ 5 5	2,000 S S 2,000 S S 230.000 S S cost	2,000 190 2,190 63,729 63,729		
Parts & labour User-defined Contingencies Sub-total: Fuel cost - proposed case Wood - pallets Sub-total: Sub-total: Fuel cost - base case	t Unit	1 9.5% 277 Quantity	s s unit	2,000 S S 2,000 S S 230.000 S S cost	2,000 190 2,190 63,729 63,729 Amount		
Parts & labour User-defined Contingencies Sub-trati: Fuel cost - proposed case Wood - pelats Sub-trati: Annual savings Fuel cost - base case Dised (#2 ell)	project cost % t Unit	1 9.5% 277 Quantity 177,700	S S Unit	2,000 S 2,000 S 2,000 S 230.000 S S COIST 0.806 S	2,000 190 2,190 63,729 63,729 Amount 143,160		
Parts & labour User-offined Contingencies Sub-tata: Fuel cost - proposed case Wood - pellets Sub-tata: Sub-tata: Annual exclipion Fuel cost - base case Disol (#2 ell) Propase	Unit L	1 9.5% 277 Guantity 177,700 19,382	S S Unit S S	2,000 S S 2,000 S S 230.000 S S CO251 0.806 S 0.450 S	2,000 190 2,190 63,729 63,729 Amount 143,160 8,722		
Parts & labour User-defined Contingencies Sub-tatal: Sub-tatal: Sub-tatal: Sub-tatal: Fuel cost - proposed case Urood - pelleta Sub-tatal: Sub-	project cost % t Unit L L	1 9.5% 277 Quantity 177,700 19,382	s s unt s	2,000 S 2,000 S 230.000 S 230.000 S 0.806 S 0.806 S 0.806 S 8	2,000 190 2,190 63,729 63,729 63,729 Amount 143,160 8,722 151,881		
Patra & Jabour User-defined Centingencies Sub-tate: Fuel cost - proposed case Vitod - petiels Sub-tate: Sub-tate: Annual savings Fuel cost - base case Diesel (#2 ell) Propaie Sub-tate:	Unit L	1 9.5% 277 Quantity 177,700 19,382	s s unit	2,000 S S 2,000 S S 230.000 S S COEST 0.806 S 0.450 S S	2,000 190 2,190 63,729 63,729 43,729 Amount 143,160 8,722 151,881		
Parts & labour User-offined Contingencies Sub-total: Sub-total: Vood - proposed case Vood - prelefa Sub-total: Annual sub-total: Fuel cost - base case Disect (#2 #) Propare Sub-total: Peripdic costs (credits)	project cost % t Unit L L	1 9.5% 277 Guantity 177,700 19,382 Year	S S Unit S S Unit	2,000 S S 2,000 S S 230,000 S S COSSI 0,806 S 0,450 S S COSSI	2,000 190 2,190 63,729 63,729 Amount 143,160 8,722 151,881 Amount		
Petrs & labour User-defined Cantingencies Sub-total: Sub-total: Sub-total: Sub-total: Sub-total: Annual (saving): Fuel cost - base case Dissel (r2 ell) Propase Sub-total: Periodic costs (credits) User-defined	project cost % t Unit L L Cost	1 9.5% 277 Guantity 177,700 19,382 Year 5	S S S Unit	2,000 S 2,000 S 230.000 S 230.000 S 5 0.806 S 0.806 S 5 5 5	2,000 190 2,190 63,729 64,729 64,		
Parts & labour User-defined Centingencies Sub-tata: Fuel cost - proposed case Wood - pellets Sub-tata: Sub-tata: Annual services Desel (#2 el) Propase Sub-tata: Periodic costs (credits) User-defined	praject cost % t Unit L L Unit Cost	1 9.5% 277 Guantity 177,700 19,382 Year 5	S S S Unit S S S	2,000 S 2,000 S 220,000 S 220,000 S 0,806 S 0,450 S COSSI COSSI S S S	2,000 190 2,190 63,729 63,729 63,729 Amount 143,160 8,722 151,881 Amount		
Parts & labour User-defined Contingencies Sub-total: Fuel cost-proposed case Wood - pellets Sub-total: Sub-total: Sub-total: Sub-total: Propose Desel (#2 #) Propose Sub-total: Periodic costs (credits) User-defined End of project life	project cost % t Unit L Unit Cost cost	1 9.5% 277 Guantity 177,700 19,382 Year 5	S S S Unit S S S	2,000 S 2,000 S 2,000 S 230.000 S 0.806 S 0.450 S 0.450 S S COST S S S	2,000 190 2,190 63,729 63,729 Amount 143,160 8,722 151,881 Amount		
Parts & labour User-defined Centingencies Sub-trate: Fuel cost - proposed case Wood - pelleta Sub-trate: Sub-trate: Annual savings Fuel cost - base case Dissel (#2 eli) Progane Sub-trate: Periodic costs (credits) User-defined End of project life	Unit Cost t Unit L Unit Cost Cost	1 9.5% 277 Guantity 177,700 19,382 Year 5	S S Unit S S Unit Vala sheef	2,860 S 2,000 S 230,000 S 230,000 S 0,806 S 0,806 S 0,806 S 5 5 5 5 5 5 5 5 5 5 5 5 5	2,000 190 2,190 63,729 63,729 Amount 143,160 8,722 151,881 Amount		
Parts & labour User-defined Contingencies Sub-tata: Fuel cost- proposed case Wood - pellets Sub-tata: Annual sub-tata: Fuel cost - base case Disect (#2 el) Progane Sub-tata: Periodic costs (credits) User-defined End of project life	roject cost % t Unit L Cost cost cost	1 9.5% 277 00amtry 177,700 19,382 Vear 5 to Emission Are	S S Unit Visio cheef	2,800 S 2,000 S 230,000 S 5 0,806 S 0,450 S 5 5 5 5 5 5 5 5 5 5 5 5 5	2,000 190 2,190 63,729 63,729 Amount 143,160 143,160 143,160 143,160		
Petro & Lebour User-defined Cantingencies Sub-Intel: Fuel cost - proposed case Wood - palets Sub-Intel: Annual Savings Fuel cost - base case Dissel (22 el) Propase Sub-Intel: Periodic costs (credits) User-defined End of project life Int + M_Start / Load & Network / Ene	ergy Model Cost	1 9.5% 277 Quantity 177,700 19,382 Year 5 fo Emission Ana Analysis / E	S S Unit Visis sheet mission Ana	2,860 S 2,000 S 2,000 S 230,000 S 0,856 S 0,856 S 5 5 5 5 5 5 5 5 5 5 5 5 5	2,000 190 2,190 3,725 63,725 63,725 63,725 43,160 8,722 151,851 Amount Amount	Rsk analyzs / Tools / 93 /	
## **Emission Analysis:**

C 3				RETScreen	-L - Microsoft Excel				00	×
Monte Invett Page Lapout	Formulas Data	Review View	Atticem						<b>V</b> -	
Link_GHG2_Start •										4
RETScreen Emission Reduction Analy	sis - Combined heati	ng & power projec	t					1		-
Emission Analysis										
9 Method 1 1 Method 2 1 Method 3										
Base case system GHG summary (Baselin	e)									
Faultype         Faultype         %           Desset (#2 oil)         530 0%         709           Propane         7.09         100.09	nik 6 6				Fuel         GHS 4           consumption         ft           MWh         nCO           1/894         143           2/037	emission lactor 0.252 0.208 0.249	GHG emission 1CO2 477.2 29.8 506.9			
Proposed case system GHG summary (Co	mbined heating & powe	r project)								
Feel n	nix				Fuel GHG o consumption fu MWh tCO	emission lactor	GHG emission 9CO2			
Wood - peliets 100.09 Total 100.09					1,516 1,516	0.005	9.7 9.7			
GHG emission reduction summary										
Combined heating & power	Base case GHG emission 8002 505.9	Proposed case GHG emission ICO2 9.7			Gross annual GHG emission GHG reduction transu 8CO2 497.2	G credits action fee 5 0%	Net annual GHG emission reduction 8C02 497.2			
Net annual GHG emission reduction	497	1002	is equivalent to	91.0	Cars & light trucks not use	eð				
H 4 + H Start Load & Network	Energy Nodel Cost /	malvas Emission	Analysis Fran	cal Analyse	Risk analysis Tools	2		100		
Ready										e

## Financial Analysis:

			RETScreen4	-1 - Microsof	t Excel				
Home Insert Page Layout	Formulas Da	ata Review View RETScre	een						() _ = )
Link_FS_Start 🔻 💿 f 🖈									
RETScreen Financial Analysis - Combined heati	ing & power pro	ject							
Financial parameters		Project costs and savings/income	summary	Yea	arly cash flows				
General Fuel cost escalation rate %	10.0%	Initial costs Feasibility study 2.	0% S	5.000 #	ar Pre-tax \$	After-tax \$	Cumulative \$		
Inflation rate %	3.5%	Development 2.	0% S	5,000 0	-61,106	-61,106	-61,106		
Project life %	9.0%	Power system 73.	1% \$ 6% \$	15,000 1 180,000 2	68,600 78,217	68,600 78,217	7,494 85,712		
Finance		Heating system 0.	0% \$	0 3	88,802 100,450	88,802 100,450	174,513 274,963		
Incentives and grants \$	75.004			5	113,268	113,268	388,231		
Debt %	183,318	Balance of system & misc. 16.	1% \$	39,424 7	127,374 142,897	127,374 142,897	515,606 658,502		
Equity \$ Debt interest rate %	61,106	Total initial costs 100.	.0% \$	244,424 8	159,978 178 773	159,978	818,480		
Debt term yr	10			10	199,454	199,454	1,196,707		
Debt payments \$/yr	26,100	Annual costs and debt payments		11	248,311 273,350	248,311 273,350	1,445,018 1,718,367		
harring day and hat	-	O&M	s	2,190 13	300,900	300,900	2,019,267		
income tax analysis		Debt payments - 10 yrs	s	26,100 15	331,213	331,213 364,564	2,350,480 2,715,044		
		Total annual costs	\$	92,020					
		Periodic costs (credits)							
		Annual savings and income							
Annual income		Fuel cost - base case	S	151,881					
Electricity export income									
		Total annual eavings and income		454 994					
		rotal annual savings and income	,	151,001					
GHG reduction income									
Net GHG reduction tCO2/yr	497	Financial viability	0/	405.0%					
Net ono reduction - 15 yrs ICO2	7,400	Pre-tax IRR - assets	%	40.1%					
			67	125.00	. /= 1 /#=	7			
Ready	ду модеі 🔬 Со	ST ANAIYSIS / EMISSION ANAIYSIS /	Financial Analysis	KISK analys	IS / 100IS / 🖓			· · · · · · · · · · · · · · · · · · ·	
									0 0. 6
			RETScreen4-	- Microsoft I	Eccel		1000		
Unk FS Start + C Ja	Formulas Dar	ta Review View RETScree	54.1						84 - H X
		Circle saubant		2.0			11		
Customer premium income (rebate)	5	Equity paytack	yr yr	0.9					1
		Net Present Value (NPV) Annual De cycle savings	5 1,1 S/yr 1	72,744 45,489					
		Denetit-Cost (II-C) ratio		20.19					
		Debt service coverage	tanca.	3.63					
Other income (cost)	t.	Cumulative cash flows graph	PXVX	Lengt L			-		
		3,000,000							
Clean Energy (CE) production income	8	2.600.000							
		2,000,000							
		*							1
		s 1,500,000			/				
		1 000 000							
		attive		/					
		B00,000	/						
					72				
		1 2 3	4 5 6	7 8	9 10 f	1 12 13	14 15		
		-500,000							
	- Martin		Year	Web and a	Total Pro-				
Ready Ready	Ty Hopei Cos	CANAGES Emesion Analysis F	mancial Analysis	Hak analysis	1008		1	DO ULAN	÷ 0 (+